# THE EFFECTIVENESS OF PARENTAL INVOLVEMENT IN PRESCHOOL EDUCATION PROGRAMS ON PARENT PERCEPTIONS OF THEIR CHILD'S

# SCHOOL READINESS

A Dissertation Submitted to the Graduate Faculty of the North Dakota State University of Agriculture and Applied Science

By

Judith Ann Konerza

In Partial Fulfillment of the Requirements for the Degree of DOCTOR OF PHILOSOPHY

> Major Program: Education

> > April 2012

Fargo, North Dakota

# North Dakota State University

Graduate School

Title

The Effectiveness of Parental Involvement in Preschool Education Programs on

Parent Perceptions of Their Child's School Readiness

Bу

Judith Konerza

The Supervisory Committee certifies that this *disquisition* complies with North Dakota State University's regulations and meets the accepted standards for the degree of

DOCTOR OF PHILOSOPHY

SUPERVISORY COMMITTEE:

Dr. Myron A. Eighmy, Chair

Dr. Shelby Barrentine

Dr. Sean Brotherson

Dr. James Deal

Dr. Joel Hektner

Approved:

April 26, 2012

Bill Martin

Date

Department Chair

## ABSTRACT

The purpose of this study was to determine if the Gearing Up for Kindergarten program created a significant impact on parent understanding of children's development and aspects of school readiness. Parent perceptions of their child's readiness to make the transition to school were also assessed.

The study also measured the Gearing Up for Kindergarten impact on children's scores on selected academic measures.

A selected sample of 75 parents were surveyed with pre, post, and post post program assessments using the Practical Parent Assessment of School Readiness survey. The survey used Likert scale measures to assess parent perceptions of readiness in the 5 domains of child development: Approaches to learning, Social and emotional development, Physical well-being and motor development, Language development, and Cognition and general knowledge. The survey found significant differences between the treatment and control group on selected measures of the social and emotional scales. The survey also measured parent perceptions of their child's readiness for the transition to school and found no significant difference between treatment and control groups.

The AIMS Web children's assessment measured children's academic knowledge with three one minute tests: letter identification, number identification, and oral counting. This assessment compared children's scores using an ANOVA and found no significant differences in children's scores between treatment and control groups.

iii

#### ACKNOWLEDGEMENTS

I would like to express a sincere thank you to Dr. Myron A. Eighmy at North Dakota State University who chaired my committee and provided guidance, perseverance, and support throughout the dissertation journey. For the thoughtfulness, dedication, and diligence in everything you do for all students, I am most grateful.

I am also appreciative of the time, assistance, and recommendations from all of my committee members. Dr. Joel Hektner, North Dakota State University, supported the statistical process with great care and explanation to this novice researcher. I appreciate your willingness to explain and guide as I sought to understand and report statistical procedures. Dr. Sean Brotherson, North Dakota State University, the father of the Gearing Up for Kindergarten program, your care for families is evident and I appreciate the opportunity to be a part of implementing and shaping this program with you. Dr. James Deal, your expertise is widely known and I appreciate your eagle eye in ensuring quality research at North Dakota State University. Dr. Shelby Barrentine, The University of North Dakota, you have been my advisor through undergrad work, graduate school, and now again as I finish this doctoral work. I cannot thank you enough for your steadfast support, expertise, and friendship; you make a difference in my life and in the lives of all your students. It is an honor to work with you. To the entire committee, thank you for providing me with role models of support, leadership, wisdom, and care. All of you gave from the abundance of your hearts, for that most of all, I thank you.

My greatest of appreciation goes to my husband, Dave, whom I love with all my heart. Thank you for supporting this work. You have given sacrificially that I might

iv

succeed. You have been the glue that held our marriage and home together in this project and you have been the leader of the cheering section; this is your degree, too.

Thank you too, to my children Kim and David, who, when I began this education journey were in grade school. You have been an integral part of my education journey as both parent and teacher. Along the way we have welcomed Richie and Jackson to our family; additions of blessing. I am grateful to all of you for your support, encouragement, insights, and the joy you bring to my heart.

Mom and Dad, you have been waiting for this day to come, I am happy to share it with you. You have always been proud of me for who I am; thank you for your love.

Thank you to the rest of the cheering committee; you have blessed my life in multiple, tangible, ways. This is not a journey to be taken alone, and along the way I have come to know the value of community and care; all of you are a lovely piece of my travels. My deepest gratitude goes to this team: to Becky Trapnell for her servant heart, prayers, and encouraging words, to Machell Thompson for her daily support, cheer, and coffee, to Michelle Connolly for the walks and talks, to my sister, Colleen for sharing in the journey of the heart, to Deb Gebeke and Jody Thompson for affording me this opportunity and the time to make this research possible, and to the Gearing Up staff, Zeljka, Marina, Michelle, Laura, and Machell – it is a joy and a privilege to work with teachers of excellence and dedication.

Finally, I thank the Lord, for being my best teacher. You continue to teach me what it means to be a beloved and cherished daughter. Your love for me continues to bring forth new found strength, understanding, and joy. Eucharisteo means thanksgiving. For this journey and the discovery along the way – Eucharisteo, thanksgiving abounds!

V

ABSTRACT	iii
ACKNOWLEDGEMENTS	iv
LIST OF TABLES	viii
LIST OF FIGURES	ix
CHAPTER 1. INTRODUCTION	1
CHAPTER 2. LITERATURE REVIEW	
CHAPTER 3. METHODOLOGY	74
CHAPTER 4. FINDINGS AND RESULTS	
CHAPTER 5. SUMMARY, CONCLUSIONS, AND RECOMMENDATIONS	
REFERENCES	
APPENDIX A	142
APPENDIX B	144
APPENDIX C	145
APPENDIX D	146
APPENDIX E	
APPENDIX F	
APPENDIX G	
APPENDIX H	
APPENDIX I	
APPENDIX J	
APPENDIX K	161

# **TABLE OF CONTENTS**

APPENDIX L	 	

# LIST OF TABLES

<u>Tabl</u>	<u>e</u>	Page
1.	Practical Parent Assessment of School Readiness: Parent Perceptions of Their Child Aggregate Measures of Constructs	110
2.	Practical Parent Assessment of School Readiness: Parent Perceptions of Their Child's Social Development Measures of Construct Components	111
3.	Practical Parent Assessment of School Readiness: Parent Perceptions of Their Child's Emotional Development Measures of Construct Components	112
4.	AIMS Web: Aggregate Measure of Children's Academic Scores	114

# **LIST OF FIGURES**

Figure		Page
1.	Parent perceptions of their child's letter identification abilities with the child's actual letter identification score on the post-test	115
2.	Parent perceptions of their child's number identification abilities with the child's actual number identification score on the post-test	116
3.	Parent perceptions of their child's oral counting abilities with the child's actual oral counting score on the post-test	117

# **CHAPTER 1. INTRODUCTION**

The increased emphasis on the importance of parental involvement in student education and school readiness is at the forefront of research efforts due to increased expectations of student knowledge and proficiency upon entrance into kindergarten. Before the start of kindergarten, children are most often under the care of parents and not part of a formal education system. Parental involvement and child-rearing practices can vary greatly from one family to the next based on parental beliefs and knowledge of child development. The Harvard Family Research Project (2006a) publication reinforces the importance of parents' involvement during children's early years and the benefits children derive from entering school ready to succeed.

Children who begin school with solid family relationships, age-appropriate social skills, an understanding of cooperation, and school-readiness skill sets are more likely to experience success and enjoyment in the learning experience (Klein, 2002).

There is no formal means to inform parents regarding school readiness in North Dakota; it is one of ten states with no state-funded preschool programming. Lack of a formal system to inform and educate parents does not provide significant opportunities for schools and agencies to share information, with parents, that could assist with early education for children. The early learning years provide important opportunities for parents; "Between birth and age five is the most leveraged opportunity for schools and parents to prevent children from falling behind in development and learning" (Fielding, Kerr, & Rosier, 2004, p. 205). This early window of development is a prime time to share best practices with young parents; "Parents can position their child for academic success long before the first day of kindergarten. A vibrant life of learning begins at birth"

(Fielding, 2009, p. 12) and the home is the child's first classroom with parents as the first teachers. Parents have the most significant role in children's lives and are considered to be their first teachers (North Dakota Department of Human Services, 2008a).

## **Literature Summary**

Research supports the idea that appropriate home learning experiences and a quality home literacy environment positively affect young children's cognitive development, language, reading readiness, and later school success (Fielding, 2006; High/Scope Educational Research Foundation, 2006; Sprenger, 2008). Brain research reinforces that providing warm and loving emotional care to young children is critical to their socialemotional and cognitive development. Positive maternal nurturing impacts children's development, of pre-reading and pre-math readiness skills, and reinforces other aspects of early development which can in turn impact school readiness (Rhode Island KIDS COUNT, 2005; Sprenger, 2008).

Brain development occurs most rapidly during the first three years of life, and it is important for children to be held, cuddled, and talked to during this time (Dickinson & McCabe, 2001; Fielding et al., 2004; Sprenger, 2008). A child's brain continues to develop rapidly until age eight or ten when it slows down, suggesting children's first years are crucial for shaping their capacity to learn (Hinkle, 2000; Sprenger 2008).

Our nation's early childhood education has strong roots in the Head Start program, but the program is limited in the population it serves. In 1989, then President H. W. Bush, together with the nation's governors, worked to form a set of national educational goals for all children, "Goals 2000" (Kagan & Rigby, 2003, p. 2). These goals emphasized the importance of education and growth in early childhood. The first goal stated, "By the year

2000, all children in America will start school ready to learn" (U.S. Department of Education, 1995, p. 1). This report emphasized the link connecting early education to later school achievement.

In 1990, the U.S. Department of Education set a goal for increased parental participation in education: "By the year 2000, every school will promote partnerships that will increase parental involvement in promoting the social, emotional, and academic growth of children" (National Education Goals Panel [NEGP], 1995, p. 13).

The panel proposed that state and local education agencies work together to develop partnership programs that would meet the varying needs of bilingual, disabled, or disadvantaged children and their parents. Programs would support the academic work of children at home, promote shared decision making at school, and hold schools and teachers accountable for high standards of achievement. In the years since the panel called for greater collaboration between schools and families, more attention has been placed on the importance of parent involvement . . . (Patrikakou, Weissberg, Redding, & Walberg, 2005b, p. 4)

Patrikakou, Redding, & Walberg, (2005b) continued to explain how the No Child Left Behind [NCLB] Act of 2001 (NCLB, 2002) has played a central role in our nation's education policy. "NCLB acknowledges that parents play an integral role in their children's learning, and that they should be given the opportunity to act as full partners in their children's education" (p. 4).

School-family partnerships have continued to be the focus of research, policy, and practice efforts in recent years, and Henderson and Mapp (2002) found children benefit when schools and families work cooperatively. Furthermore, effective parents participate in

their children's education by partnering with teachers in working toward the core educational objectives of socialization, emotional growth, and academic learning (Henderson & Mapp, 2002; Patrikakou et al., 2005b). The benefits of such partnerships include higher grades and test scores, better attendance, improved behavior at home and school, better interpersonal skills, and more responsible decision making (Henderson & Mapp, 2002; Patrikakou et al., 2005b).

# **Statement of the Problem**

Schools today face great challenges as they grapple with instructing children at multiple levels of readiness and skills. Children who enter school behind in their development and academic knowledge tend to stay behind as they move into the next grade (Fielding, 2009).

The National Center for Education Statistics [NCES] study of children who entered kindergarten in 1998 found that cognitive and social skills are strongly correlated with income at school entry. Although children in poverty are the furthest behind, children from middle-income families are as far behind children from higher income families as poor children are behind middle class. Most American children are not achieving their potential prior to school entry, and those who start behind tend to stay behind. (National Institute for Early Education Research [NIEER], 2004, p. 2)

While research has demonstrated parental involvement is a vital component to kindergarten readiness, up to 42% of the nation's children begin kindergarten behind (Espinosa, Thornburg, & Mathews, 1997).

In a 1995 survey of 3,500 kindergarten teachers from across the country, many teachers reported that large proportions of their students lacked important school readiness skills. For example 46% of the kindergarten teachers reported that at least half of the students in their classroom had difficulty following directions, 36% reported that at least half of their class lacked academic skills they needed, and 34% reported that at least half of their class had difficulty working independently. In Maryland, only 52% of children who entered kindergarten in 2002 were considered "fully ready." In a 2001 statewide survey, Colorado kindergarten and first-grade teachers reported that four out of 10 children were not academically prepared for school and that about one-third were not socially and emotionally prepared. (NIEER, 2004 p. 2)

Espinosa et al. (1997) conducted a Carnegie Foundation for the Advancement of Teaching research study of over 7,000 kindergarten teachers' perceptions of student readiness revealed more than a third of students entering kindergarten were not ready. Kindergarten teachers from across the country shared their varied views of school readiness; and their perceptions of readiness spanned multiple domains of development and included physical and emotional maturity, language development, general cognition, and a sense of right and wrong. Percentages of readiness "ranged from Hawaii teachers reporting 47% of students were not ready for school to North Carolina teachers reporting that 23% of their incoming students were not ready to participate successfully in kindergarten" (Espinosa, Thornburg, & Matthews, 1997, p. 121). Language proficiency was cited most frequently as teachers' biggest area of concern. Other areas identified as moderately concerning included social awareness and confidence, moral awareness, and overall knowledge. "*Parent education* was most frequently suggested as a way to improve children's readiness for school" (Espinosa et al., 1997, p. 121).

Espinosa et al. (1997) also asked teachers in this study to identify the school readiness of their most recent class. Forty-two percent of the respondents said that fewer students were ready at the time of the study than five years previously; 33% said students were comparably prepared; and only 25% said entering kindergarten students were more prepared than in previous years.

The importance of parents in children's developmental and academic achievement is widely known and understood (Fielding, 2009; Fielding et al., 2004; High/Scope Educational Research Foundation, 2006; NIEER, 2004). A consistent and positive relationship between parent and child is the foundation for children's future success:

It is through consistent relationships that children develop "... self-awareness, social competence, conscience, emotional growth and emotion regulation, learning and cognitive growth" as well as other developmental accomplishments (NRC, 2000, p. 265). Parents are more important to the development of a child's readiness for school than childcare and schools (Child Trends, 2003). (High/Scope Educational Research Foundation, 2006, p. 78)

Existing studies validated the importance of parents' involvement, even calling parental involvement vital in children's educational process and outcomes (Henderson & Mapp, 2002). Children's success in both school and life depends upon multiple factors of support for development and growth (Browning et al., 2006; Fielding, 2009; Fielding et al., 2004; Hanson, 2008). Research supported the critical nature of parental support and nourishment, from birth through age five, for kindergarten readiness and later educational

achievement (Fielding, 2009; Fielding et al., 2004; Weiss, Caspe, & Lopez, 2006). Knowledgeable parents can positively impact the educational achievement of children, especially when parental involvement begins at birth (Fielding, 2009).

If parents do not understand kindergarten readiness, they may believe education begins with formal schooling. This misconception can contribute to academic deficiencies in children (Espinosa et al., 1997; Fielding, 2009; Hanson, 2008; High/Scope Educational Research Foundation, 2006; Weiss et al., 2006). The expectations of kindergarten readiness have changed over time, but especially in recent years; hence, parents may lack a clear understanding of current educational expectations for their kindergartners (Espinosa et al., 1997; Fielding, 2009; Fielding et al., 2004; Hanson, 2008). Research indicates preschool and family education programs with a parent-child component can teach parents in a non-threatening atmosphere about daily learning opportunities and can give parents knowledge of child development that does not require advanced degrees (Rockwell, Andre, & Hawley, 1996). Knowledge of child-rearing practices can empower parents in their parenting practices (Epstein, 1995; Rockwell et al., 1996).

Epstein (1995) and Rockwell et al. (1996) reported that many parents, especially those of minority cultures or those who are themselves uneducated, may feel overwhelmed by the challenges of taking an active role in their child's education. Because some cultures assign the responsibility of formal education to teachers alone, parents from those cultures are not familiar with a parental role in early childhood education and parental involvement when their children are in school. Partnering with schools in their children's education is a foreign concept to some parents (Rockwell et al., 1996).

South Carolina's *First Steps* school-readiness program recognized the importance of the relationship between parents' and children's school readiness and focused on improving parenting practices and increasing school readiness through these core objectives:

- Increase family literacy and parent education levels.
- Improve parental employability and employment.
- Increase the effectiveness of parenting related to child nurturance, learning, and interaction, language, health and safety.
- Increase successful parenting and family literacy programs targeting, service integration, and results documentation.
- Increase parent involvement in preK–12 education settings. (High/Scope Educational Research Foundation, 2006, p. 78)

In a 2004 poll conducted by the National Institute for Early Education Research (NIEER), the following result was noted:

[A] large majority [of] kindergarten teachers believe that their students would be better prepared for school if they participated in prekindergarten. In the poll (2004), 66% of kindergarten teachers rated children who attended prekindergarten as 'substantially better prepared' to start school ready to succeed compared to one percent of teachers who said prekindergarten kids were "less prepared." The vast majority of the teachers, ranging from 78 to 93 percent, said children who had attended quality prekindergarten programs were more likely to get along with others and be sensitive to their feelings, count, have problem-solving skills, know

letters of the alphabet, and follow directions better, and were less likely to disrupt class. (NIEER, 2004, p. 2)

Children who attend high-quality pre-kindergarten programs scored significantly higher on mathematics, reading, and language arts upon entry into kindergarten (Ackerman & Barnett, 2005). Children who began kindergarten with fewer readiness skills than their peers fell further behind as they proceeded through their K-12 school years. Research by Kamerman and Gatenio-Gabel (2007) linked "high-quality preschool education" with school success and reduced crime rates. In addition, as adults those students showed greater levels of employment as well as reduced rates of smoking and risky behaviors. Economically speaking, children who attend a high-quality preschool can save society "up to \$17" for every \$1 spent on preschool (Kamerman & Gatenio-Gabel, 2007, p. 24).

Consistent research findings showed young children are capable learners and their linguistic, mathematical, and other readiness skills are influenced and improved by their educational and developmental experiences during the early years (Fielding et al., 2004). A number of studies identified participation in quality early childhood programs has been consistently associated with increased levels of cognitive development, school achievement, and motivation as well as lower rates of grade retention and special education placement (Barnett & Hustedt, 2005; Peisner-Feinberg & Schaaf, 2007).

Dr. Joyce Epstein's (1987a) original work on parents' roles and involvement in their children's education established evidence that parental involvement in children's development and education affects children's academic growth and achievement, attitudes, and aspirations. Students gain in personal and academic development if their families emphasize education (Epstein, 1987a).

This study will examine parental perceptions of school readiness through participation in the Gearing Up for Kindergarten school-readiness program. North Dakota is 1 of 10 states with no formal early childhood funding for education, and many children are underserved due to a lack of program availability and funding (Barnett, Epstein, Friedman, Boyd, & Hustedt, 2008). This study will research parent perceptions and student growth through participation in the Gearing Up for Kindergarten program. Parents are assessed for perceptions of their child's school readiness at three points in time: preprogram, post-program, and a post post-program assessment when their child begins kindergarten. Children are assessed on selected academic measures at the same time parents complete their assessments. The Gearing Up for Kindergarten program provides a unique focus in children's school readiness by combining two facets of education, parent involvement and school readiness. Many preschool programs focus on preparing children for school. This program works with parents and children to prepare for their joint journey in education (Brotherson, Query, & Saxena, 2009).

# **Purpose of the Study**

The first purpose of this study is to determine if the Gearing Up for Kindergarten school-readiness program played a significant role in teaching parents developmental skills and practices to help children begin school ready to learn, at kindergarten levels of readiness. The second purpose of this study is to determine if children's readiness for school, on selected academic measures, is significantly impacted through participation in the Gearing Up for Kindergarten program.

#### **Research Questions**

This quasi-experimental study will examine the following research questions:

- Is the change over time in the Practical Parent Assessment of School Readiness (PPASR) scores for the treatment group parents different from the change over time for the control group parents for the 5 domains of child development?
- 2. Is the change over time in PPASR scores for the treatment group parents different from the change over time for the control group parents for the transition-to-school knowledge and familiarity construct?
- 3. Are there differences in change over time between treatment and control group in children's knowledge of letter identification, number identification, and oral counting as measured by AIMS Web scores?
- 4. Is there a difference between treatment and control group in the strength of the relationship between the parent PPASR scores and child AIMS Web scores?

#### Significance of the Study

The academic outcomes possible through parental involvement range from gains in early childhood to adolescence and beyond. Early childhood programs with family participation as an integral part of programming show that children from those programs have fewer delinquency problems and experience greater success with employment and relationships in adult life (Patrikakou et al., 2005b).

Research evidence on home-school relations and academic achievement has indicated that children whose parents are involved with early childhood programs such as Head Start score higher on cognitive development scales, use a richer vocabulary, and speak in more complex sentences than do children whose families are not part of such programs (Mathematica Policy Research and the Center for Children and Families at Teachers College, Columbia University, 2001). Also, participants in early childhood programs that had a family collaboration and support component are more likely to score at or above national norms on scholastic readiness tests at school entry. Most importantly, these gains continue to be prominent in later educational performance, with fewer grade retentions and increased high school completion rates (Henderson & Mapp, 2002). (Patrikakou et al., 2005b, p. 7)

The significance of this study is its contribution to a growing body of knowledge through examination of children's learning, in a unique classroom setting that involves parent participation alongside child learning. The learning outcomes are measured to determine if program involvement significantly impacts children's academic scores and parent perceptions of child development and school readiness.

#### **Research Study Design**

This quasi-experimental research study has two main components, a student assessment, on selected academic measures, and a parent perception assessment for understanding of child development and school readiness. The study is designed to assess the effectiveness of the treatment by comparing the scores of students and parents in the treatment and control groups.

This study compares two groups of children on selected academic readiness skills in math and language. One group, the treatment group, participated in the Gearing Up for Kindergarten, a 16-week, interactive, parent/child school-readiness program. The second group, the control group, did not participate in the Gearing Up for Kindergarten program

but participated in the same assessments given to the treatment group. Each student group was assessed at equal intervals, pre-program, post-program and post post-program, with the quantitative academic measuring tool, AIMS Web, to determine the students' knowledge of letter identification, number identification, and oral counting.

The assessment was given to both groups prior to the beginning of the Gearing Up for Kindergarten program in the fall of 2010, and at the conclusion of the Gearing Up for Kindergarten program in the spring of 2011. Students were tested a third time, at the beginning of their kindergarten year.

This study is also designed to compare the effectiveness of the Gearing Up for Kindergarten program on parental perceptions of child development and perceptions of their child's feelings on the transition to school. The assessment instrument for parents is the Practical Parent Assessment for School Readiness (PPASR). This assessment was given to parents of children in both the treatment and control groups. The assessments were administered, pre-program, post-program, and post post-program.

Results will be shared with administrative personnel in the school district; with North Dakota State University (NDSU) Extension administration; and with NDSU researchers who are writing, editing, and researching the Gearing Up for Kindergarten curriculum in partnership with the Parent Education Network in North Dakota.

This study underscores the importance of a program promoting a shared responsibility between families and educators. The Gearing Up for Kindergarten program is designed to facilitate parent understanding regarding the importance of their dual role as parents and educators. A culture of success for children can be created when schools and

parents share goals, contribute cooperatively in student learning, and accept joint accountability for students' success in school (Fantuzzo, Tighe, & Childs, 2000).

Kindergarten readiness is a complex and multidimensional construct influenced by many early childhood experiences. For this study, school readiness is defined using the guidelines developed by the National Education Goals Panel (Kagan, Moore, & Bredekamp, 1995). These goals are the current standards generally accepted by the early childhood community. The NEGP identified five dimensions of children's development and learning that are important for school success:

- Physical well-being and motor development
- Social and emotional development
- Approaches toward learning
- Language development
- Cognition and general knowledge

These five dimensions are interrelated and dependent upon each other; development in one area will affect development in others. In this perspective, there is no single or uniform standard of readiness. Children's development is variable at age five; thus, no child will demonstrate all necessary skills in every area. No one dimension is more important than another, and the goal is for children to develop competencies across all five domains.

## **Definition of Terms**

The following terms and definitions will be used in this study.

**AIMS Web Children's Assessment:** AIMS Web is a benchmark and progress monitoring system based on direct, frequent, and continuous student assessment. The

results are reported to students, parents, teachers, and administrators via a web-based datamanagement and reporting system to determine response to intervention (AIMS Web, 2011). The children's assessments in this study include: letter identification, number identification, and oral counting.

**Early Childhood Education Programs**: Educational programs are available for children prior to the age of formal school entry, including preschool and kindergarten programs (Hanson, 2008).

**Early Learning Guidelines**: State standards are specific about the education of preschool children. These guidelines cover all the readiness areas identified by the National Education Goals Panel, including physical well-being, motor development, socialemotional development, approaches toward learning, language development, cognition, and general knowledge (NEGP, 1995).

Gearing Up for Kindergarten: Since the fall of 2006, the North Dakota State University Extension Service has worked to develop a school failure prevention-focused parent and family education program designed to facilitate child development and school readiness for families in North Dakota. With funding from the United Way of Cass-Clay and the Parent Information Resource Center, the Gearing Up for Kindergarten educational program is offered to parents and children at local schools or child care centers. In the 2007-08 school year, the program operated at 15 sites across North Dakota with 234 families participating. The 16-week program is offered in two 8-week sessions in the fall and the spring of the year before a child enters kindergarten. The program combines parent-child learning activities, parent education, and school-readiness skills for children (Brotherson, Query, & Saxena, 2008).

**Parent Involvement**: Christenson and Sheridan (2001) referred to parent involvement as a parent's role in educating their children at home and in school. Jordan, Orozco, and Averett (2001) identified factors that influence levels and aspects of parental involvement. Family (e.g., education level, family structure, family size, parent gender, and work outside the home) and child (e.g., age, gender, grade level, and academic performance) characteristics are of particular relevance in determining parental involvement. Research has shown that undereducated and single parents are less involved in certain activities (Jordan et al., 2001).

Because no universally accepted definition or framework exists to describe a comprehensive parental involvement program, it is difficult to develop such a program (Patrikakou et al., 2005b). "The lack of a common framework or definition lies primarily in the *multidimensional* [emphasis in original] nature of parent and teacher influences on children, as well as the complexity of home-school connections" (Patrikakou et al., 2005b, p. 2). Studies indicate that a number of factors influence parenting today. These factors include multiple cultural perspectives, differing "*beliefs and expectations* [emphasis in original] held by all," varying levels of knowledge regarding child development, and distinct differences in role expectations throughout the country (Patrikakou et al., 2005b, p. 2).

Dr. Joyce Epstein (1987a), a leading researcher in parental education and involvement, established a common framework for parents' involvement with children's education. Early work established clear evidence that

Parental encouragement, activities, and interest at home and participation in schools and classrooms affect children's achievement, attitudes, and aspirations . . . students

gain in personal and academic development if their families emphasize schooling, let the children know they do, and do so continually over the school years. (Epstein, 1987a, p. 120)

**Parent Response Form**: This family information form gathers demographic information from parents who participate in the Gearing Up for Kindergarten education program assessments. Parents answer questions regarding income, education, marital status, employment, ethnicity, and child information. This form is only completed at the first assessment.

**Practical Parent Assessment for School Readiness (PPASR)**: The PPASR is an assessment tool parents use to rate their perception of their child's development and readiness (in selected areas) for entering school. The PPASR is intended as both a learning tool and an assessment tool to increase parents' awareness of their child's practical and developmental skills and abilities (Brotherson, 2007). Parents completed this assessment three times throughout the study. The use of this instrument can help parents see their child's development over a period of time.

# Limitations

The study is limited to children in one school district and the use of one student assessment tool, the AIMS Web assessment. Thus, this study will not be generalizable to other populations and other children's assessment tools.

The population of Gearing Up students was taken from a mailing to all elementary families in an urban Midwest school district in the fall of 2010. This mailing asked families to identify incoming kindergarten students for the school district in the fall of 2011 and explained that children were eligible to participate in the Gearing Up for Kindergarten

study. Elementary families receiving the letter could also pass the letter on to other families with 4-year olds who would be attending kindergarten in the same school district during the fall of 2011. Participation in the Gearing Up program and the control group study was voluntary, and the population sample may not represent the true pre-kindergarten population.

Parent education and income levels may vary significantly and, thereby, influence parenting practices with their children. Because not all recipients of the recruitment letter responded, participants in this study may not represent an equal proportion of income levels for the studied city. Due to the length of the study, a significant number of participants available at the start of the Gearing Up program may have been unavailable or unwilling to participate in the final stage of assessment.

#### Delimitations

There was an unusually short time between receiving the names of elementary families in the school district and the beginning of the Gearing Up for Kindergarten program. Therefore, participants were chosen by convenience sample and not random assignment.

This study was conducted in only one school district. Therefore, the results may not be generalizable for all incoming kindergarten populations.

#### **Organization of the Study**

Chapter 1 provides an overview of child development and the importance of parental involvement in children's school readiness, the Statement of the Problem, Purpose of the Study, and the Significance of the Study. The Definition of Terms, in relation to early childhood education, parent role, and assessment tools used in the study, is addressed along with the Limitations of the study. An outline is provided for the organization of the study. Chapter 2 examines the literature related to this study in the area school readiness and the parent's role in contributing to a child's readiness skill set. The core areas of literature include

- Theories of development, transition, and environmental influence
- Early childhood education and development
- Expected school readiness skill sets
- Parental involvement in children's education and development
- School connection with parents who foster relationship and involvement in education
- Increased school readiness expectations for students
- A parent child program model

Chapter 3 presents the research methods used in the study and describes the pilot study, researcher's role, participants, assessment instruments, limitations, validity and reliability, study design, ethical consideration, and data collection. Chapter 4 presents the quantitative findings of the study. Chapter 5 presents a summary of the findings and recommendations for future research and implementation of the Gearing Up for Kindergarten program.

#### **CHAPTER 2. LITERATURE REVIEW**

## Introduction

Kindergarten readiness is a complex issue comprised of many factors. The strength of each factor helps to determine the degree of a child's readiness for kindergarten. The literature reviewed in this chapter examines five connected components and factors related to school readiness. Presented first are theoretical foundations for child development; a new theory, transition-to-school; and parent involvement. The research provides a framework for understanding the multiple facets of early childhood development, education, school readiness, transition-to-school, and parent involvement. This research includes a varied number of educational and multi-modal activities that can be integrated into daily life.

The literature presents an expected readiness skill set upon entry into kindergarten in the state of North Dakota. Skills are reviewed along with the impact that readiness has on later school achievement. This research investigates the need for high-quality, parentchild education in the years prior to kindergarten. The research examines the types of parent involvement and how schools can support meaningful parent involvement in their children's education. This section reinforces the importance of the school connection with parents and suggests avenues to increase parental involvement. The impact for the increased expectations of developmental readiness, upon entry into kindergarten, is noted and addressed to better understand the gap in educator expectations and parental understanding of school readiness. Finally, a parent/child program is examined for its connectedness to the parent involvement and support research.

# Social Constructivist Theory of Child Development

The understanding of children's development in school readiness is a key focus of the Gearing Up for Kindergarten program. Vygotsky's (1978) theory on social constructivism established a framework for appropriate developmental instruction and an understanding of children's development. His research emphasized a developmental approach that focuses on children as active participants in their learning and development. At each stage of development, children are acquiring tools as they make sense of, gain understanding of, and interact with their environment. Through interaction with people and objects, they create meaning and make sense of their world. He posits:

The course of child development is characterized by a radical alteration in the very structure of behavior; at each new stage the child changes not only her response but carries out that response in new ways, drawing on new "instruments" of behavior and replacing one psychological function by another. Psychological operations that were achieved through direct forms of adaptation at early stages are later accomplished through indirect means. The growing complexity of children's behavior is reflected in the changed means they use to fulfill new tasks and the corresponding reconstruction of their psychological processes. (Vygotsky, 1978, pp. 72-73)

Child development is a process that deems education cannot be delivered in a one size fits all manner. Children develop at varying rates and abilities. Vygotsky described children's visible operating level versus the potential level with support and guidance from an adult as the zone of proximal development. The actual development level is the mental level at which a child functions. "The zone of proximal development described those

functions that have not yet matured but are in the process of maturation, functions that will mature tomorrow, but are currently in an embryonic state" (Vygotsky, 1978, p. 86). A child's actions provide more than meets the eye; his/her activity does not represent the whole of what is happening developmentally in the brain. Kagan et al. (1995) supported stages of development as a multi-dimensional, highly variable, and culturally and contextually influenced over time; school readiness is, likewise, developed over time and influenced by the child's own characteristics and interactions with the environment.

In addition, Vygotsky cited play as an essential component of children's development. Through play, children begin with imaginary situations that closely mimic their real lives. "As play develops we see a movement toward the conscious realization of its purpose. It is incorrect to conceive of play as activity without purpose" (Vygotsky, 1978, p. 103). Play is the work of children as they master the complex work of growth and maturation in making sense of their world.

Speech development is a process that begins when a word is applied to something that has meaning. The word is used to express the meaning through sound. Before the word comes the thought. "Thought is not merely expressed in words; it comes into existence through them. Every thought tends to connect something with something else, to establish a relation between things" (Vygotsky, 1978, p. 218).

"Thought undergoes many changes as it turns into speech. It does not merely find its expression in speech; it finds its reality and form" (Vygotsky, 1978, p. 219). The two planes of speech, the internal speech, the meaningful semantic aspect of speech, and the external speech, the phonetic aspect, work together to make the unit of thought and expression. Vygotsky characterized development as "a complex dialectical process,

characterized by periodicity, unevenness in development of different functions, metamorphosis or qualitative transformation of one form into another, intertwining of external and internal factors, and adaptive processes which overcome impediments that the child encounters" (Vygotsky, 1978, p. 73).

The social constructivist theory of development pointed to the importance of children as active participants in the learning process. The interactive process of development occurs as children relate to others and initiate the learning process through "play, speech, and relationship" (*Theories of child development and learning*, n.d.).

## **Transition-to-School Theory**

An important dimension of development includes the ability to adapt to the transitions that occur in life. One adaptation is that of school readiness. School readiness is well defined in the literature, but the transition process which includes the dimension of "school knowledge and familiarity" is largely undefined or little-described to this point in the literature (S. E. Brotherson, personal communication, November 28, 2011). This dimension consists of the knowledge and familiarity acquired primarily through "environmental exposure" and "experiential learning," the functional knowledge and familiarity that allow a child to understand and negotiate the processes of learning in a formal education setting. This readiness includes knowledge and familiarity with the environment, expectations, routines, peers and adults, equipment, and other elements of the learning atmosphere in kindergarten (S.E. Brotherson, personal communication, November 28, 2011).

Bronfenbrenner (1979), in his study of human development, called transitions an integral part of

mutual accommodation between an active, growing human being and the changing properties of the immediate settings in which the developing person lives, as this process is affected by relations between these settings, and by the larger contexts in which the settings are embedded. (p. 21)

This ecological transition takes place in settings that begin small, the microsystem. As children grow, their world expands from the microsystem to also include a larger area outside of the home, the mesosystem. The mesosystem can include neighbors, churches, daycare, and other small entities with which children may interact. The mesosystem is followed by an even larger setting, the macrosytem, which encompasses the community at large and the world. An ecological transition occurs whenever a person's position in the ecological environment is altered as the result of a change in role, setting, or both (Bronfenbrenner, 1979).

Two of Bronfenbrenner's hypotheses are significant for the transition to school: Hypothesis 27 stated, the "developmental potential of a setting in a mesosystem is enhanced if the person's initial transition into that setting is not made alone" (1979, p. 211), and Hypothesis 42 stated that, "upon entering a new setting, the person's development is enhanced to the extent that valid information, advice, and experience relevant to one setting are made available on a continuing basis, to the other" (1979, p. 217).

Fabian and Dunlop (2007) noted that transition to school often does not take place until children are already in school. This theoretical framework supported a journey of preparation leading to the start of school. The transition process occurs naturally through experiences with routine, sharing, learning to follow directions and rules, and visits to kindergarten classrooms. Transition programming is often suggested as a component of

school readiness but lies in a grey zone of how, what, and who should be responsible for implementation. Both children and parents who are prepared and know what to expect in the school environment tend to experience less anxiety and greater enjoyment at school entrance. "If going through a transition is a learning skill in its own right, it is important that children build resilience to change but are also given support to help to both mark and negotiate change" (Fabian & Dunlop, 2007, p. 3).

# **Overlapping Spheres of Influence Theory**

The importance of parental involvement, as a significant contribution, in a child's life and education is supported through Dr. Joyce Epstein's Overlapping Spheres of Influence theory of education. This theory supports the combined efforts of home, school, and community for maximum growth and development in children (Epstein, 2001).

Epstein's (1987b) Overlapping Spheres of Influence model is fluid and recognizes an external and internal structure of overlapping and non-overlapping spheres that represent family, community, and school. Forces, time, experience in family, and school experience influence the degree of overlap and connection.

Force A (Epstein 1987b) represents a developmental time and history for students, families, and schools. Home is a child's first environment and is often a separate sphere where parents are the primary influence. The parent-child instruction, although not formal, occurs in that early setting. Parents apply their knowledge of child rearing, their understanding of school readiness, their own school experiences, and other information from outside sources to educate their children. The spheres of school and community begin to formally overlap as children enter school (Epstein, 1987b).

Force B and Force C (Epstein, 1987b) are the experiences and pressures encountered as the learning environment evolves and changes. Parents' involvement or lack of involvement in their children's school education directly influences the overlap of the spheres.

Epstein (1987a) also promoted a shared responsibility with each segment of the sphere; the child is at the center as the object of focus. Each entity works together at times, and separately at times, in a role that supports the child. Parents are a child's primary connection, and the first type of parent involvement, as supported in the six types of parental involvement, relates directly to care of the child.

As the child enters school and the community, a shared partnership develops. The shared partnership enlarges the scope of support available for the child's education and growth. The Overlapping Spheres of Influence model works to support cooperation and collaboration between school and home (Epstein, 1987b). Piotrkowski, Botsko, and Matthews (2001) cited the importance of schools and communities working together to prevent failure in school and said that parents share the responsibility for educating their child along with the education community

Maximum benefit is gained when schools and families work closely together, in a cooperative manner, to assist the child with the education journey. "Children are connected to the same families but different teachers over their school years" (Epstein, 1987b, p. 128).

At the neighborhood level, school readiness resources include affordable, high quality child care and preschool for all; well-stocked libraries welcoming to children and parents; safe playgrounds and streets, and so forth (Shore, 1998). Local school readiness resources include strong, accountable leadership; transition

programming and parent involvement activities, on-going professional development and support for teachers; high quality individual instruction and so forth. Family readiness resources include a rich literacy environment, nurturing parenting, financial resources, and social support for child rearing. Ideally, these resources are integrated to facilitate the optimal development of each child. (Piotrkowski et al., 2001, p. 540)

#### Early Childhood Learning and Development

The first five years of a child's life are generally regarded by families, researchers, and practitioners as critical years for the development of foundational skills and competencies that support continuous, lifelong learning that allows children to reach their full potential in development and life (National Governors' Association [NGA], 2005). The NGA task force in its report on school readiness, *Building the Foundation for Bright Futures,* wrote:

Getting children ready to succeed in school begins at birth. . . . Children are born learning. The first years of life are a period of extraordinary growth and development. During this time the brain undergoes the most rapid development as neurological connections (synapses) are made at incredible rates that are reinforced and solidified or lost through attrition over time. Development in young children is continuous. (National Governors Association, 2005, p. 11)

The early and rapid development of the brain and related neurological functions during this period is well documented and researched (Shonkoff, 2007). "Early experiences actually influence brain development, establishing the neural connections that provide the foundation for language, reasoning, problem solving, social skills, behavior and emotional
health" (Rhode Island KIDS COUNT, 2005, "Executive Summary"). "The first five years of a child's life are marked by a significantly accelerated period of growth and development" (North Dakota Department of Human Services, 2008a, p. 4). This time of significant growth and development is critical to later success as they begin school and experience life outside the home. Research demonstrates there are "windows of opportunity" for the developing brain (North Dakota Department of Human Services, 2008b, p. 4; Sprenger, 2008, p. 15).

The North Dakota Department of Human Services' (2008a) early education guidelines also supported the important roles that family and community play in healthy child development. These guidelines work to aid parents and caregivers as they prepare children for school. Families that are economically stable, understand child development and care, and have positive relationships are more likely to experience success as children enter kindergarten. Children develop best in an atmosphere of care, love, and support that also provides opportunities to explore and learn (Fielding et al., 2004; National Association for the Education of Young Children [NAEYC], 2009; NGA, 2005; North Dakota Department of Human Services, 2008b; Rhode Island KIDS COUNT, 2005; Sprenger, 2008; Wesley & Buysse, 2003). Parents are their child's first and most important nurturer and teacher, and parents should be supported by family, community, friends, childcare, health care, and educators (National Scientific Council on the Developing Child, Center on the Developing Child, 2007; North Dakota Department of Human Services, 2008b).

The brain, as noted by Sprenger, (2008) is the only organ in the body shaped through its interaction with the environment. Because the brain is the organ dedicated to learning and memory, educators and parents must understand the importance of the

developmental stages and conditions that coexist for maximum development in a child's learning journey.

Dr. Marilee Sprenger (2008), noted researcher and educator in the area of brainbased learning, supported general principles that guide a child's developmental learning: (a) every brain is totally unique; (b) emotions guide our learning; (c) stress affects learning; (d) there is a brain-body connection; (e) the brain has multiple memory systems and multiple modalities; (f) the brain seeks meaning and relevance; (g) the brain learns through experience; (h) the brain is social; (i) the brain learns in patterns; and (j) the brain grows through enrichment (Sprenger, 2008, pp. 2-5).

Dr. Sprenger's work emphasized the importance of interacting and teaching children from birth to help them learn to make sense of their environment and to maximize the potential in every child. "According to the National Association for the Education of Young Children, early childhood begins at birth. The achievement gap can also begin then" (Sprenger, 2008, p. 5).

Supporting Dr. Sprenger's research, The National Association for the Education of Young Children (NAEYC, 2009) listed very similar principles which also include additional realms of development, beyond the brain, for best practices in raising children: (a) development in one area is closely related to development in other areas; (b) skill development begins with basics and progresses to the complex; (c) development does not occur at even steps or intervals and can vary among children; (d) personal experience has an effect on development; (e) development is enhanced when meaning is attached to the learning; (f) play is important for children's growth and development; (g) children need many opportunities to acquire skill and greater proficiency; (h) children express their

knowledge through multiple modalities; and (i) children thrive and develop in a safe environment where they are loved and valued (NAEYC, 2009, pp. 11-16).

# National Education Goals Panel Early Childhood Development Dimensions and the National Governors' Association

In congruency with Sprenger (2008) and the NAEYC (2009), the National Education Goals Panel (NEGP, 1995) identified five dimensions of developmental early learning to support school readiness. The NEGP recognized the well-being of America's young children is a shared responsibility of family, society, and educators and only by working collaboratively across sectors and institutions, will America be able to realize its "readiness" vision identified in Goals 2000. The NEGP supported five dimensions of early childhood development and learning important for school readiness and success. These five dimensions work to affirm the connection between early development and learning, and children's later success in school and in life (NEGP, 1995).

The National Governors' Association, in its updated 2005 report taken from *The National Education Goals Report, Building a Nation of Learners*, reinforced support for early childhood education and learning. The report lists multiple areas but focuses on five key domains to help children and parents with school readiness: physical well-being and motor development, social and emotional development, approaches toward learning, language development, and cognition and general knowledge.

# **Physical Well-Being and Motor Development**

Kagan, Moore, and Bredekamp (1995) reported a link to children's health and their performance in school. Low birth weight, as well as improper nutrition, can greatly influence children's learning and development. It is important for children to have

opportunities to develop gross motor skills through outdoor physical activity and to have the opportunity for fine motor skills through simple tasks: buttoning a shirt, holding a crayon, or mastering a puzzle.

# Social and Emotional Development

The second domain presented by Kagan et al. (1995) reported children's earliest family social experiences are the basis for building relationships with friends and teachers. Stable home relationships aid children in developing a sense of self and well-being that will guide them as they enter school and become an integral part of the school community.

## **Approaches Toward Learning**

Kagan et al. (1995) cited children as unique, and their approach to life varies greatly. It is important to allow children to explore and learn in a safe and accepting environment that encourages exploration and discovery. Experiences should help children discover aptitude for certain activities along with likes and dislikes. Children, when given the opportunity to explore, examine things of curiosity, discover independence, and begin to develop a sense of self and self-confidence. Through the growing and learning process, children begin to develop an understanding of who they are and a sense of belonging.

#### Language Development

Kagan et al. (1995) emphasized the importance of language development in helping children make stronger connections with their world. As language develops, it helps children connect with others and provides tools for relating and expressing thoughts, feelings, wants, and needs. Language development is also an important link to early literacy and is foundational to the reading experience.

# **Cognition and General Knowledge**

The fifth domain cited by Kagan et al. (1995) represents a full palette of incoming information and how children process new information to make meaning and to apply the information to their lives. Children are continuously learning and adding new information; as that process moves forward they begin make connections, understand relationships, and develop problem-solving skills. General knowledge acquisition presents itself in the practical realm of self-care and ability to dress and toilet, adapt to routine and to change, and to begin contributing through basic chores and family interactions.

These five domains are intrinsically interrelated and dependent upon each other. Development in one area will affect development in others. Inherent in this perspective is the belief that there is no single or uniform "standard" of readiness. Development varies from child to child, thus no child will demonstrate all necessary skills in every area. The goal is for children to develop competencies across all five domains with the understanding that no one dimension is more important than another (High/Scope Educational Research Foundation, 2006).

North Dakota's *Early Learning Guidelines* and development indicators complemented the research of Sprenger; the NAEYC's early learning guidelines, and the NGA research to reinforce the NEGP dimensions of early learning and goals for school readiness. North Dakota's development indicators progressed from birth to age five, provided a framework for further development and understanding of school readiness, and prescribed a common set of developmentally appropriate expectations for the early years of growth and development (North Dakota Department of Human Services, 2008a; 2008b).

The state of North Dakota segmented early learning guidelines into two groups, birth to age 3 and ages 3 through 5 (North Dakota Department of Human Services, 2008a; 2008b). Birth to age 3 was divided into four domains: (a) Social and Emotional Development, (b) Language Development and Communication, (c) Cognitive Development, and (d) Physical and Motor Development. Each domain of development is related to and influences the others. Four of these domains are closely matched with the National Education Goals Panel's dimensions of development and school readiness: Social and emotional development, Language development, Cognition and general knowledge, and Physical well-being and Motor development.

"The *social and emotional* domain included the development of trust, emotional security, self-awareness, self-regulation, and the beginning of relationships with other children" (North Dakota Department of Human Services, 2008b, p. 15). *Social and emotional development* is a foundational aspect of children's growth and development; children who trust and feel secure have greater opportunities for meaningful relationships and experience a greater ability to self-regulate. Shonkoff (2007) called emotional well-being and social competence the "bricks and mortar" of children's growth and development.

The *language development and communication* domain is important in the development of children's expressive and verbal skills. Language development encompasses multiple senses: listening, speaking, the process of understanding, and the beginning of early literacy (North Dakota Department of Human Services, 2008b).

The North Dakota Department of Human Services (2008b) noted the *cognitive development* domain represents children's thinking and how they make sense of the world

around them. As children grow and develop physically, their sense of imagination, the complexity of their play, and their ability to make connections with cause and effect also grow towards higher and more complex thinking

*Physical and motor development*, as described by the North Dakota Department of Human Services (2008b), was characterized by tremendous early growth. By the end of the first year, most children have learned to roll over, crawl, begin to climb, and may be in the early stages of walking. As children move about, they experience increased coordination and have greater control over muscles. Strong physical growth is essential to the exploration and discovery that allows children to explore their environment.

North Dakota Early Learning Guidelines: Ages 3 Through 5 (North Dakota Department of Human Services, 2008a) provided a general framework of developmental mastery for children as they enter kindergarten. All domains are important, but it is equally important to note children develop at their own pace and that "no domain is more important than another" (North Dakota Department of Human Services, 2008a, p. 6).

There are eight domains for ages 3 through 5: (a) Health and Physical Development, (b) Social and Emotional Development, (c) Approaches to Learning, (d) Expressive Arts and Creative Thinking, (e) Language and Literacy, (f) Mathematics and Logical Thinking, (g) Science and Problem Solving, and (h) Social Studies (North Dakota Department of Human Services, 2008a).

The *health and physical development* domain listed by the North Dakota Department of Human Services (2008a), for ages 3 to 5, described greater expectations for children's stamina: their ability to complete a task and interact with balls, puzzles, or other objects connected with movement. Children should experience a greater skill level with writing, kicking a ball, and other skills as muscles develop and grow stronger.

The North Dakota Department of Human Services (2008a) expected children at this stage to begin to learn how to dress themselves, tie shoes, master zippers, and do other tasks once foreign to them. This domain also included an understanding of hunger and fullness and knowledge of the body parts and their roles

Preschool children are naturally active; they have a built-in drive for motor activity. They tend to explore their environment by checking out all the corners of the room or playground, working to achieve physical closeness, and seeking out communication with others. These activities and actions are all essential for proper cognitive, emotional, and physical development. "When we allow children to run, play, and explore in a safe environment, they will naturally get the motor activity they need" (Poole, 2000, p. 41).

The *social and emotional development* domain cited by the North Dakota Department of Human Services (2008a) supported the development of children's selfconcept through healthy relationships and interactions with parents and caregivers. Children become connected through important daily activities: playing, sharing, and through learning to express themselves appropriately.

The North Dakota Department of Human Services (2008a) referred to *approaches to learning* as a child's temperament and tendencies in learning styles. This domain accounts for variety of personality types, variances in learning styles, and patterns and norms in cultures. This domain places an emphasis on the implementation of multiple avenues of learning and instruction.

The North Dakota Department of Human Services (2008a) defined the *expressive arts and creative thinking* area of development to encompass an important added dimension of life that enhances the quality of children's learning and their environment. This area of growth can include dancing, music, and exposure to drama and the visual arts.

The *language and literacy* area of development is continued from the birth to 3 framework by the North Dakota Department of Human Services (2008a). Healthy children continue to learn more and better ways of expressing themselves; demonstrate a greater understanding and use of language, both expressive and receptive; and acquire pre-reading literacy skills.

The North Dakota Department of Human Services (2008a) noted that, as children develop and grow, they will demonstrate progress and growth in language and literacy skills as demonstrated through communicating with words and writing, creating ideas, learning to listen to stories, retelling a story, asking questions for clarification, and listening to and identifying sounds. Familiarity with books and their purpose is also a major characteristic of language and literacy development.

Strategies that have been shown to be effective at promoting children's early literacy development include reading aloud to children, fostering children's understanding of print concepts; arranging the environment for children have an opportunity to interact with books and other print materials; providing opportunities for children to experiment with writing; familiarizing children with letters of the alphabet and their corresponding sounds; and involving children in activities that promote phonological skill and development (Green, S.D., Peterson, R., & Lewis, J.R., 2006, para. 8-9).

Green, Peterson, & Lewis (2006) substantiate reading teachers' beliefs; reading to children is the most important aspect in helping them learn to read. Parents who read to children expose them to a multitude of learning with one simple activity. Stories help children to understand print has meaning, letters make sounds, and letters together make words. Through the reading experience, children enjoy closeness with the reader and absorb vital information that connects with learning to read.

The *mathematics and logical thinking* domain of development focused on the area of children's logic development and a continually growing sense of mathematical knowledge, which included number sense, the ability to see patterns, measurement, and accounting (North Dakota Department of Human Services, 2008a). A child's sense of mathematical thinking begins as early as six months of age (Fielding, 2009).

The North Dakota Department of Human Services (2008a) guidelines created for child development noted that, as children develop, they begin demonstrate greater understanding of math concepts. Children's development in math begins with counting and then recognizing that one of something has meaning. Continued development and growth of math concepts also includes time sense, recognition of coins as something that have value, understanding colors and shapes, and increasing problem-solving ability.

Young children often have a spontaneous and explicit interest in mathematical ideas. Naturalistic observation has shown, for example, that in their ordinary environments, young children spontaneously count, even up to relatively large numbers, like 100, and may want to know what is the 'largest number.' Also, mathematical ideas permeate children's play: in the block area, for example, young children spend a good deal of time determining which tower is higher than another,

creating and extending interesting patterns with blocks, exploring shapes, creating symmetries, and the like.

"Play provides valuable opportunities to explore and undertake activities than [*sic*] can be surprisingly sophisticated from a mathematical point of view" (North Dakota Department of Human Services, 2008a, p. 38).

The *science and problem solving* domain of development (North Dakota Department of Human Services, 2008a) encompassed children's understanding of the world around them. They begin to wonder the "why" questions and often ask the same question repeatedly. This important aspect of development is the early beginning of understanding force, gravity, and other scientific phenomenon they will learn later school.

The *social studies* domain of development (North Dakota Department of Human Services, 2008a) supported children's growth in understanding the various environments to which they are exposed: daycare, home, church, other family, and neighborhood. Through interaction with different people and environments, children learn appropriate social skills and accepted behaviors. Established cultural norms become a part of who they are and provide children with a sense of place and belonging.

## Synthesizing Early Childhood Development

Early childhood research terms have many common identifiers that create a general framework of early development guidelines and competencies. Sprenger (2008) and the NAEYC (2009) noted common developmental dimensions and ideas, although stated in different language: Each brain is unique, and children display different strengths and varying rates of development. Emotions guide learning, and the learning environment greatly influences the early childhood window of learning, with children in poverty often

experiencing delayed development (NAEYC, 2009; NGA, 2005; Rhode Island KIDS COUNT, 2005; Sprenger, 2008). Children learn in stages of development and patterns with prior experiences giving the understanding and support to new learning (Sprenger, 2008). The recognition of a brain-body connection is noted in children's physical action and in play (Sprenger, 2008). Challenge builds higher skill levels and allows children to use prior learning as the new skill set is developed. Learning happens best in a community and in a safe environment that fosters a sense of value and emotional security (NAEYC, 2009; North Dakota Department of Human Services, 2008a; 2008b; Sprenger, 2008).

The common readiness indicators supported by this author's research suggest a strong thread of emphasis that supports a focus on the five domains of school readiness: Physical well-being and motor development, Social and emotional development, Approaches toward learning, Language development, and Cognition and general knowledge (NEGP, 1991; NGA, 2005; North Dakota Department of Human Services, 2008a; 2008b; Rhode Island KIDS COUNT, 2005).

Wesley and Buysse (2003) cited the importance of children's emotional health as a key factor in readiness. Children in poverty may have independence skills as a result of being left alone to care for themselves, whereas a child from a nurturing, two-parent home may never have had to concern himself/herself with basic survival skills. One would be independent, the other not necessarily so. Does that independence constitute emotional health or emotional neglect? The social aspect of Wesley and Buysse's report recommended that children attend some type of high-quality, developmentally appropriate preschool to assist with school readiness to support social development and interaction.

The report also emphasized the importance for a shared collaboration of family, educators, and community members in socializing and educating children.

High; the Committee on Early Childhood, Adoption, and Dependent Care and the Council on School Health, in *Pediatrics* (2008), cited the responsibility of schools to be ready for students at all levels of development. A child's early learning environment can significantly affect school readiness and later school success. This report listed developmental factors for readiness, including social, emotional, cognitive, and physical development. This report cited early brain and child development as modifiable factors in a child's early experiences that can greatly affect a child's learning trajectory. Many children in the United States enter kindergarten with limitations in their social, emotional, cognitive, and physical development that might have been significantly diminished or eliminated through early identification of and attention to child and family needs. (High et al., 2008, p. 1008)

The report concluded with three conditions for children to enter school ready to learn: intellectual skills, motivation to learn, and strong social-emotional capacity and support (High et al., 2008).

Raver and Knitzer (2002), in their report *Ready to Enter*, linked the connection to early emotional development to academic learning. Children at risk often enter school with poor social, emotional, and behavioral development. At-risk children often demonstrate other inappropriate anti-social behaviors which often affect their academic performance negatively. This report recommended early interventions and education for teachers and child-care providers in helping them learn to deal with difficult behaviors.

#### Linking Early Childhood Development and School Readiness Expectations

The early years are an important period during which children acquire the basic skills that serve as the foundation for later learning (Fielding et al., 2004; NAEYC, 2009; Sprenger, 2008). Moreover, these years are the time when parents' beliefs about their children's abilities are shaped and when children's own academic self-concepts begin to form. When parents value learning and education, they set the stage for their children to value learning, too (Epstein, 1987a). Young children who are active learners in their early years will more likely enter kindergarten ready to continue the learning process and experience greater success in school (Fielding et al., 2004; NAEYC, 2009; NGA, 2005).

The NAEYC supported appropriate developmental and academic learning in preschool by incorporating play as the major vehicle for learning. "Research demonstrated the links between play and foundational capabilities such as memory, self-regulation, oral language abilities, social skills, and success in school" (NAEYC, 2009, p. 14). Today, academics play a stronger role in school readiness, and parents and educators are learning to finesse the fine balance of instruction with appropriate amounts of play. Play has a greater value beyond entertainment; it is an important means of learning for children, too. Researcher and parent educator Michael Popkin (1996) emphasized fun while teaching your child:

A four year old should be able to sit for 20 minutes with a book and know that people read from left to right and that words and letters produce different sounds. He should be able to tell rhymes and sing songs. (p. 6)

Play is the umbrella under which the best early learning takes place and is a vital part of early childhood education and development. When children explore, experiment, and

interact through play, they learn how the world works and begin to develop a sense of their place in the world (NAEYC, 2009). Part of that development is caring and knowledgeable adults who are guiding and ensuring appropriate learning experiences through play and other modalities of discovery (NGA, 2005). Sprenger (2008) and Weiss et al. (2006) reinforced the importance of parent participation in child-centered activities, specifically play. Play is also important for children's social and emotional development. Children who play at home and whose parents understand the importance of play in development are more likely to demonstrate prosocial and independent behavior in the classroom. In addition, parent participation with their children in activities such as arts and crafts is associated with children's literacy development.

The NAEYC (2009) and the North Dakota Department of Human Services (2008a) also supported developmentally appropriate practice as a significant aspect of early childhood education. Research in human development and learning emphasized play among other strategies. Using developmentally appropriate practice, teachers intentionally meet individual children at their particular stage of development and provide challenging, achievable goals to promote the development and learning that support school readiness.

The North Dakota Department of Human Services (2008a) supports play as a most important early learning tool. School readiness, like play, is understood to be important, but has recently been reformed to a different understanding and meaning. How did the concept of school readiness change so significantly in a relatively short period of time?

#### **Changes in School Readiness Expectations**

Kindergarten has long been known for its socialization practice and introduction to letters and writing. The current educational standards are far beyond what most people

remember from their kindergarten days. The basic socializing and introduction of letters that once began in kindergarten now begins in preschool (Fielding et al., 2004).

In the past, educators primarily used a maturational viewpoint of school readiness with a set age for kindergarten entry (Zigler, Gilliam, & Jones, 2006). The general school-readiness guidelines today are the result of a journey of progress in understanding child development and achievement rather than simply stating a set age of readiness. The guidelines are also a result of the NEGP establishment of school-readiness goals in 1995: "All children will start school ready to learn" (Wesley & Buysse, 2003, p. 352). Just what did "ready to learn" look like in the past, and what does "ready to learn" look like today?

# **Teacher Perspectives on Education Expectations**

In 1993, a school-readiness survey of 1,339 kindergarten teachers showed that 75% felt the top three readiness attributes were for a child to be (a) physically healthy, rested, and well nourished; (b) able to communicate his or her thoughts and needs in words; and (c) curious and enthusiastic in his or her approach to new activities. More than half of the teachers in this study also indicated that readiness included not being disruptive, being sensitive to other children's feelings, and being able to take turns and share. Less than 10% thought that being able to count to 20 or more or knowing the letters of the alphabet were important in terms of kindergarten readiness (Ackerman & Barnett, 2005). Similar findings in a study by Fabian (2002) reported teachers' perceptions of readiness: "the ability to be part of a large group competing for the attention of one adult; the capacity to concentrate; to be self-sufficient; use their initiative and sit for long periods of time" (Fabian & Dunlop, 2007, p. 10).

Ackerman and Barnett (2005) reported from the Early Childhood Longitudinal Study–Kindergarten Class of 1998-1999 (ECLS-K) study about the importance of nonacademic readiness skills for kindergarten teachers. Specific academic tasks–such as using a pencil, knowing the names of the colors and shapes, recognizing letters, or counting to 20 or more–were likely to be rated as essential or very important by less than one-third of the teachers. Seventy-five percent of the 3,305 kindergarten teachers sampled in this study regarded being able to follow directions and to communicate needs and thoughts, as well as not being disruptive, as essential or very important readiness skills.

Ackerman and Barnett (2005) described a 1993 study where parent perceptions of readiness were focused on a positive attitude towards school and good social skills. Parents' thoughts regarding academic readiness differed from the teacher's perceptions of readiness; 58% of the parents thought their children should be able to count to 20 and know their letters.

A more recent survey of kindergarten teachers and parents (PNC Financial Services Group, Inc., 2007) also confirmed the differing perceptions that parents and teachers held about readiness skill development upon entry to kindergarten. Approximately 80% of the parents felt their own children were well prepared in social-skill development while only 15% of the teachers agreed. These perceptions held for academic skills as well with approximately 70% of parents indicating their child was very well prepared academically and just 15% of the teachers agreeing. Overall, that report indicated that only 25% of the parents and 7% of the teachers described children in the United States as being very well prepared for learning upon entry into kindergarten (PNC Financial Services Group, Inc., 2007).

Fielding (2006) reported that, in a short period of time, there has been a significant shift in school-readiness expectations. The April 2006 edition of the *American School Board Journal* identified typical characteristics for incoming kindergartners in the United States. Targets expectations for literacy were as follows: Children enjoy being read to and can retell a story, know 12-15 upper- and lowercase letters and their sounds, memorize 5-6 nursery rhymes, hear ending sounds (rhyme) and beginning sounds (alliteration) in words, speak in complete sentences, and have a vocabulary of 4,000 to 5,000 words. Target expectations for math and social development included counting from 1 to 20, one-to-one correspondence with numbers, and settling into new groups or situations.

Because school readiness is a complex and ever-evolving issue, a basic definition of school readiness is important. "School readiness," according to Piotrkowski (2000),

is the set of skills and attributes that a child needs at school entry in order to profit from the kindergarten experience and meet societal expectations of competence in the classroom. Rather than learning readiness, educational readiness, or academic readiness, which are more abstract in focus, school readiness implies being ready for adequate functioning in a tangible location – school. (Zigler et al., 2006, p. 29) In the Rhode Island KIDS COUNT (2005) report, school readiness was summarized:

School readiness encompasses children's curiosity and enthusiasm for learning, their physical and mental health status, their ability to communicate effectively, their capacity to regulate emotions, and their ability to adjust to the kindergarten classroom environment and cooperate with their teachers and peers. Ready children are those who, for example, play well with others, pay attention and respond

positively to teachers' instructions, communicate well verbally, and are eager to participate in classroom activities. They can recognize some of the letters of the alphabet and are familiar with print concepts (e.g., that English print is read from left to right and top to bottom on a page and front to back in a book). Ready children can also identify simple shapes (e.g., squares, circles, and triangles), recognize simple-digit numerals, and of course, count to ten. (pp. 11-12)

School readiness has also been redefined by changes in education law throughout the course of time, especially in the last 20 years. Recent changes in education law (No Child Left Behind Act of 2001) resulted in a demand for greater accountability in student achievement and more stringent learning expectations at younger ages. This process has resulted in states responding with greater demands on educators and a clearer set of expectations and standards for early learning.

In 1994, President William Clinton signed the Goals 2000: Educate America Act. This federal mandate was designed to ensure all children would begin school ready to learn. Following the passage of Goals 2000, President Clinton created the National Education Goals Panel (NEGP) to monitor the progress of the education goals (Kagan & Rigby, 2003). The NEGP (1995) assigned the readiness goal to a subgroup that created a definition of school readiness encompassing the whole child. Domains of readiness were divided into five categories: (a) physical well-being and motor development, (b) social and emotional development, (c) approaches to learning, (d) language development, and (e) cognition and general knowledge (NEGP, 1995).

Fielding et al. (2004) cited the education law, No Child Left Behind (2002), which further increased the requirements for student growth and achievement. No Child Left

Behind (NCLB), in effect, created a 95% reading and math goal for all students and schools, stating that, by the year 2014, 100% of students must achieve minimum standards in reading and math. All 50 of the United States responded with legislation to increase school accountability for that achievement, and many states have responded by increasing their academic standards.

The National Governors' Association (NGA; 2005) cited the 2002 NGA project which used NEGP's work and established a gubernatorial task force on school readiness to determine what states could do to advance school readiness. The NGA (2005) task force incorporated the following principles into its school-readiness recommendations: (a) family is integral to a child's life; (b) school readiness is an adult responsibility; (c) the first five years of life are critical; (d) the five dimensions of school readiness are equally significant; and (e) no single school-readiness plan can meet every state's needs. These five principles formed the NGA's school-readiness framework of Ready States, Ready Schools, Ready Communities, and Ready Children. The work of both groups broadened the definition of school readiness to include both child readiness and environment readiness.

Wesley and Buysse (2003) agreed with educators and believed parents and communities are responsible to produce a child ready for school. In addition, they wanted a greater responsibility for parents and communities in preparing children for school. Most of the educators felt that the responsibilities of the school did not begin until the day a child began school. Their viewpoint reinforced the school's lack of involvement with the community in supporting preschool children and their families toward school readiness. The eighth national education goal, however, mandated school collaboration with parents to encourage early learning and development. Without a source of consistent

communication and information, parents reported little understanding of the schools' expectations regarding school readiness (Wesley & Buysse, 2003).

Wesley and Buysse (2003) further reported how the NEGP's understanding of readiness to learn hinged on a range of factors, including a child's health and physical development, social and emotional development, approaches to learning, language and communicative skills, and cognition and general knowledge. Efforts to improve school readiness, therefore, should begin long before children enroll in kindergarten. No one entity should bear all the responsibility to prepare a child for school; communities should collaborate to support families, educate parents, and raise the quality of early care and education. Helping all children to begin school "ready to learn" is the shared responsibility of all institutions in a community.

#### **North Dakota Readiness Expectations**

The North Dakota Department of Public Instruction's readiness skills for an incoming kindergartner are for him/her to have a vast general knowledge: how to write his/her first name, recognize most of the letters of the alphabet and their sounds in isolation, understand rhyming words, count to 20, count backwards from ten to one, and demonstrate a one-to-one correspondence with numbers to ten. Children should also be familiar with books; they should know top-to-bottom print, understand left-to-right paging, and be able to know books tell a story (Department of Public Instruction, 2005a). The Department of Public Instruction standards combined with the North Dakota Department of Human Services' *Early Learning Guidelines* make it clear that children entering kindergarten in North Dakota should have a readiness skill set of growth and learning in all realms of development.

#### **Transition to School**

Families have their children for a lifetime; educators have the children for one academic year. Educators can be a link for parents to assist with transitions in educational settings. When educators come alongside parents with information, connection, and support, it can ease tension and stress for both parents and children (Weiss et al., 2006).

Weiss et al. (2006) cited a growing consensus in research that supports a responsibility between early childhood settings and elementary schools to sustain the family involvement link through the transition from preschool to the public school setting. It is at this critical time schools should maintain rigorous efforts to connect and reach out to parents in this transitional period. Schools that provide multiple opportunities through the transition period enjoy increased levels of family participation.

It is important that local communities establish processes for consensus-building around common readiness goals for young children, develop continuity between preschool and kindergarten–including but not limited to integrated curricula–and assume joint responsibility for ensuring that all children be ready for success when they enter kindergarten. (Piotrkowski et al., 2001, p. 555)

Transition practices cited in Piotrkowski et al. (2001) included preschool and kindergarten teachers visiting each other's classrooms as well as meetings with early childhood administrators and elementary school administration. These interactions would promote the understanding of barriers, help clarify expectations, and provide the basis for transition planning that best meets the needs of individual families. This study suggested that the responsibility of transition practice resided with the schools, not the parents.

Piotrkowski et al. (2001) expected schools to improve the readiness of young children by making connections with local child-care providers along with preschools and by creating policies that ensure smooth transitions to kindergarten. Children entering kindergarten vary in their opportunities for experiences, skill development, acquisition of knowledge and language. In addition culture, and family background can vary greatly, and schools must be ready to address the diverse needs of the children and families in their community and must be committed to the success of every child.

Fabian and Dunlop (2007) cited steps schools can take to reach out to future and current families. Schools that provide families with logistical information regarding procedures, routines, and academic expectations help to reduce fears and transition difficulties. Schools and families who partner together experience greater satisfaction in relationship and smoother student transitions to school.

## **Home-School Contact and School Readiness**

Fielding et al. (2004) believed that parents need a clear understanding of childhood development and academic expectations in order to begin to recognize the importance of their role as educators during their children's preschool years. The home is the child's first classroom and continues to have a great impact on the child throughout life.

The research of Hanson (2008), the North Dakota Department of Human Services (2008a), Patrikakou et al. (2005a); and Schulting (2008) reinforced the importance of early childhood programs in recognizing the needed the support of families and communities to best educate children. To acquire this support, there is a need to encourage the family-involvement processes that research has shown to be effective in encouraging children's learning and socio-emotional development. Parents' beliefs and competencies greatly shape

the home and learning environment of their young children. In the early childhood years, the home-school relationship refers to the formal and informal connections between families and their young children's educational settings. Both participation in preschoolbased activities and regular communication between families and teachers are related to young children's outcomes. Parents who maintain direct and regular contact with the early educational setting experience fewer barriers to involvement and have children who demonstrate positive engagement with peers, adults, and learning. In addition, teachers' perceptions of positive parental attitudes and beliefs about preschool are associated with fewer behavior problems as well as higher language and math skills among children. A positive home-school relationship can buffer some of the negative impacts of poverty on academic and behavioral outcomes of poor children (Schulting, 2008). Early childhood education often has the benefit of a home-visit component, thus strengthening the parentschool relationship. Epstein (1987a) emphasized the importance of including parents in a wide variety of activities, which keep parents connected with their children's learning and with teachers.

Fielding (2009) and Sprenger (2008) shared strategies for educational parentinvolvement activities that can be as simple as inviting parents to school to share in classroom activities with their children. This kind of experience can directly influence parent practices at home with their children. When parents have a sense of what denotes competency, they are more likely to get involved. In early childhood, the learning emphasis is most likely to be on language along with literacy support and skills.

Christenson and Sheridan (2001) and Gestwicki (2004) provided ideas outside the investments of time or money; simple activities like Play Dough letter building; counting

numbers talking about least to greatest; and other simple, but effective, early childhood activities are helpful. Everyday classroom materials and lessons can be a spark for parent applications of learning in the home. Teacher interactions with learning-based questions serve to provide a basis for parents to ask learning-related questions and for teachers to provide insights for parents about the capacity of their children for learning.

Fantuzzo and his colleagues (2004) recently showed that practices associated with responsibility for learning (e.g., providing a place for educational activities, asking a child about school, reading to a child), above and beyond the aspects of the home-school relationship, are related to children's motivation to learn, attention, task persistence, and receptive vocabulary and to fewer conduct problems. (Harvard Family Research Project, 2006a, p. 4)

The Harvard Family Research Project (2006a) findings demonstrated the effectiveness of educators who work with young children in promoting family-involvement opportunities. Workshops, trainings, and parent-child groups helped to promote warm and nurturing relationships outside the classroom. The research recommended teachers communicate often with parents, in a variety of modes, regarding their children's learning. Teachers shared ideas to help student progress, lesson content, and ways to find patterns in learning. Teachers who provided opportunities for parents to visit the classroom, both to participate and observe, helped to contribute to a greater sense of belonging for both children and parents. Frequent and ongoing communication and connection were recommended as a regular part of teacher practice. The important aspect of family history was emphasized to teachers for effectiveness in communicating with families. Educators were encouraged to take initiative in connecting and relating to parents. Preschool

administration was also advised to initiate relationships with elementary schools and bring in personnel to ease the transition process for parents and children.

Children will not enter school ready to learn unless families, schools and communities provide the environments and experiences that support the physical, social, emotional, language, literacy, and cognitive development of infants, toddlers and preschool children. Efforts to improve school readiness are most effective when they embrace the rich cultural and language backgrounds of families and children. Research has provided great insights into children's development and numerous strategies to assist learning. The strongest effects of high quality early childhood programs are found with at-risk children—children from homes with the fewest resources and under social and economic stress. (Rhode Island KIDS COUNT, 2005, "Executive Summary")

Zigler et al. (2006) cited the Chicago Longitudinal Study that showed parental involvement in child activities, both in the home and at preschool, boosted school readiness and that parent participation had positive effects on children's achievement and behavior, grade retention, dropout rates, and juvenile delinquency. Parental involvement in preschool was also associated with greater reading achievement and less grade retention through eighth grade. Parents who were involved in preschool education with their children were also more likely to be involved in their children's elementary school

Parent involvement, beginning in preschool, influences children's academic success through four interrelated pathways: (a) Parent involvement in preschool is linked to involvement in supportive activities at home. (b) Parent involvement in preschool and at home has direct effects on children's concurrent academic performance. (c)

Parent involvement in preschool and at home has effects on children's social behavior and motivation, which in turn should affect their school performance. (d) Parents who are involved in preschool are more likely to continue to be involved in elementary school, and this continued participation is associated with better academic achievement and behavioral adjustment. (Zigler et al., 2006, p. 155)

The research of High et al. (2008), Piotrkowski et al. (2001), and Raver and Knitzer (2002) cited the importance of parent practices and how those practices impact a child's development and growth. Children who experience positive, prosocial, and nurturing relationships in their homes also progress in the other areas of development. Parent education and support can create a positive impact in helping children prepare for school. Programs can provide families with a sense of relationship and connectedness, along with trainings, strategies, and interventions for problematic situations.

# **Conditions for Creating Successful Parent Involvement: What Works**

Henderson and Mapp (2002) found parents want to be involved in their children's education but often lack the knowledge to do so. How do schools wrap their arms around the multiple factors, represented in families, to bring about effective parent involvement? Because parent involvement is a multi-faceted concept, Christenson and Sheridan (2001) identified four conditions that are helpful in creating effective parent partnerships: approach, attitudes, atmosphere, and actions.

#### **School Approach Matters**

Christenson and Sheridan (2001) cited the importance of the approach schools use in establishing a relationship. Their work supported the concept that a relationship begins with teachers and administrators communicating to parents they are integral to the

education process and their input is valuable and needed. Teachers should discuss the school and classroom mission statement, create an atmosphere for bi-directional communication, help parents understand assessment practices, and communicate in a variety of ways with families. The school should expect parents to be involved and invite them to do so. Teachers should help parents recognize learning takes place throughout the day and share learning opportunities outside the classroom.

# **Teacher Attitudes Make a Difference**

Christenson and Sheridan's research (2001) illustrated how positive school views about families send strong signals to parents that school is a good place for them and their children. The school communicates there are no problematic individuals, but opportunities for problem solving. Teachers and principals should listen to parent perspectives and work to understand family needs and situations. Listening and responding create positive conditions in establishing parent partnerships. An attitude of mutual respect fosters parent comfort and can aid in parent openness in dealing with difficult issues.

## School Atmosphere of Trust and Openness is Important

An open school atmosphere, according to Christenson and Sheridan (2001), recognizes the value of parent input in the decision-making process regarding children and promotes positive outcomes for them. Trust is foundational to relationships. Creating an atmosphere of openness and trust opens the door for building and continuing communication and relationship. Opportunities for connection can be created through family participation in school learning, cross-cultural awareness events, and education training classes.

## **Including Parents on the Planning and Action**

Building a shared responsibility, according to Christenson and Sheridan (2001), begins with school personnel inviting parents to be a part of the ongoing school processes. Parents can participate in school-improvement discussions and policy-making decisions. Teachers and administration can create effective communication strategies to ensure all families know what is happening at the school and how the school functions. Parents can be invited to staff development to create a better understanding of educational objectives and to build relationships with teachers. Effective school-family relationships can greatly enhance the learning environment and aid in academic, social, and emotional growth for students. A good relationship can make the job easier for teachers as they send home communication and homework. The partnership also creates conditions that allow parents to feel connected with their child's school.

### Six Types of Parental Involvement

Dr. Joyce Epstein was the frontrunner for studying the effects of parental involvement in children's education. Through her research, she established the Six Types of Parent Involvement and set an education industry standard for parental involvement through activities at home and participation in schools and classrooms. The research demonstrates "a positive effect on children's achievements, attitudes, and aspirations. Students gain in personal and academic development if their families emphasize schooling, let the children know they do, and do so continually over the school years" (Epstein, 1987a, p. 120).

Epstein looked at teacher practices to involve parents in their children's education journey by conducting surveys with over 3,700 teachers at 600 schools in 16 school

districts in Maryland. Eighty-two teachers with a wide variety of parental involvement practices were interviewed along with 1,200 parents from those teachers' classrooms. The study also collected data on the achievement and behaviors of 2,100 students in those classrooms. Results connected teacher partnership practices with parents and children in each classroom. The results of that study helped determine the need to identify other types of parent involvement taking place in and outside the classroom. Four types of parent involvement were identified as important to establishing the parent-school relationship (Epstein, 1987a). Later, two additional types of involvement were identified (Epstein, 1995, 2006). The six types of involvement: parenting, communicating, volunteering, learning at home, decision making and collaborating with community, can guide schools as they develop effective parent relationships that influence and enhance children's learning experiences.

## **Basic Parenting Practices**

Parents meet the basic obligations of family by providing food, safety, shelter, and health. They also establish positive home conditions to encourage school success through early childhood teaching and preparation for school from infancy through high school. "Parents lay the groundwork for students' success in school by building their children's self-confidence, self-concept, and self-reliance" (Epstein, 1987a, p. 121).

# **Effective Communication with Parents**

The communication between home and school facilitates the flow of information about school curriculum and the child's progress. "The school has an obligation to inform parents about school programs and their children's progress. And, parents are expected to act on the information received from the school" (Epstein, 1987a, p. 122). Two-way

communications between home and school reflect the reciprocity that a collaborative relationship must have in order to be most effective (Patrikakou et al., 2005a). Home and school communication remains a strong predictor of academic achievement even in high school (Catsambis, 2002; Patrikakou, 1997, 2004). Written communication is the most commonly referenced method of communication; it is also important to include person-to-person interactions, such as open houses at the beginning of the school year, parent-teacher conferences, parent-teacher organization meetings, American Education Week activities, and other school events that include parents. Each interface allows an opportunity for relationship development.

# Parent Volunteers: Effective Means of Involvement

Parents' involvement at school can include a variety of tasks: assisting teachers in the classroom, assisting in the office, preparing materials for students, recruiting parents to act as volunteers, supporting school initiatives and functions, helping in the lunchroom, supervising on the playground or at crosswalks, and assisting in other areas. It is important to extend involvement opportunities to parents on a continuous basis. Well-organized parental involvement sends the message that parents are wanted and welcome at the school and that their assistance is valuable.

## **Applying Classroom Learning at Home**

Parental involvement as promoted by Epstein (1987a) at home refers to teachers taking the lead to ask parents to support learning through help with homework, extended learning activities reinforcing classroom instruction, and other learning-enhancement activities. At home, parents can read with their children, play instructional games, tutor, and foster rich discussion using information and requests provided by the school. School

support for at-home learning can be reinforced through contracts, weekly reports from home, parent workshops to guide at-home instruction, and interactive homework activities with instructions and references for parents. Reading to and with children is a widely used home activity recommended by teachers. A study conducted by Epstein (1986) showed almost all parents are involved with helping their children at home on some level and reflected that more than 85% of parents work with their children for 15 minutes or more. Parents indicated that they would spend more time if given the knowledge about how to help.

# Parents as Partners in Decision Making

Epstein (2006) and Epstein et al. (2002) reported parents who are involved at this level have the opportunity to impact school climate and can be advocates for growth and change. This involvement can be through parent teacher organization (PTO) leadership or participation, school board membership or meeting attendance, or simply through conversations with administration and leadership. Parents can be invited to attend schoolimprovement meetings, participate in school assessment for accreditation, advocate for underrepresented groups, and be a part of orientation programs for new families and staff. Schools can offer training to encourage parental involvement and to create a sense of efficacy when working in the school setting. Results of parental involvement at this level often bring a sense of ownership in the school, and at this level, parents can serve as conduits of information to other parents

### Group Collaboration: Parents, School, and Community

This type of collaboration, as reported by Epstein (2006) and Epstein et al. (2002), shows parent involvement can be a network for both the school and other community

entities. There are opportunities to link medical services, public health information, and involvement in other education programs; to enhance current programming; and to gain knowledge about other programs or services. By working with the community, there can be a shared use of funding and building space, and increased quality as programming is shared. Through community collaboration, there is a greater awareness of services and resources in the community

# **Culture of Success**

Fantuzzo et al. (2000) emphasized that the goal of family-school connections for children's learning must be to create a culture of success, one that enhances learning experiences and competencies across home and school and that underscores that the partnership means shared goals, contributions, and accountability. Creating a culture of success should begin with teacher outreach to parents and a positive focus in early interactions.

Research has established a significant correlation for student learning and parent involvement. Christenson, Godber, and Anderson (2005) reported parental involvement has additional benefits in student performance: improved grades, higher test scores, and better attitudes toward school. Parental involvement has also been found to reduce school dropout rates, reduce special education referrals, increase college enrollment, and promote higher attendance rates in school.

Defining specifics in order to create a culture of success is difficult given the complexity of families today. "Christenson and Sheridan (2001) defined four features" that help to create a partnership "for families and schools: 1) a student-focused philosophy" that "encourages educators and families [to] collaborate to enhance learning opportunities,

progress, and success for students in [the] . . . academic, social, emotional, and behavioral [domains]; 2) a belief in shared responsibility for educating and socializing children . . .; 3) an emphasis on the quality of the interface and ongoing connection," keeping the child's educational growth as the goal; "and 4) a preventative, solution-oriented focus" where both teachers and parents work collaboratively to address issues from the onset and work together to address needs on a continuous basis (Christenson et al., 2005, p. 23).

Christenson et al. (2005) reported parents who understand their roles in parenting and education find themselves better able to fill the responsibilities accompanied with raising and educating children. They are more likely to become involved in school readiness activities and, later, participate in school programming if they view their participation as a requirement of parenting.

"McWilliam, Tocci, and Harbin (1998) researched . . . a family-centered approach [to parental involvement] considers the needs, wishes, and skills of the family" along with "the needs of the child" (Christenson et al., 2005, p. 23). The focus is then on welcoming the family into the school community and responding to family needs and concerns. They also noted that there is often "a 'disconnect' between pre-K and the K-12 systems" (Christenson et al., 2005, p. 23).

Pre-kindergarten programs that are designed with a purpose of preparing both students and parents for the school experience can involve daily interaction with teachers and parents, notes home inviting parent feedback, and many family activities. When families transition to the public school, the child-teacher relationship should be emphasized, keeping the parent as an integral part of the educational process. The K-12 systems need to make a better effort at welcoming the incoming kindergarten families as

they enter the public school system and integrating them as an integral part of their child's education journey.

#### **Increasing Parent Involvement**

The Appleseed Report (2006) and Ann Henderson's testimony to the senate on NCLB (2007) reported parents need a place at the table to know their presence and involvement matters and, further, to know they are needed and wanted as a part of the school community. Parental participation will increase with three key factors. First, parents need to have a good sense of their role construction. The parent who is confident in his/her role and knows positive parenting practices will demonstrate a greater likelihood for involvement in all aspects of his/her children's lives. Teachers who demonstrate the importance of parental involvement help parents' mindset to become one of thinking they are supposed to be involved in helping their children. The teacher influence should not be underestimated in influencing parental thinking. Second, parents who have a sense of efficacy in the parenting role and who feel confident they can assist and guide their child's education will become more involved in their child's education and feel freer to ask questions. They are more likely to be involved if they think their help will make a difference in their child's learning. Lastly, parents are more likely to increase their involvement if they are invited to participate in school-related activities and programs. The sense of invitation is strongly influenced by signals parents receive from their children and the school staff (Appleseed, 2006). The invitations send an important message of expectations and presence at school.

Christenson et al. (2005) reported on the data in education reflecting the importance of educators working in partnership with families. Educators can be the catalyst at

improving school climate, parent confidence, and student success. Teachers can create the inroads for parents to know the teachers on a deeper level and the work they do. Parents who understand educational practices and expectations tend to become better advocates for their children. The research demonstrated parents are often involved when very satisfied and when dissatisfied.

Epstein (1986) and Epstein et al. (2002) emphasized educators must work to reinforce the message that children are connected to the same families but to different teachers over the course of their school years. This message can aid parents in better understanding their role in this ongoing journey. When schools and families operate as true partners, they can weather the difficulties that can occur in any relationship: misunderstandings, conflict, and differing opinions. It is the continuous efforts that focus on the goal that brings about maximum success for students.

Rockwell et al. (1996) found teacher invitations to include parents in classroom learning do not have to take a great deal of additional time. Increasing the daily contact can be as simple as inviting parents to share in the day's opening activities and then be on their way. Parents can be invited to attend a principal's round table discussing school issues over a muffin and coffee. Teachers can collaborate and conduct grade-level information sessions where parents are free to ask questions. Back-to-school picnics are effective ways to help new families feel welcome at the school. Each of these activities can also have information for other involvement opportunities and upcoming events. Parent rooms can indicate to parents that their presence is important and welcome. Parent rooms can also be places for teachers to leave projects for volunteer completion, a place for meeting with teachers, and a
place for parents to interact during the school day. Some parent centers house information libraries.

## Adults as Partners in the School Environment

Merriam, Caffarella, and Baumgartner (2007) established the importance of families and educators working together in partnership roles. The attitudes they hold about each other and an atmosphere that is conducive to collaboration set the stage for the action of moving forward in collaboration and support for the student. Malcolm Knowles, an adult learning theorist and researcher, affirmed the need to model and give respect when working with parents. Parents know their children best, and when they are a positive part of the education team, it becomes easier to work towards the goal of student achievement.

Knowles in Merriam et al (2007) focused on characteristics of the adult learner to bring about maximum results for both families and schools. These characteristics can be applied to the school-home relationship formation:

- Adults move from being dependent to self directing. They should be respected and treated with a spirit of mutuality in the learning journey. Both teachers and parents can be joint inquirers in the relationship.
- Parents bring a rich reservoir of experience from their background and homes and should be asked for input regarding their knowledge and ideas.
- Teachers should be observant of parents' social and educational backgrounds and use them as springboards upon which to build a relationship.
- Adults want information to be useful and applicable. Keep the focus on the children and the concrete things parents can do to support them.

- Parents want their children to succeed; motivation for that result is internal.
  Teachers can help, too, by providing their emotional support and encouragement for children's growth and success.
- Adults need to know why they need to learn something; help parents by explaining what children are learning and the value of parents' participation in the children's learning (Merriam et al., 2007).

# Linking Research to Readiness Programming

Identifying research-based readiness programs that align with research expectations and systematically incorporate a parent involvement component is somewhat challenging. Most readiness programs provide curriculum for teachers' use and may not be designed to align with the research. Because this program became the basis for the current study, the program is analyzed here for its relationship to research literature.

# **Gearing Up for Kindergarten: A Parent-Involvement**

## and School Readiness Program

The key building blocks and objectives of the Gearing Up for Kindergarten program, include: school readiness for children entering kindergarten, parent education, parent involvement with children's learning and school, building home-school-community partnerships, and transition-to-school knowledge and practices involvement with children's early learning, (Brotherson, 2007). These objectives support and reinforce the research literature (Appleseed, 2006; Christenson and Sheridan, 2001; Fielding, 2010; 2001; Epstein, 1987a; Fabian and Dunlop, 2007; Fielding, 2009; Fielding et al., 2004; Harvard Family Research Project, 2006a; NAEYC, 2009; Rhode Island KIDS COUNT, 2005; Sprenger, 2008; Wesley and Buysse, 2003; and Zigler et al., 2006) on child development, school readiness and parent involvement in their child's development and education.

# School Readiness and Gearing Up for Kindergarten

The concept of school readiness, emphasized in the Gearing Up for Kindergarten program, relates well to Wesley and Buysse's (2003) research, citing educator perspectives regarding the importance of school readiness for children entering kindergarten. Epstein's (1987a) research established the importance of parent care for children's basic needs along with a healthy environment that prepares children for the school experience.

## **Parent Education**

Fielding et al. (2004) reinforce the importance of parent knowledge and understanding of child development in supporting children's preparation for life. The Gearing Up for Kindergarten program supports this concept along with the previously referenced Appleseed Report (2006) that cited parent efficacy and competence as key factors in parents' decisions to become involved in their children's education. The Gearing Up for Kindergarten program provides a parent education component that teaches basic child growth and learning information in the 5 domains of development.

### Parent Involvement in Children's Learning

The Gearing Up for Kindergarten program reflects the parent involvement research of Epstein (1987a), Fielding (2010), NAEYC (2009) and Sprenger (2008) by incorporating early learning guidelines with parent practices to provide learning opportunities for parents. Recall Zigler et al. (2006) found parental involvement in preschool has a direct effect on children's success in social and emotional behaviors and supports their academic success. The Gearing Up for Kindergarten program incorporates and teaches parent/child activities each week in the parent session. Student learning at home is reinforced with lessons that include take home and practice activities where the parent is the teacher initiating and teaching new activities. In addition, parent/child activities are a primary part of the classroom experience.

### **Building Parent Partnerships**

Christenson and Sheridan, (2001), Epstein (1987a), and the Harvard Family Research Project (2006a), as reported earlier, promote parent partnerships as a part of children's learning and education. These partnerships help to develop parent understanding of the learning environment and provide opportunities for parents to work with their children at home, as volunteers at school, and as advocates for program and policy development. Opportunities may also be realized to enhance classroom learning as parents become aware of learning concepts and use skills they may have to support and develop the learning. The Gearing Up for Kindergarten program uses a workshop and training model (Harvard Family Research Project, 2006a) and promotes parent, school, and community partnerships through the education lessons on all areas of a child's development, including relationships in the home, school, and community. As reported earlier, community connections support the culture of success (Christenson and Sheridan, 2001) and work to instill a shared responsibility for education. Through participation in the Gearing Up for Kindergarten program parents experience a positive school atmosphere (Christenson and Sheridan, 2001) and experience a positive approach to school building school connections and relationships (Christenson and Sheridan, 2001).

### **Transition to School**

The Rhode Island KIDS COUNT, 2005, Executive Summary research, noted previously in the chapter, showed children will not enter school ready unless they are provided with early learning experiences that support whole child development. These developmental experiences make the difference in children's transition into school and set the stage for success in school. Fabian and Dunlop (2007) reported families with logistical transition information, and who partner together with schools, experience greater satisfaction in relationships and smoother transitions to school. The Gearing Up for Kindergarten parent curriculum and classroom instruction provides information for parents and students to support success in the transition to school. Classes, supporting Bronfenbrenner's (1979) research, provide practical information and support for school registration, locating school boundaries, and informing parents and students of standard school practices.

### Gearing Up for Kindergarten Research and Development

The Gearing Up for Kindergarten program has been studied by researchers at North Dakota State University since the program's inception in 2006. Study results with parents suggested that the parent-school relationship is enhanced through participation in the Gearing Up for Kindergarten program (Brotherson, Query, & Saxena, 2007; Brotherson et al., 2008; 2009). The program is designed to enhance parent-school relationship through participation in the Gearing Up for Kindergarten program. Through intentional conversation and targeted curriculum information, a central message is laid out as a part of the program; parents an important part of their child's education journey. The Gearing Up

for Kindergarten program curriculum uses research to emphasize parents as children's first and most important teachers knowing best how their children learn.

Gearing Up for Kindergarten (Brotherson, 2007; Brotherson et al., 2007) is designed as a prevention-focused, preschool-parent education program for parents and their children who will be entering kindergarten in the following year. Gearing Up for Kindergarten curriculum targets child development and school readiness. The Gearing Up for Kindergarten 16 week program design provides an intensive educational experience combining preschool learning activities for pre-kindergarten children with a parent education and involvement component as the key elements of the program. The program's intended design is for families with a child entering kindergarten the following year, but can also be used with three year olds.

The program was developed in 2006 at North Dakota State University. A federal mandate to increase parental involvement in schools resulted in an opportunity for the federal Department of Education to partner with the North Dakota State Parent Information Resource Center (PIRC). Through the PIRC, Gearing Up for Kindergarten was added to additional sites across the state of North Dakota in 2008 (Brotherson et al., 2009).

The Gearing Up for Kindergarten program (Brotherson, 2007; Brotherson et al., 2007) has a unique parental involvement emphasis in addition to its preschool program for children. Through parental participation in education centers, parents are expected to experience relevant learning activities and, in the parent session, how to implement learning at home. The hands-on experience teaching is designed to provide a sense of efficacy in knowing what kinds of activities support childhood development and learning. The take-home activities parents create each week are intended to provide opportunities for

immediate application at home. Research demonstrates parental knowledge and efficacy increases the likelihood for critical learning and development towards school readiness to occur at home (Epstein 1987a; Zigler et al., 2006).

The curriculum was piloted and refined over a five-year period to develop an educational program that adheres to research using effective parental involvement and school readiness methods. The preschool education curriculum consists of a flexible, adaptive set of parent-child learning and activity sessions. The parent education curriculum consists of a series of structured educational sessions that focus on child development, school readiness, and healthy parenting (Brotherson, 2011).

The preschool education curriculum centers on learning and activity sessions for parents and children. The program design has parents and children together at the beginning of each class. At this time they participate in a 45-minute session of different learning stations. Each station is created with activities designed to address and enhance knowledge and skills in math, science, reading/literacy, sensory awareness, motor ability, imagination, and social-emotional ability. Built into that timeframe is a preschool-style "circle time" for reading.

The next segment of the curriculum has children separate from parents. Children spend the remainder of the session in additional preschool learning activities planned by the preschool teachers using the curriculum guide. The curriculum for the parent-child activity sessions was developed and compiled by North Dakota State University researchers. Curriculum design is created for two contracted facilitators for the child and parent sessions, as well as a classroom assistant (Brotherson, 2007)

The parent education curriculum consists of 16 lesson modules designed to be adapted to the program model used for any specific site utilizing the Gearing Up for Kindergarten program. Typically, as in this study, the program model is delivered in two 8-session split semesters (fall and spring sessions), or a single 16-session program (spring prior to kindergarten). There is an alternate model of 10-sessions available. Parent-education lesson modules were adapted from the research-based curriculum Bright Beginnings for Young Children, developed by Dr. Sean Brotherson of the North Dakota State University Extension Service, and other available parent-education topics. Parent/child take-home activities and handouts were written by Parents as Teachers, and adaptations for New Americans were developed by Even Start. Learning topics for the modules include: Parenting Styles, Raising a Reader, Effective Discipline Techniques, Developing Responsibility and Self-Help Skills, Learning Styles and School Readiness, and other selected topics.

#### Summary

Parental involvement in children's education has been shown to be a benefit for student achievement as well as emotional and social growth. Families that work well with the schools have a greater support system and are likely to experience success in high school and in decision making. The research (Christenson & Sheridan, 2001; Epstein, 1986; 1987a; 2001; Fantuzzo et al., 2000; Gestwicki, 2004; Harvard Family Research Project, 2006b; Henderson, 2007; Patrikakou et al., 2005a; Rockwell et al., 1996) is clear that teachers and schools do impact the likelihood of parents' participation in their child's education. Teachers have the ability to influence involvement through their attitudes and

beliefs about parental involvement. Schools that provide opportunities for meaningful involvement reinforce the importance of the parent role in education.

It is also important for schools and parents to recognize the change that has occurred in educational expectations. The increased academic standards should be shared with parents, especially parents of young children. Getting parents on board is important in elementary school, but should ideally begin in early childhood. Epstein's (1987a) sixth type of parental involvement is an example of how communities and schools can work together to educate parents about the privilege and responsibility of teaching their children both before and after school begins.

The research demonstrated the important roles parents, community, and schools play in a child's development and achievement. Responsibility for education is shared, along with the success of achievement, and is supported by the Overlapping Spheres of Influence theory framework (Christenson & Sheridan, 2001; Epstein, 1986; 1987a;; 2001; Fantuzzo et al., 2000; Gestwicki, 2004; Harvard Family Research Project, 2006b; Henderson, 2007; Patrikakou et al., 2005a; Rockwell et al., 1996).

Many pre-kindergarten programs focus on preparing children for school. The Gearing Up for Kindergarten program has additional supportive components that target parents and schools with education, transition information and expectations, and an understanding of school readiness to better equip parents in effectively preparing their children for school (Fabian &Dunlop, 2007).

"Young children's earliest experiences and the environment set the stage for future development and success in school and life" (Rhode Island KIDS COUNT, 2005, "Intro"). As the research (Christenson & Sheridan, 2001; Epstein, 1986; 1987a; 2001; Fantuzzo et

al., 2000; Gestwicki, 2004; Harvard Family Research Project, 2006b; Henderson, 2007; Patirkakou et al., 2005a; Rockwell et al., 1996) indicated, early experiences influence brain development, establishing the neural connections that provide the foundation for language, reasoning, problem-solving skills, behavior, and emotional health. Families and the preschool community play a crucial role in working comprehensively to assist children in coming to school ready to learn at a kindergarten level. Children from families with economic security and healthy relationships are more likely to enroll in preschool and to experience learning opportunities that support school readiness and, ultimately, success in school. Young children thrive when parents, family, educators, and the community surround them with love; support; and continuous, positive opportunities on their learning journey (North Dakota Department of Human Services, 2008a).

## **CHAPTER 3. METHODOLOGY**

## Introduction

Schools today face great challenges as they grapple with instructing children at multiple levels of readiness and skill. Children who enter school behind in their development and academic knowledge tend to stay behind as they move into the next grade (Fielding, 2009). Research demonstrates one-third of all students entering school experience some kind of learning difficulty and 16% enter school experiencing great difficulty (Gebeke, 2010).

The literature review provided research demonstrating the importance of parental involvement in children's education. Parent involvement research has been demonstrated to be a benefit for student achievement as well as emotional and social growth. The research supports the idea that when families work well with the schools and have support system children are likely to experience success in high school and in decision making. The research (Christenson & Sheridan, 2001; Epstein, 1986; 1987a; 2001; Fantuzzo et al., 2000; Gestwicki, 2004; Harvard Family Research Project, 2006b; Henderson, 2007; Patrikakou et al., 2005a; Rockwell et al., 1996) affirmed the value of teachers and schools in connecting with parents and providing opportunities for participation in their child's education.

Kindergarten is the formal entrance into the education world, and today, many children do not arrive at school with the skills they need (NIEER, 2004). Increased learning expectations for kindergarten have resulted in fewer children ready to learn upon entrance into school. Until recently, kindergarten has been thought of as an environment to learn social skills and the alphabet (Fielding et al., 2004).

The expectations of kindergarten readiness have changed significantly in the past two decades; hence parents may lack a clear understanding of the educational expectations for their kindergartner (Espinosa et al., 1997; Fielding et al., 2004; Fielding, 2009; Hanson, 2008). For example some of the increased expectations for children include: knowing some basic letters, counting to ten or higher, taking turns, listening for ten minutes at a time, and writing their first name (Fielding, 2009). Misconceptions regarding expectations can contribute to academic deficiencies in children (Espinosa et al., 1997; Fielding, 2009; Hanson, 2008; High/Scope Educational Research Foundation, 2006; Weiss et. al, 2006). A lack of parent understanding about readiness expectations can contribute to students entering school behind in learning, even before they begin.

Kindergarten readiness is a complex and multidimensional construct influenced by many early childhood experiences. For this study, school readiness is defined using a composite of benchmarks developed by the National Education Goals Panel (Kagan et al., 1995). These benchmarks are the often-identified, current standards generally accepted by the early childhood community and are reflected in the PPASR assessment and in the Gearing Up for Kindergarten curriculum. The NEGP identifies five domains of children's development and learning important for school success:

- Physical well-being and motor development
- Social and emotional development
- Approaches to learning
- Language development
- Cognition and general knowledge

These five domains are interrelated and dependent upon each other; development in one area will affect development in others. In this perspective, there is no single or uniform standard of readiness. Children's development is individual and can vary greatly according to ability and exposure to environment. At age five, a child will not demonstrate all necessary skills in every area. It is important to note one domain is not more important than another. Children's readiness for school encompasses skill sets in all the domains. While parental involvement in education and school readiness has been researched extensively, there is little research on combined preschool and parental involvement programs, along with transition-to-school practices, to support greater readiness for children at the start of kindergarten (Rhode Island KIDS COUNT, 2005).

## **Research Questions and Study Design**

The research questions drive the study design and impact procedures and program assessment. The research questions are followed by description of the study design.

### **Research Questions**

- Is the change over time in the Practical Parent Assessment of School Readiness (PPASR) scores for the treatment group parents different from the change over time for the control group parents for the 5 domains of child development?
- 2. Is the change over time in PPASR scores for the treatment group parents different from the change over time for the control group parents for the transition-to-school knowledge and familiarity construct?
- 3. Are there differences in change over time between treatment and control group in children's knowledge of letter identification, number identification, and oral counting as measured by AIMS Web scores?

4. Is there a difference between treatment and control group in the strength of the relationship between the parent PPASR scores and child AIMS Web scores?

# **Study Design**

This study design included both a treatment and a control group of 4-year olds, who entered kindergarten the year following the program, and one or both parents. The study was designed to measure whether participation in the Gearing Up for Kindergarten program significantly impacted children's academic growth, on selected measures, and parent perceptions of their child's development and school readiness.

This study has a unique approach that includes both parents and children in the intervention phase of the program. The treatment group received the intervention; participation in the Gearing Up for Kindergarten program. The control group did not receive the intervention.

The quasi-experimental design was used due to the fact that the groups were not arranged by random assignment. The preferred design for research is the experimental design. Without random assignment, that was not possible for this study.

A pretest, posttest, post posttest design was used to assess both students and parents. This design was used with both the treatment and control groups in the study. Treatment group: 0 (pretest)  $X_1$  (intervention) 0 post-test 0 post posttest Control group: 0 (pretest)  $X_2$  (no intervention) 0 post-test 0 post posttest The pretest-posttest control group design criterion was met with the two groups. Both groups were given the pretest followed by the intervention for the treatment group. Both

also included a post posttest, for both parents and children, at the children's entrance into kindergarten.

# **Description of Intervention Program**

## **Program Design**

This program used a 16 week model, with classes held once a week for 16 weeks. This model included a 6 week break after the first 8 weeks. The series resumed after the break. The 90 minute classes began at 6:00 pm and concluded 7:30 pm. There was no cost to families to participate in the program.

The evening was divided into two segments, parent child time in the classroom for the first 45 minutes followed by separate instruction for children and parents. Each child's classroom had one preschool teacher and one para-educator along with a parent educator who worked with families the first 45 minutes of each session.

The parent educator and parents left the classroom at the 45 minute mark for a parents' only instruction component. During the 45 minute parent session parents learned how to support their child's learning and development in preparation for the start of school. The parent educator used Gearing Up for Kindergarten curriculum for instructional materials and handouts.

Children remained in the classroom with the preschool teacher for continued learning activities. The preschool teacher and para-educator worked together with curriculum, to create lessons and centers for each evening's class.

The teachers were trained by Dr. Brotherson, creator of the Gearing Up for Kindergarten program, in a three hour training session that provided background research on child development and school readiness. The training also included an overview of the Gearing Up for Kindergarten curriculum, evaluation practices, and steps for effective program implementation. Teachers developed and practiced a lesson with each other before beginning the program with students and parents.

## **Study Logistics**

The treatment study group consisted of classrooms at two different sites: a local elementary school and a children's center. The elementary school held classes on Monday evenings, and the children's center held classes on Tuesday evenings. The elementary site had one classroom with 13 students; the children's center had two classrooms, the Green Room with 14 students and the Blue Room with 13 students.

## **Evening Format**

Upon arrival each evening, parents and children signed in outside the classroom. Student names were listed on a paper in bold letters. Students found their printed name, some with the help of parents, and then wrote their name on the line below. Parents also signed a weekly attendance sheet. Both parents and children had nametags prepared by the teacher. Parents and children then went into the classroom to participate in a variety of learning centers; teachers encouraged parents to let children choose which centers to visit. Parents were an integral part of the center time and were encouraged to participate, with the child leading the way. Center time emphasized the development of language and literacy, motor skills, cognitive-thinking skills, social skills, and play. Each center provided directions for ideas, or parents and children were free to explore and play as they chose. Books were also available in a library corner for families wishing to read. This opening activity lasted for 30 minutes. Following center time, there was a 15-minute reading time conducted by the teacher. The segment brought all participants together with children

usually sitting on their parent's laps. The format varied each week: reading in family units, children reading to parents, guided teacher instruction, or participants responding by following along with the teacher. During this time, the teacher emphasized the importance of early literacy and developing a love of reading. The teacher modeled an interactive reading style and encouraged parents to visit with their children during and after the story. Parents learned to ask children questions relating to the story and acquired skills to make literacy connections with everyday life.

At 6:45 pm, parents left with the parent educator for parent education in a separate room; the Gearing Up for Kindergarten Parent Education curriculum was used. Instructional topics included Brain Development in Children, Parenting Styles, The Importance of Sleep, Learning Styles, School Readiness, Raising a Reader, Temperament, and Guidance and Discipline. These topics were shared with parents through activities and discussion. Parents also made hands-on, take-home activities, for use with their child, each week relating to the evening's lesson. While creating the activity, parents also learned the importance of the particular skill associated with the activity. Each week, parents took home educational information, from the parent curriculum, on child development in the skill or subject area for that week.

During the parent time, children worked in the preschool classroom, participating in learning activities as well as gaining an understanding of listening to the teacher, following directions, and experiencing time away from Mom and Dad. Lessons included alphabet sound and writing, number sense application, fine and gross motor activities, more play, and reading.

Following the parent education, parents joined their children in the classroom for a final activity. The evening ended with children picking out a library book, provided by the researcher's parent center, to take home each week.

To support families and encourage attendance, the parent center offered free, onsite childcare for siblings of the program participants. These children were cared for in a room next door to the classroom. Staff implemented all aspects of the program with careful attention to beginning and ending on time in an effort to model the school environment for families.

The focus of the literature reviewed in this study supported appropriate early childhood development, development of school readiness, parental involvement in education, and parental understanding of readiness and transition-to-school knowledge and familiarity. The parental assessment, the PPASR, supported the child development domains and school-readiness perceptions as cited by the NAEYC (2009), NEGP (1995), and the North Dakota Department of Human Services (2008a). The Gearing Up for Kindergarten program curriculum also supported early childhood development and readiness as promoted by the NAEYC (2009), NEGP (1995), and the North Dakota Department of Human Services (2008a). The Literature Review also examined avenues to support parents' understanding of school readiness. When parents understand their roles in parenting and education, they are better able to fulfill the responsibilities that accompany raising and educating children. They are more likely to become involved in school readiness and, later, school activities, if they view their participation as a requirement of parenting (Christenson & Sheridan, 2001).

Research for this study was conducted with resources from the Educational Resources Information Center (ERIC), the Harvard Research Exchange, the National Institute for Early Education Research (NIEER), the National Education Goals Panel (NEGP), the North Dakota Department of Human Services (DHS), the North Dakota Department of Public Instruction (DPI), and the North Dakota State University Library.

### **Population and Sample**

The North Dakota State University Institutional Review Board approved the Gearing Up for Kindergarten research study. The study was advertised to all incoming kindergarten students and their parents, for fall 2011, in an urban Midwest public school district.

Prior to sending study information, a designated number of 90 participants had been predetermined for an opportunity to participate in the study, 45 to attend the Gearing Up for Kindergarten program and 45 to participate in the study. Classroom space was limited to a recommended 15 students per class, and funding levels allowed for 3 classes.

Study participants were difficult to locate as there is no formal list of 4-year olds available through public agencies. The researcher chose to work through the local school district where a comprehensive list of elementary families was legally available, though an up to date list of elementary families was not available prior to the start of the 2010 school year. The researcher obtained family names and addresses on September 3, 2010 and recruitment letters were mailed to the entire population of 2,317 elementary families on September 7, 2010. (Appendix *A*) The recruitment letter invited families with an incoming kindergartner in the fall of 2011 to participate in the Gearing Up for Kindergarten program.

Recipients were also encouraged to share the opportunity with others, who may not yet have entered the school system, but had a child entering kindergarten in the fall of 2011.

Interested parents were given until September 15, 2010, to respond. The time before the pre-determined start date of the Gearing Up pre-program assessments was, short, less than three weeks after the return of the letter indicating acceptance into the program. Parents were advised that program space was limited and admission into the program would be in order of received responses. Positive responses were received from 98 families.

The second letter, September 22, 2010 (Appendix B, Appendix C), was mailed to the 98 families responding to the first recruitment letter. Two separate letters were sent, to respondents. The first 49 respondents were assigned to the treatment group and the next 49 respondents were placed in the control group. In each letter, parents were given group assignment information and a registration form for either the treatment group (Gearing Up for Kindergarten program participation) or the control group (Gearing Up for Kindergarten assessment information only). The forms outlined the details for each group: treatment group, participation in the 16-week Gearing Up for Kindergarten series, and control group, completing assessment information on school readiness. Participants were provided an addressed, postage paid envelope to return interest forms to the researcher's work site at a local elementary school.

Although random assignment is preferred in a quantitative study, given the time constraints of the class start date, the decision was made to give the opportunity to attend the Gearing Up program to the first 100 respondents. After receiving acceptance letters for the study, some families decided not to attend due to the length of the program. This was

expected and was the reason for an initially accepted larger number. The study began with 40 children in the treatment group and 35 children in the control group.

The population of the study may not accurately portray the population of kindergartners in the school district due to a lack of response on the part of some or due to not having a child in elementary school, thereby not receiving the letter regarding the Gearing Up for Kindergarten program.

## Instrumentation

The quantitative study gathered data with an emphasis on collecting and analyzing information in the form of numbers (Creswell, 2008). The PPASR, parent instrument looked at parent perceptions of their child's development, as described by the 5 domains of development, and school readiness. The AIMS Web, children's assessment assessed for student proficiency in letter identification, number identification, and oral counting.

### **Assessments for Parent Perceptions and Demographics**

Parent assessments included the Practical Parent Assessment for School Readiness (Appendix D) and the Parent Response Form (Appendix E). The PPASR contains Likert-scaled measurements and the Parent Response Form contains demographic information.

**Practical Parent Assessment of School Readiness.** The Practical Parent Assessment of School Readiness (PPASR) is a measure completed by parents, assessing perceptions of their pre-kindergarten children. The PPASR contains assessments pertaining to the 5 domains of child development and the transition-to-school construct. The PPASR was developed and pilot tested in collaboration with the Gearing Up for Kindergarten program by Dr. Sean Brotherson at North Dakota State University. It was first pilot tested in 2006 and 2007 with 80 participants in the Gearing Up for Kindergarten program and then subsequently refined in 2008 (Brotherson, Query, & Saxena, 2007; Brotherson et al., 2008). It was further tested with 102 participants in 2008 (Brotherson et al., 2009). It was intended to be a holistic assessment of child development and school readiness that encompasses multiple domains of child development and preparation for school entry. The primary assessments contained in the PPASR are each described in the following paragraphs.

The Practical Parent Assessment of School Readiness was reviewed for face validity and content validity, which are important elements of construct validity, in the pilot-testing process (Brotherson, 2011). First, a systematic review of a wide variety of existing measures of a child's school readiness was conducted, and sample indicators were organized into multiple key domains of child development and school readiness. Once these indicators were organized, the draft instrument was prepared, and then, all assessments in the PPASR were re-checked against existing literature and measures to ensure content validity. Next, the draft instrument was reviewed by six to eight early childhood educators and professionals, including both university professors in child development and early childhood education as well as practicing early childhood professionals with at least a master's degree, for both face validity and content validity. Feedback about the instrument was gathered and incorporated into the revision process. Finally, the instrument was pilot tested with parents of pre-kindergarten children, and selected parents were asked to provide feedback on the wording of questions and the perceived face validity of the instrument. This testing and review process covered two

years and resulted in the current version of the Practical Parent Assessment of School Readiness (Brotherson, 2011).

Additional testing for construct validity was conducted with the PPASR using convergent validity, in which validity is illustrated for the instrument in question (PPASR) through significant correlations with existing measures that assess a similar construct. The instrument selected for comparison was the School Entry Profile, a sound early childhood assessment used extensively by the state of Missouri in pre-kindergarten and kindergarten testing of young children. This instrument has regularly been used in assessment efforts in the Gearing Up for Kindergarten program over the past 3 years, and teacher assessment scores are available from the same population in the same time period as the PPASR. Each sub-scale of the PPASR was correlated with the similar topical sub-scale in the School Entry Profile.

The PPASR is a 42-item parental assessment of a child's school readiness across 5 key domains of development that is completed by a child's parent. It includes the 5 domains of child development and a school transition domain sub-scale. The questions for the 5 domains are answered on a Likert-type scale from 1 (*hardly ever*) to 5 (*almost always*). The questions for transition are answered on a Likert-type scale from 1(*not at all*) to 4 (*to a great extent*). Each sub-scale and its characteristics are described. Four of the 5 child development domains, approaches to learning, social and emotional (note: the social and emotional domain of development is measured in separate subsets in the PPASR), cognition and general knowledge, and language have comparable measures to the School Entry Profile. The physical well-being and motor development does not have a comparable

measure in the School Entry Profile; instead a correlation was run for this domain with a dissimilar construct.

*Approaches to learning scale*. The approaches to learning scale of the PPASR indicated a significant and positive correlation with the Learning to Learn sub-scale of the School Entry Profile, r(67) = .366,  $p \le .01$ . This relationship suggested that the Approaches to learning sub-scale has a substantial level of compatibility with a similar measure in the School Entry Profile, an important indicator of construct validity. The scale, as measured in the PPASR, is a 9-item scale designed to assess a child's learning approach and effort. The Cronbach's alpha measured with Gearing Up for Kindergarten test subjects in 2010 was .77 (S. E. Brotherson, personal communication, May 20, 2011).

*Social development scale*. The social development scale of the PPASR showed a significant and positive correlation with the Working with Others sub-scale of the School Entry Profile, r(67) = .332,  $p \le .01$ . This relationship suggested that the PPASR sub-scale has a substantial level of compatibility with a similar measure of social development in early childhood. The scale, as measured in the PPASR, is a 10-item scale designed to assess a child's social development and peer relationships. The Cronbach's alpha measured with Gearing Up for Kindergarten test subjects in 2010 was .89 (S. E. Brotherson, personal communication, May 20, 2011).

*Cognition and general knowledge and language development scale.* The cognition and general knowledge scale of the PPASR showed a significant and positive correlation with the Mathematical and Physical Knowledge sub-scale of the School Entry Profile, r(67) = .355,  $p \le .01$ . Again, this relationship provided evidence for construct validity through a significant correlation with a similar existing scale. The scale, as measured in the PPASR,

is an 8-item scale designed to assess a child's proficiency with regard to such items as basic literacy, numeracy, and communication skills. The Cronbach's alpha measured with Gearing Up for Kindergarten test subjects in 2010 was .78 (S. E. Brotherson, personal communication, May 20, 2011).

*Emotional development scale.* The emotional development scale of the PPASR showed a significant and positive correlation with the Learning to Learn sub-scale of the School Entry Profile, which also assessed aspects of emotional development. This correlation was significant at r(67) = .327,  $p \le .01$ . The scale, as measured in the PPASR, is an 8-item scale designed to assess a child's emotional development. The Cronbach's alpha measured with Gearing Up for Kindergarten test subjects in 2010 was .78 (S. E. Brotherson, personal communication, May 20, 2011).

Discriminant validity was used to establish construct validity for those sub-scales in the PPASR that did not have a comparable measure in the School Entry Profile. Discriminant validity suggested that the measure is distinctive from other measures that theoretically should assess a different construct. The physical well-being and motor development scale of the PPASR did not have a comparable measure that allowed for assessment in the School Entry Profile, so a correlation was run for its relationship to a dissimilar construct, Mathematical and Physical Knowledge sub-scale of the School Entry Profile. There was no significant correlation between these constructs, r (67) = .208, thus showing, through discriminant validity, that this scale also had evidence of construct validity. The scale, as measured in the PPASR, is a 7-item scale designed to assess a child's physical abilities in such areas as gross and fine motor skills. The Cronbach's alpha measured with Gearing Up for Kindergarten test subjects in 2010 was .71 (S. E. Brotherson, personal communication, May 20, 2011).

*Transition-to-school scale.* The scale consists of an eight-item assessment of a preschool child's school knowledge and familiarity. The eight-item transition-to-school scale has questions that are answered on a Likert-type scale from 1 (*not at all*) to 4 (*to a great extent*). The scale, in the PPASR, asks parents to assess a child's feelings and experiences related to school knowledge and familiarity. The Cronbach's alpha measured with Gearing Up for Kindergarten test subjects in 2010 was .85 (S. E. Brotherson, personal communication, May 20, 2011).

Taken together, these findings suggested that the PPASR sub-scales in the PPASR demonstrate solid construct validity and actually measure what they are intended to measure. The transition-to-school scale was added to the instrument after its initial pilot testing and revision. Each of these sub-scales were assessed for face and content validity using a similar process to that described above, thus establishing construct validity. Further testing for construct validity of these scales using other approaches has not yet been conducted.

**Parent Response Form.** The demographic questions included on the Parent Response Form are standard items designed to provide basic information regarding the characteristics of study participants. The questions include participant gender, age, number of children, residential setting (rural to urban), family status, education level, employment status, racial or ethnic background, child gender, and family relationship to child. Additional questions of interest that are included address participant eligibility for Head Start, first child in kindergarten, and parent education level.

Parents in the treatment group received the independent variable, participation in the Gearing Up for Kindergarten school-readiness program. Parents in the control group did not receive the Gearing Up for Kindergarten school-readiness program. Parents in both groups were assessed for their perceptions of their child's development and school readiness. The measurements of the two groups were compared to determine if there were differences in parent perceptions due to participation in the Gearing Up for Kindergarten program.

### Assessments for Measuring Children's Academic Performance

Children were assessed on three academic measures, letter identification, number identification, and oral counting using the AIMS Web student assessment tool. One minute probes on each measure were given during the same time intervals as the parent assessments.

**AIMS Web.** The AIMS Web (Appendix F, Appendix G, Appendix H) assessment tool is a scientifically based, assessment that is a continuous method of measuring student progress in academics. This tool is designed inform teachers and to tailor instruction to meet individual student needs.

AIMS Web provides its users the assessment materials and ability to organize and report Curriculum-Based measurement (CBM)–standardized measures of basic skills–including reading, early literacy, and mathematics. CBM is an approved set of testing practices based on over 30 years of federally funded research and has been reviewed as meeting professional assessments standards by the Reading First Assessment Committee and the National Center on Student Progress Monitoring. More than 30 years of research has shown that listening to a child read graded passages aloud for 1 minute and calculating the number of words read correct per minute provides a highly reliable and valid measure of general reading achievement, including comprehension, for most students.

This testing practice, Reading Curriculum-Based Measurement (R-CBM) has met the standards for use in Reading First as determined by the Secretary of Education's Committee on Reading Assessment and the Office of Special Education Program's National Center for Student Progress. (AIMS Web, 2011)

More than 200 empirical studies published in peer-reviewed journals (a) provide evidence of CBM's reliability and validity for assessing the development of competence in reading, spelling, and mathematics and (b) document CBM's capacity to help teachers improve student outcomes at the elementary grades. (Fuchs & Fuchs, 2004, p. 1)

Fuchs (2004) identified CBM as an effective tool that specifically identifies student instructional needs and demonstrates growth over time. This tool, when used consistently throughout the year, meets the research requirements of performance level at one point in time, the slope of performance actually associated with overall growth, and usefulness in teacher instruction.

Curriculum-based measurement is a specific testing strategy that uses limited numbers of standardized and valid testing measures. "The core CBM testing strategies are varied systematically and combined with other pieces of information to make one of five decisions in the Problem-Solving model" (Shinn, 2002, p. 679). The first step is to determine whether there is a problem in an area of learning, Problem Identification, which identifies the specific area of deficiency. The next step is Problem Certification, where the

severity of the problem is determined. Solutions to the problem are explored in the next step, followed by the Goal Setting process and, finally, Evaluating the Solutions. The CBM model is consistent and continuous, and reliable in its usefulness in individual instruction for student remediation and improvement (Shinn, 2002).

These standardized tests are based on general outcome measurement principles in order to efficiently and accurately evaluate student progress relative to a year-end target, regardless of curriculum or intervention. AIMS Web assessments are designed for kindergarten through eighth grade, and are valid and reliable when used with those populations. The AIMS Web pre-K measures are norm based with other schools that use these measures across the United States. The sample populations have been determined significant; however, at this time, the United States does not have fully funded pre-K programs available to all students. States and schools using the pre-K assessments have helped to formulate a norm referenced set of tests. Each test lasts in length from 1-4 minutes, and each test has 33 forms to reduce the rate of test familiarity.

Pre-kindergarten, 4-year-old students in the experimental group received the independent variable, participation in the Gearing Up for Kindergarten school-readiness program. Pre-kindergarten, 4-year-old students in the control group did not receive the Gearing Up for Kindergarten school-readiness program. Students in both groups were assessed for knowledge in three school-readiness content areas with the AIMS Web assessment tool: letter identification, number identification, and number sense. The academic measurements of the two groups were compared to determine if there were differences in school readiness due to participation in the Gearing Up for Kindergarten school-readiness program.

## **Data Collection**

This quasi-experimental study used a research design with pre, post, and follow-up measurement data from AIMS Web assessment probes given to children in the Gearing Up for Kindergarten program (the treatment group) and the control group over a period of 12 months. This study also gave the Practical Parent Assessment for School Readiness assessment for parents in a pre, post, and follow-up design to parents of students in both the experimental and control groups. The research design was used to establish cause and effect between independent and dependent variables.

In the previously referenced September 22, 2010 letter, parents were given group assignment information and forms outlining the details for each group: treatment group, participation in the 16-week Gearing Up for Kindergarten series, and control group, completing assessment information on school readiness. Each group's parents were asked to bring their child to the parent center in a local elementary school on either October 4th or 5th for the treatment group, or on October 6th or 7th for the control group. On the assigned evenings, parents filled out the assessment forms with trained Gearing Up for Kindergarten staff members. Those forms included the PPASR and the Parent Response Form. The assessments were handed out with a number code identifying each family. The code linked the children and parents, and the data gathered were identified by the code. There was one sheet used with parent's name and address as well as the child's name. That page was taken out and placed in a locked cabinet along with the number code.

Parents signed an Informed Consent Agreement (Appendix I) to participate in the Gearing Up for Kindergarten study. This agreement indicated their understanding of the study, involvement in the study, and the purpose of the study. Parents were assured that this

study was confidential and that no names would be used in sharing the data. This study was not totally anonymous, and participants were informed, prior to the study, that there was no penalty for withdrawing from the study at any time.

Because the students in this research were minors, parental permission was obtained for participation in the research study. Both parental and student confidentiality were assured for participants in the study. IRB approval was granted under the umbrella of North Dakota State University Extension and the Gearing Up for Kindergarten research project (Appendix J).

During that same time, the children were assessed with the AIMS Web assessments for letter identification, counting, and number identification. Trained AIMS Web staff from the local school district conducted the three 1-minute assessment probes. Children were identified by a number code, the same as their parents, and no names were used in the assessments. The assessments each had a one-page score sheet. Each child's score sheets were clipped together with his/her number code. Upon completion of the assessments, parents were given a \$25 gift card to a local retailer with funds provided from a North Dakota State University research grant.

On separate days but in the same timeframe, treatment and control-group evenings were held to keep the assessment timeframe as close together as possible without the effects of group interference. Thirty-five families showed up for the treatment group, and 31 families showed up for the control group. Members who did not show up for the assessments on the assigned evenings were called and rescheduled for the following week. Any participants who did not show up that week were given one more opportunity during the second night of class for the treatment group and at the parent center one evening that

same week for the control group. The study was then closed to participants who signed up, but did not respond to the follow-up letters or phone calls. At that time, the treatment group had a total of 40 members (N = 40), or 89% of possible members, and the control group had 35 members (N = 35), or 78% of possible members. The total possible N was 90, with 45 members in each group. The size limit was determined by the number of student seats available for the treatment group, and then, a corresponding number was allotted for the control group.

Treatment group assessments were administered, post program, at the final Gearing Up for Kindergarten session on either March 14th or 15th. Student assessment took place throughout the evening. Assessors took the children into a separate classroom and administered the same 1-minute probes as the first assessment. Parent assessments were completed during the parent portion of the evenings. Once again, parents received a \$25 gift card for a local retailer. The treatment group had 35 completed assessments.

Post program control-group assessments were administered at the parent center in conjunction with a Gearing Up for Kindergarten Learning Fair on March 21, 2011. Parents took the assessments in Room 1 with trained Gearing Up staff administering the same assessments as in the fall. Child assessments were administered in Room 3 with the same trained AIMS Web assessors used in the fall assessment. Families then played games at the learning fair and received their gift card. The control group had 33 completed assessments.

The final parent assessments for both groups were given at the parent center on September 28 and 29, 2011. Parents were sent a letter refreshing their memory about the study (Appendix K), and all parents were invited to come to the parent center from 1:00-6:00 pm on either day to fill out the final assessments. Forty assessments were completed at

that time. Final assessments were then mailed to parents who did not attend that meeting, and 15 additional assessments were received. A deadline of November 11, 2011, was set for receipt of the final assessments. Parents who filled out the final assessment received a final \$25 gift card. The treatment group members completed a total of 30 assessments, and the control group members completed 28 assessments.

The same AIMS Web student assessments used in the Gearing Up for Kindergarten study are automatically given to students in the school district. Site administrators were able to access this year's data, with permission from school district administration. No further testing was done for the final student assessments.

The assessments administered in the fall of 2010 were used to establish student baseline academic-level measurements for both the control and treatment groups on the three school-readiness indicators prior to the start of the experimental group in the Gearing Up for Kindergarten program. Upon completion of the 16-session intervention phase for the treatment group, both student groups were assessed on the three school-readiness indicators a second time. Both groups entered kindergarten in the fall of 2011 where the third AIMS Web assessment was automatically given to each child. The AIMS Web scores were measured before the intervention, the pretest, and again after the intervention, a posttest. The third set of scores, in kindergarten, was intended to be used to compare levels of readiness skill sets between the treatment and control groups using an analysis of variance (ANOVA). The ANOVA is used to determine if an observed statistical difference is sufficiently larger than a difference that would be expected solely by chance (Gay et al., 2006)

After all participants completed each round of assessments, the assessments were sent to North Dakota State University's College of Human Development and Education. There, the assessments were analyzed and placed in a secure location. These documents only identified participants by the assigned number code. Documents will be destroyed 5 years after the study is completed.

The school district granted permission to use the AIMS Web database and purchased the required student space to test and record student data. These student data were transferred to each student's teacher at the beginning of the 2011 school year. Classroom space and permission to conduct the research study in the school district was also obtained from the Assistant Superintendent's office (Appendix L).

### **Researcher's Role**

The researcher's role for this study was to organize the study, obtain permission from the local school district and a children's center to hold classes in their buildings, recruit and hire instructors and assessors, obtain IRB approval, recruit participants for both the Gearing Up for Kindergarten experimental and treatment groups, supervise personnel, oversee finances, train facilitators, and oversee all aspects of programming to ensure a quality experience for parents and students The researcher also provided data analysis for the study data.

The researcher is the parent coordinator for the local school district, and works with the local Extension office and conducts programming for the region. She has held this position for eight years, and her duties include parent programming; outreach to incoming kindergarten families; parent instruction at Head Start and at an Air Force base;

development of new programs for parents and children; implementation of the "red," read everyday campaign; and awareness of community needs with regard to parenting.

The researcher has been an educator for 13 years. Prior to becoming parent coordinator, she taught third grade for 5 years.

The researcher has a bias towards parent involvement in children's growth and development, believing that parents are a child's first and most important teacher. The researcher did not teach the Gearing Up for Kindergarten program. The researcher did, however, get to know families during the course of the program. To avoid researcher bias, the researcher did not conduct any of the assessments with parents or children.

#### **Pilot Study**

The researcher conducted a pilot study in the spring of 2010 to better understand the Gearing Up for Kindergarten school-readiness program and to create an effective implementation model for the experimental study. A 10-week Gearing Up for Kindergarten pilot program was held at the local Head Start site with 2 classrooms that had 14 children in each class. The adapted 10-week pilot study was utilized as a model for the current 16-week study. Adaptations for this study were made after determining the weaknesses of the pilot model.

Student recruitment for the pilot study was drawn from the winter 2010 Head Start 4-year-old waiting list. Recruitment letters were sent to 49 potential participant families, with follow-up phone calls giving parents an opportunity to ask questions regarding the program. Families were invited again with a personal invitation from Head Start personnel. Twenty-eight families committed to the program, and 14 families completed the 10-week series.

The AIMS Web assessment tool was used to create baseline data of student academic knowledge. Trained assessors, who were familiar with the instruments and with assessing children, were hired to assess pre and post program. The first night of class, the parent questionnaires and assessments, the PPASR and Parent Response Form, were given by trained personnel to ensure instrumentation reliability. Survey instruments were lengthy, and administration the first night of class did not create a smooth introduction or a get-toknow-you period. The parent instructors hired for the program were familiar in working with Head Start parents and were comfortable speaking to parents.

## **Modifications Applied to Current Study**

The pilot study identified a lack of consistent attendance. Of the 28 families (N = 28) that began the Gearing Up for Kindergarten program, only 14 (n = 14), or 50%, completed eight (80%) of the sessions. The researcher attributed low attendance to a 10-week series with no breaks between sessions, a lack of weekly parent communication between sessions, and a weak parent-instruction component.

To address the lengthy 16 week series, this study divided the Gearing Up for Kindergarten program into two 8-week segments with a break between sessions. To increase communication and build relationships, a program coordinator was hired to communicate weekly with parents, by email, flyers, or phone calls. These strategies were used to keep Gearing Up for Kindergarten on parents' minds and to inform them about upcoming topics and events. Parents were also contacted if they missed a class. The program coordinator's job description included goals to establish a caring relationship and feelings of connectedness, for parents and children.
The parent curriculum was strengthened by adding two take-home activities to each Gearing Up for Kindergarten lesson. Two kindergarten teachers, who have taught the Gearing Up curriculum, created 32 hands-on, take-home lessons for parents to make in the parent sessions each week. During the parent sessions, parents learned the objectives of each topic, had discussions centered on each topic, and created two take-home activities relating to that topic. These activities were sent home for use with preschoolers during the week. Each activity was designed to reinforce an area of development and learning, and to encourage parent-child time together.

The two parent facilitators for the study were kindergarten teachers. Both individuals taught with an interactive instruction model and knowledge of kindergarten-readiness expectations.

To mitigate the lengthy assessment period for parents, a separate assessment night was conducted prior to the first and last nights of class. Parents were invited to a separate location to complete the assessments, and their children also participated in AIMS Web assessments. Parents received a \$25 gift card for a local retailer as compensation for their time. Children, upon completing the AIMS Web assessments, played in a classroom under adult supervision.

### **Ethical Considerations**

Parents were given the option of participating in the study and were also told there was no penalty for leaving the study. Participants were assured of confidentiality in their responses and could indicate a desire to see the results of the study. Assessments for both parents and children were kept in a locked cabinet at the researcher's office during the

assessment period, and then transferred to North Dakota State University and locked in a secure location where they will be kept for five years and then destroyed.

### Validity and Reliability

Participants were not chosen by random selection but, instead, in the order response forms were received; therefore, it was considered a convenience sample. The turnaround time from original letter of recruitment to the class start time was short, less than 1 month.

One possible threat to the validity of the research design was the natural maturation process of 4-year olds in a 12-month time period and the date on which they turned 4. Entry into school mandates that children turn five on or before July 15th the year they enter kindergarten.

Selection of the groups was also limited to children entering kindergarten in the school district in the fall of 2011. Participants' awareness of the program may have introduced threats that influence outcomes, such as environment, preschool exposure, and ability levels.

Instrumentation was controlled, as much as possible, with rigorous training for the AIMS Web assessors and a commitment for the duration of the research. Fidelity in administration for all assessment instruments was required of the parent educators when administering assessments to parents.

#### **Data Analysis**

This quasi-experimental study examined the following research questions related to school readiness and parent understanding of readiness:

## **Research Question 1**

The first research question was as follows: Is the change over time in PPASR scores for the treatment group parents different from the change over time for the control group parents for the 5 domains of child development?

The first research question addressed whether differences will be evident between treatment-group and control-group parents regarding perceptions of their child's development across multiple domains. Parental perceptions of children's school development were measured using instruments designed to assess a child's development across the 5 domains: approaches to learning, physical well-being and motor development, social and emotional development, cognition and general knowledge and language development. The domains were measured using the sub-scales in the Practical Parent Assessment of School Readiness (PPASR). This question was examined utilizing a repeated-measures analysis of variance (ANOVA) approach to compare pre- and postprogram participant scores and to explore interaction effects.

## **Research Question 2**

The second research question was as follows: Is the change over time in PPASR scores for the treatment group parents different from the change over time for the control group parents for the transition-to-school knowledge and familiarity?

The second research question replicated the first question; however, it focused specifically on a newly constructed transition-to-school knowledge construct as a dimension of school readiness. Parental perceptions of children's transition-to-school knowledge were assessed using the 8-item Child-School Knowledge sub-scale (included in the PPASR). This question was also studied utilizing a repeated-measures ANOVA

approach to compare pre-program and post-program participant scores and to explore interaction effects.

### **Research Question 3**

The third research question was as follows: Are there differences in change over time between treatment and control group in children's knowledge of letter identification, number identification, and oral counting as measured by AIMS Web scores?

The third research question addressed whether there was a difference between children's AIMS Web scores by treatment versus control group. The children in both groups were assessed, by trained evaluators, regarding their current knowledge of specific aspects of numeracy and literacy using three identified sub-scales in the AIMS Web measure, letter identification, number identification, and oral counting. A repeatedmeasures ANOVA was employed to analyze pre-program and post-program participant scores and to explore interaction effects.

### **Research Question 4**

The fourth research question was as follows: Is there a difference between treatment and control group in the strength of the relationship between the parent PPASR scores and child AIMS Web scores?

The final research question explored whether there were relationships between the parental perceptions of a child's school readiness scores (PPASR, etc.) and the child's AIMS Web scores in both the treatment and control groups. A correlation analysis was conducted to determine whether there were meaningful relationships between the children's scores on letter identification, number identification, and oral counting and the parent perception of their child's scores on the same measures in the PPASR. Those

measures were found under the cognition and general knowledge and language scales of the PPASR. The measure assessed for strength of the relationships between these scores.

A low level of correlation would be less than +.35 or -.35. A moderate level of correlation would be "between +.35 and +.65 or between -.35 and -.65" (Gay et al., 2006, p. 194). A high level of correlation exists when the scores are "higher than +.65 or -.65" (Gay et al., 2006, p. 194). "Coefficients of plus or minus .60 or .70 are usually considered adequate for group prediction purposes . . ." (Gay et al., 2006, p. 194).

Chapter 3 described the Methods and Procedures used in the study, including study design, program implementation, population and sample, instrumentation, data collection, pilot study, validity and reliability, ethical considerations, and methods of data analysis of the data as related to the four research questions. Chapter 4 will present the Analysis and Results of the study. Chapter 5 will summarize the study and provide Conclusions and Recommendations for further study.

## **CHAPTER 4. FINDINGS AND RESULTS**

This chapter contains the following sections: introduction to the chapter including statement of research questions, study participants, demographic information, research question 1 report and tables, research question 2 report, research question 3 report and tables, and research question 4 report and figures.

The purpose of this quasi experimental study was to examine the impact of the Gearing Up for Kindergarten program in shaping parent perceptions of child development and student readiness for the kindergarten transition. The study also examined three subtests of children's academic readiness. Three research questions were proposed in order to identify characteristics of child development and school readiness through the parent lens. The fourth question was designed to measure change over time in student growth on selected measures of academic readiness.

This study was designed to research the effectiveness of the Gearing Up for Kindergarten program from a parent understanding of child development and school readiness and to measure student gains in selected academic school readiness indicators

This quasi experimental study examined the following research questions:

- Is the change over time in the Practical Parent Assessment of School Readiness (PPASR) scores for the treatment group parents different from the change over time for the control group parents for the 5 domains of child development?
- 2. Is the change over time in PPASR scores for the treatment group parents different from the change over time for the control group parents for the transition-to-school knowledge and familiarity?

- 3. Are there differences in change over time between treatment and control group in children's knowledge of letter identification, number identification, and oral counting as measured by AIMS Web scores?
- 4. Is there a difference between treatment and control group in the strength of the relationship between the parent PPASR scores and child AIMS Web scores?

# **Participants**

The pretest PPASR assessment had 40 members in the treatment group and 35 members in the control group for a total of 75 participants. The posttest PPASR had 35 responses from the treatment group and 33 responses from the control group, for a total of 68 participants. The post posttest PPASR had 30 participants in the treatment group and 28 participants in the control group for a total of 58 participants.

The pretest AIMS Web assessment had 40 students in the treatment group and 35 students in the control group for a total of 75 participants. The posttest AIMS Web assessment had 35 students in the treatment group and 33 students in the control group for a total of 68 participants. The post posttest AIMS Web assessment had 35 students for the treatment group and 30 students in the control group for a total of 65 participants.

### **Demographic Data**

Frequencies were run on the Gearing Up for Kindergarten control and treatment groups for the following variables: parent and child gender, ethnicity, parent education, family status, employment status of parent, number of children in the family, relationship of parent filling out assessment, first child in kindergarten, Head Start eligibility, child experience in preschool, and in home daycare experience, Seventy five children participated in the study, with 35 in the control group and 40 in the treatment group. Parent Response Form found in Appendices.

Seventy-five parents filled out the assessments; the control group had 31 females (88.6%) and 3 (8.6%) males and 1 other (2.9%) complete the surveys. The treatment group had 34 females (85.0%) and 6 males (15.0%) complete the surveys. There were 20 boys in the participating in the children's control group, (57.1%) and 15 girls, (42.9%). The treatment group had 24 boys, (60%) of the group and 16 girls, (40%) of the group.

Family ethnicity was reported with Caucasian as the largest group represented in the study; 32 for the control group (91.4%) and 34 for the treatment group (85.0%). Native American or Alaskan Native was reported as 1 parent for the control group (2.9%) and 2 parents for the treatment group (5.0%). Asian was reported as 2 parents for the control group (5.7%) and 1 parent for the treatment group (2.5%). There were 2 Hispanic parents in the treatment group, (5.0%) and 1 other (2.5%).

The education level of participants varied greatly with the majority of parents reporting some college, 11 in the treatment group (27.5%) and 10 in the control group (28.6%) and a 4-year college degree, 11 in the treatment group (27.5%) and 13 in the control group (37.1%). Other reported education levels were: some high school 1 in the control group (2.9%) and 2 in the treatment group (5.0%). Completion of high school was reported by 2 parents in the control group (5.7%) and 4 parents in the treatment group (10.0%). A 2-year college degree was reported at 2 parents for the control group (5.7%) and 7 parents for the treatment group (17.5%). Participants who reported a master's degree or higher were 7 for the control group (20.0%) and five for the treatment group (12.5%).

Family status was listed for the following categories: married control group had 22 participants (62.9%) and the treatment group had 28 (70.0%), for single (never married) the control group 5 (14.3%) and treatment group 6 (15.0%), for separated/divorced the control group reported 4 (11.4%) and the treatment group reported 4 (10.0%), persons living together or cohabitating were reported at 3 for the control group (8.6%) and 2 for the treatment group (5.0%), lastly the control group had 1 remarried person (2.9%).

Parent employment status was reported with full-time work ranking the highest. The control group had 17 working full-time (48.6%) and the treatment group had 16 (40.0%). The control group had 3 (8.6%) people working 26-39 hours a week and the treatment group had 8 (20.0%) at working at that rate. Those working less than 25 hours a week had 1 member (2.9%) for the control group and 2 (5.0%) for the treatment group. Seeking employment had 6 in that category, 2 (5.7%) for the control group and 4 (10.0%) for the treatment group. Those not seeing employment consisted of 9 (25.7%) for the control group and 9 (22.5%) for the treatment group. The other category had 3 (8.6%) in the control group and 1 (2.5%) in the treatment group.

Families with two children were the largest grouping for number of children in a family. The control group had 16 families (45.7%) and the treatment group had 18 (45.0%). Three children in a family was the next largest category reported with the control group having 9 (25.7%) and the treatment group having 16 (40.0%). Three families (8.6%) reported one child in the control group and 2 families (5.0%) in the treatment group. Five (14.3%) control group families had four children and 3 (7.5%) families reported four children in the treatment group. Two families (5.7%) from the control group reported 5 children and 1 family (2.5%) reported 6 children in the treatment group.

Of the parents filling out the survey 32 (91.4%) in the control group and 34 (85.0%) in the treatment group were biological mothers. Fathers filling out the survey numbered 2 (5.7%) for the control group and 6 (15.0%) for the treatment group. There was 1 (2.9%) respondent in the other category for the control group.

Additional frequencies reported contain dichotomous data reporting yes responses. Fifteen (42.9%) of the families in the control group and 11 (27.5%) indicated the child in this study was their first child to enter kindergarten. Thirteen (37.1%) of the families in the control group and 10 (25.0%) of the families in the treatment group indicated eligibility for Head Start enrollment. Lastly, attendance averaged 26 children per week for the group. Thirty-four children attended the first class, 25 attended session 8, and 27 were in attendance for the last class.

## **Findings and Results from Analysis**

## **Research Question 1**

Aggregate scores for PPASR constructs. An ANOVA was run at p = .05 to determine if there were differences between the treatment and control groups in the change over time between the mean scores of the parents in the treatment and control groups on each of the parent rated PPASR constructs: approaches to learning, physical well-being and motor development, social development, cognition and general knowledge and language, and emotional development. See Table 1.

To further explore the program's effect on parents' perceptions of social development; all the items that comprised social development were examined and tested in an ANOVA. See Table 2.

# Table 1

	Treatment (n = 30) Baseline Follow-up				Control ( $n = 27$ ) Baseline Follow-up				Inter- action
Variable	M	SD	M	SD	M	SD	M	SD	Р
Approaches to Learning	4.02	.549	4.38	.450	4.06	.697	4.28	.657	.246
Physical Well-Being and Motor Dev	4.62	.472	4.85	.275	4.68	.336	4.85	.439	.523
Social Dev	3.85	.658	4.35	.528	3.92	.653	4.13	.860	.034
Cognition and Language	4.24	.609	4.77	.333	4.38	.585	4.70	.379	.095
Emotional Dev	4.09	.580	4.48	.466	4.14	.596	4.26	.587	.063

Practical Parent Assessment of School Readiness: Parent Perceptions of Their Child Aggregate Measures of Constructs

*Note:* P-values shown are for group X time interaction effects in repeated measures ANOVA. Analysis of variance did not show statistical significance for group by time interaction on the PPASR assessment instrument as a whole. The analysis of variance did show a significant group by time interaction effect on social development. The treatment group had greater gains in social development than the control group.

The group by time interaction effect on the emotional development construct, though not significant at a .05 level, was very close to significance at .063. To further explore the program's effect on parents' perceptions of emotional development, all the items that comprised emotional development were examined and tested in an ANOVA. See Table 3.

# Table 2

	Treatment $(n = 30)$				Control $(n = 27)$				Inter- action
Variable	Bas M	eline SD	Follo M	ow-up SD	Bas M	seline SD	Follo M	ow-up SD	Р
Shares with others	3.80	.805	4.33	.661	3.89	.801	4.00	1.03	.058
Works/Plays Cooperatively	3.93	.785	4.43	.679	4.00	.845	4.19	1.07	*.033
Listens in Group	3.87	.900	4.40	.621	4.00	.784	4.07	1.10	.098
Takes Turns	3.80	.925	4.37	.669	4.00	.734	4.04	1.05	*.038
Remembers and Follows Directions	4.17	.834	4.67	.479	4.11	.698	4.26	1.09	.116
Enjoys Talking/Playing	4.40	.814	4.67	.606	4.22	1.01	4.52	.893	.896
Gets Along	3.50	.861	4.07	.691	3.67	.920	3.96	.980	.395
Understands/ Follows Rules	3.80	.64	4.43	.626	3.85	.662	4.15	.907	.086
Understands How to Enter Talking	3.80	.997	4.13	.900	3.89	.974	4.22	.934	.336
Tries to Solve Problems	3.43	.935	4.07	.828	3.52	.975	3.93	.781	.156

Practical Parent Assessment of School Readiness: Parent Perceptions of Their Child's Social Development Measures of Construct Components

P-values shown are for group X time interaction effects in repeated measures ANOVA. \*p < .05. Analysis of variance showed a significant group by time interaction effect of parents' perceptions, on works or plays cooperatively with friends, F(2,110) = 3.53, p = .033 and takes turns with others, F(2,110) = 3.38, p = .038. The interaction effect was such that the treatment group had greater gains in two aspects of social development, works

or plays cooperatively with friends and takes turns with others, than the control group.

Table 3

Practical Parent Assessment of School Readiness: Parent Perceptions of Their Child's Emotional Development Measures of Construct Components

	Treatment $(n = 30)$				Control $(n = 27)$				Inter- action
Variable	Base M	eline SD	Follo M	w-up SD	Bas M	eline SD	Follo M	ow-up SD	Р
Feels Loved and Supported	4.87	.434	4.93	.254	4.78	.424	4.89	.424	.566
Spends Time Away W/out Anxiety	4.17	.913	4.67	.547	4.33	.832	4.59	.572	.405
Identifies Feelings	4.33	.844	4.67	.606	4.15	.864	4.44	.847	.287
Can Talk About Others Feelings	3.87	1.04	4.23	.817	3.93	.874	4.00	1.00	.503
Manages Feelings	3.50	.820	4.17	.747	3.81	.962	3.59	1.04	*.004
Adapts to New Situations	3.93	.785	4.40	.621	4.00	1.00	4.19	.921	.296
Shares Feelings with Adults	3.93	.907	4.33	.802	4.00	.784	4.07	.917	.111
Appreciates Getting to Know People	4.13	.900	4.50	.777	4.19	.786	4.33	8.32	.547

P-values shown are for group X time interaction effects in repeated measures ANOVA. \*p < .05. Analysis of variance showed a significant group by time interaction effect on manages feelings, such as anger or frustration, without hurting or being mean to others, F(2,110) = 5.87, p = .004. The interaction effect was such that the treatment group had greater a gain in emotional development, on the management of feelings, whereas the control group declined.

## **Research Question 2**

**Transition-to-school.** An ANOVA and a significance level of .05 was used to determine if there were differences, between the treatment and control groups in the change over time on the parents' perception of their child's transition-to-school. The treatment group consisted of 30 parents and the control group had 26 parents. The mean for the treatment group was 3.36 (SD = .616) at the pre-test and 3.86 (.259) on the post-test. The mean for the control group at the pre-test was 3.31 (.671) and on the post-test 3.78 (.358).

Analysis of variance did not show an interaction effect for the transition-to-school construct, F(2,108) = .30, p = .970

### **Research Question 3**

AIMS web. An ANOVA was run at p = .05 to determine if there were differences between the treatment and control groups in the change over time between the mean scores of the children in the treatment and control groups on each of the AIMS Web sub tests: letter identification, number identification, and oral counting. To explore the interaction effect of change over time in children's academic scores see Table 4.

### **Research Question 4**

A Pearson product-moment correlation coefficient was computed to assess the relationship between the parent PPASR scores and the children's AIMS Web scores.

A weak correlation would be less than +.35 or -.35. A moderate correlation would be between +.35 and +.65 or between -.35 and -.65. A strong correlation exists when the scores are higher than +.65 or -.65. (Gay et al., 2006, p. 194).

## Table 4

	Treatment (n = 36) Baseline Follow-up				Base	Inter- action			
Variable	М	SD	М	SD	М	SD	М	SD	Р
Letter Identification	8.30	10.7	17.80	15.23	15.84	11.8	27.75	17.91	.373
Number Identification	19.44	17.44	31.30	16.80	27.31	17.44	36.56	16.79	.278
Oral Counting	26.00	19.12	35.25	22.99	27.78	18.20	42.00	22.98	.249

## AIMS Web: Aggregate Measure of Children's Academic Scores

*Note.* P-values shown are for group X time interaction effects in repeated measures. \*p < .05. Analysis of variance did not show an interaction effect for letter identification, F(1, 66) = .804, p = .373; number identification F (1, 66) = 1.19, p = .278; or oral counting F(1, 66) = 1.35, p = .249.

A Pearson product-moment correlation coefficient was computed to assess the relationship between the parent PPASR scores and the children's AIMS Web scores on the letter identification post-test. A scatterplot summarizes the results (Figure 1).

In the control group, student letter naming fluency and the parent perception that the child is familiar with (or able to recognize) letters of the alphabet had a weak positive correlation, Pearson's r(32) = .199, p < .05.

In the treatment group, student letter naming fluency (letter identification) and the parent perception that the child is familiar with (or able to recognize) letters of the alphabet had a moderate positive correlation, Pearson's r(35) = .418, p. < 05.



*Figure 1*. Parent perceptions of their child's letter identification abilities with the child's actual letter identification (indicated in Figure 1 as letter naming fluency) score on the posttest.

In the treatment group, the parent perceptions of their child's letter identification were more strongly related (.418) to the children's actual letter identification scores than they were in the control group (.199).

There was no significant difference between these two correlation coefficients,

*Z* = .958, *p* = .169.

A Pearson product-moment correlation coefficient was computed to assess the

relationship between the parent PPASR scores and the children's AIMS Web scores on the

number identification post-test. A scatterplot summarizes the results (Figure 2).



*Figure 2*. Parent perceptions of their child's number identification abilities with the child's actual number identification score on the post-test.

In the control group, student number identification and the parent perception that the child is familiar with (or able to recognize) numbers between 1 and 10 (or higher) had a moderate positive correlation, Pearson's r(33) = .418, p < .05.

In the treatment group, student number identification and the parent perception that the child is familiar with (or able to recognize) numbers between 1 and 10 (or higher) had a strong positive correlation, Pearson's r(35) = .608, p < .05.

In the treatment group, the parent perceptions their child's number identification scores were more strongly related (.608) to the children's actual number identification scores than they were in the control group (.418).

There was no significant difference between these two correlation coefficients, Z = 1.016, p = .154.

A Pearson product-moment correlation coefficient was computed to assess the relationship between the parent PPASR scores and the children's AIMS Web scores on the oral counting post-test. A scatterplot summarizes the results (Figure 3).



*Figure 3*. Parent perceptions of their child's oral counting abilities with the child's actual oral counting score on the post-test.

In the control group, student oral counting measure and the parent perception that the child is able to do counting had a weak positive correlation, Pearson's r(33) = .217, p < .05.

In the treatment group, student oral counting measure and the parent perception that the child is able to do counting had a moderate positive correlation, Pearson's r(35) = .469, p < .05

In the treatment group, the parent perceptions of their child's oral counting were more strongly related (.469) to the children's actual oral counting scores than they were in the control group (.217).

There was no significant difference between these two correlation coefficients, Z = 1.134, p = .128.

# **CHAPTER 5. SUMMARY, CONCLUSIONS, AND RECOMMENDATIONS**

The first purpose of this study was to investigate if parent perceptions showed that the Gearing Up for Kindergarten, child development and school-readiness, program played a significant role in helping parents prepare their child for school, at kindergarten levels of readiness, as reviewed in the Chapter 2 literature. In addition to children's developmental school readiness, the study investigated children's readiness to make the transition to school. The final purpose of this study was to investigate if children's readiness skills, on selected academic measures, were significantly impacted through participation in the Gearing Up for Kindergarten program as reviewed in the Chapter 2 literature.

# Summary

## **Research Question 1**

Is the change over time in the Practical Parent Assessment of School Readiness (PPASR) scores for the treatment group parents different from the change over time for the control group parents for the 5 domains of child development?

This study, as a whole, did not find the Gearing Up for Kindergarten program as being significant in changing parents' perceptions of their children's readiness for school in 4 of the 5 domains of child development.

The lack of random assignment for the groups was quite likely the most limiting factor in this study's lack of finding significant results in the change over time in parent perceptions of child development. The small number of participants may also have contributed to the lack of significance.

As measured by the ANOVA, there were significant differences found in one domain, social development. The scores obtained on the measure of social development indicated the significance was due to stronger social skill development in the area of being able to work or play cooperatively with friends and taking turns with friends. The measures of sharing and understanding and following rules were not significant, but did measure near the p < .05 significance level.

The social constructivist theory of development (Vygotsky, 1978) supports the value of learning in social settings. The Gearing Up for Kindergarten program offers learning opportunities through interaction with other children and parents in a familiar setting designed to enhance children's experiences and expand knowledge of the world around them.

Research demonstrates children learn best in group settings where taking turns and sharing is expected and emphasized (NAEYC, 2009; Sprenger, 2008). The Gearing Up for Kindergarten program emphasizes the social aspect of children's development. Through participation in the 16 week Gearing Up for Kindergarten program children were exposed to a lengthy time period of expected social behaviors. This program component, along with a period of time to learn and develop social skills is likely to have contributed to the significant difference in social development between the two groups.

As measured by the ANOVA, there were near significant differences found in one domain, emotional development. The scores obtained on the measure of emotional development indicated the near significance was due to stronger skill development in the area of being able to manage feelings.

The literature supports that children who are taught how to understand and process emotions experience a greater ability to appropriately express feelings (NAEYC, 2009; North Dakota Department of Human Services, 2008a; 2008b; and Sprenger, 2008). The

Gearing Up for Kindergarten program emphasizes the emotional aspect of children's development. Through participation in the 16 week Gearing Up for Kindergarten program children were exposed to a lengthy time period of appropriate emotional behaviors. This program component, along with time to learn and develop and practice those skills is likely to have contributed to the near significant difference between the two groups.

Research by Sprenger (2008) and the North Dakota Department of Human Services (2008a; 2008b), validates the importance of a healthy social and emotional environment for children. The quality of the home environment directly influences children's ability to relate to and interact with others in a healthy manner. Children who are nurtured and loved are better able to respond in a healthy manner emotionally and socially (High/Scope Educational Research Foundation, 2006). In addition, children who are regularly exposed to taking turns, being acknowledged, and listened to, have greater skill ability to learn and exhibit those behaviors in outside settings.

#### **Research Question 2**

Is the change over time in PPASR scores for the treatment group parents different from the change over time for the control group parents for the transition-to-school knowledge and familiarity construct?

The analysis of variance did not show significant difference in the transition-toschool construct between the treatment and control groups. The researcher did not expect this result, but also realizes the central focus of the Gearing Up for Kindergarten program is not on the transition into kindergarten, but preparation for success in the school setting. The program curriculum does not contain specific lessons designed to address the transition into kindergarten. There are also variations in where the Gearing Up for Kindergarten program

is held. Program locations in this study were not at the schools most children attended for kindergarten. The study, did however, add information for parents to support school transition preparation and shared the dates of kindergarten registration along with opportunities to participate in a summer program, Intro to Kindergarten.

The overlapping spheres of influence theory supports the shared role of schools and families in the transition process, and beyond (Epstein, 1987b). The assessment of this theory explored the importance of parent knowledge in recognizing their place in the transition process from home to school.

Bronfenbrenner's (1979) research emphasized the importance of support for children as they move from a small area of interaction, the home and daycare, to a larger area of interactions and expectations at school. Fabian and Dunlop (2007) reported though many schools recognize the importance of support in transitions, few schools or programs assign responsibilities to ensure this support mechanism is in place for families (p.3).

#### **Research Question 3**

Are there differences in change over time between treatment and control group in children's knowledge of letter identification, number identification, and oral counting as measured by AIMS Web scores?

The analysis of variance did not show significant differences in change over time on any of the AIMS Web scores between treatment and control groups on the selected academic measures.

This lack of significant effect could be due to a variety of other factors as well as the lack of random assignment. The mean of children in the control group was higher on all

of the pretest measures; therefore there was less room for significant growth as each measure had a ceiling limit.

The post-test mean for letter identification was higher for the treatment group than the control group. The mean for number identification and oral counting was higher for the control group than that of the treatment group. Each group made progress from the beginning of the study to the conclusion. This study did not take into account children's expected growth over time and parent high ratings at the beginning which could have influenced study results. The small n in the control group could also have contributed to a lack of significance in the results. A recommended minimum for a study is at least 30; the control group did not reach that number.

Lack of significance could also be attributed to the teachers used in this study. The teachers at the children's center had strengths in the domain of children's development with regard to play and social skills. Their instruction in the academic section of class left room for improvement. The researcher spent time reviewing the curriculum with them and working to strengthen instruction in that area. The instructors were slow to respond to a stronger academic focus and as a result, the students did not receive the best instruction possible in that area. The students in the school, however, had an instructor who incorporated academics in all aspects of the evening, play, reading, and instruction.

Due to the small n of the study, there was not the ability to effectively measure the difference in classroom instructors. Fielding et al. (2004) and the researcher believes the quality of targeted instruction can make a significant difference in student achievement.

Another possible factor in the control group's higher scores was the education of the parent group; the control group had fewer numbers, but a greater percentage of parents

with four year and master's degrees. Parent education levels are known to be direct correlates of children's academic scores (Rhode Island KIDS COUNT, 2005).

#### **Research Question 4**

Is there a difference between treatment and control group in the strength of the relationship between the parent PPASR scores and child AIMS Web scores on post-test measures? There was not a significant difference between treatment and control group parents in the strength of the relationship between their perceptions of their child and their child's scores on the AIMS Web assessment.

Parents who participated in the Gearing Up for Kindergarten program were expected to have a better sense of their child's academic development than those in the control group who did not participate in the program. The lack of random assignment most likely limited the effect of the program in producing significant results.

The treatment group parents' perceptions of the three academic measures, letter identification, number identification, and oral counting, showed more accurate recognition of their child's academic abilities than did the control group parents. Though the treatment group parents more accurately perceived their children's abilities, in all cases there was no significant difference between treatment and control group parents on perceptions of their child's academic abilities.

It is possible that treatment group parents had a better sense of their child's academic abilities due to the activities in the classroom and the parent education component of the program. The Gearing Up for Kindergarten program does not have lessons identifying specific academic readiness skills for kindergarten though some skills

were practiced. This factor could have influenced parent understanding of the academic measures and the role academics plays in school readiness.

The strengths of this study were the opportunities provided for parents and their preschoolers to participate in the Gearing Up for Kindergarten program. This was not a qualitative study, but many positive comments from parents reinforced the importance of a program of this nature. Parent comments included references to education not requiring a significant investment of money, but use of things already in the home. Other comments included the importance of understanding what specifically constitutes school readiness, learning that education takes place every day in every way, the value of understanding children's temperaments and learning styles, and the time spent one on one with children.

The limitations in this study were primarily the lack of random assignment, but there were others. The researcher experienced a change of advisors during the course of the study and this was the first full implementation of the program. The teachers at the children's center were also tied to grant dollars to implement the program. Under normal conditions they would not have been hired for the program. In addition, this was the first major research project of the researcher. A great deal of learning has taken place and changes will be made for future research.

## Conclusions

This study did not establish overall significance of the Gearing Up for Kindergarten program. The research findings however on social development corroborate with previous research on the Gearing Up for Kindergarten program. This program component was further strengthened with this study (Brotherson et al., 2007; Brotherson et al., 2009). Previous findings also show program significance in the area of cognition and general

knowledge (Brotherson et al., 2009) and were expected in this research. This study did not support those findings and it is the researcher's belief that lay with the lack of teacher quality and expertise at the children's center factored in with first time delivery of the Gearing Up for Kindergarten program to the entire community.

Though overall significance was not found, the researcher supports the program concept. The program's target audience, preschool children and their parents do not have adequate early childhood education resources available in their state. Due to the lack of formal state funding for preschool children, this program can help in a small way, to close the gap of this underserved population. This study served approximately 5% of the district's preschool student population. While the percentage served was small, there are still 40 students who benefitted from participation in the program. This program replicated the school environment with procedure, routine, group activities, school readiness activities and the use of actual classrooms. Though there was no measurable significance, this exposure to school practices likely made a difference in both parent and child readiness for school.

A benefit of this study is the first time inclusion of a children's assessment. Though this measure did not produce significance, it did provide an opportunity to measure specific aspects of children's academic growth over time and to explore the merit of a children's assessment. The third student assessment was not administered with fidelity of implementation and was not able to be used in the data analysis for the study. A third wave of assessment would allow for greater measurement of the program's impact on student academic achievement and further opportunity to research the correlation between children's scores and parent perceptions of their child.

This program, though not a daily program, does provide a learning environment in which teachers and parents have the opportunity to experience children's growth over a period of time. Through the program teachers can address academic and developmental growth with a variety of educational activities. One of the teacher practices in this study, children writing their names each week, allowed for the children and parents to observe growth over time and the ability to have a clearer picture of the progress on that specific skill.

It is the opinion of the researcher that the small n impacted the lack of significance along with the instructor component at the children's center. Time constraints did not allow for participants in this study to be chosen randomly. Those factors contributed to a lack of this study's ability to be generalizable to other pre-school populations.

The social constructivist theory of child development was supported in this study along with the emphasis of the shared role of parents, educators, and community in raising a child. The overlapping spheres of influence theory was reinforced with the Gearing Up for Kindergarten instruction model. The new transition-to-school theory was introduced in this study as a piece of the foundation for support in transitions. Lack of significant findings in this area do not negate the importance of future research in this area. One aspect of the research clearly demonstrated lack of a formal system for integrating children into the school environment (Wesley & Buysse, 2003) and perhaps will spark additional discussion between the local preschools and school community in addressing this issue.

This study introduced the Gearing Up for Kindergarten program to the community for the first time and as a new program did not generate as much response as the researcher had hoped. At the time of the conclusion of this report, the school district has a waiting list

for next year's students. The program has grown in its recognition and popularity in this community and in the state where it has received endorsement from other school districts. **Recommendations** 

**Curriculum realignment.** Recommendations for future implementation of the Gearing Up for Kindergarten program based on the research presented in the literature include: curriculum alignment with stronger emphasis on skills for the 5 domains of children's development, specific strategies for transition to kindergarten and targeted instruction for parents on expected academic readiness skills for incoming kindergarteners.

The social and emotional development of children is a significant strength of the Gearing Up for Kindergarten program; as reflected in the statistical difference between the treatment and control group. The expanded curriculum model should include the addition of specific skill practices aligning with the other domains of development; approaches to learning, physical well-being and motor development, cognition and general knowledge, and language development. The use of NAEYC (2009) standards, North Dakota Department of Human Services (2008a; 2008b) research, and NEGP (1995) skills and strategies would incorporate well into the current curriculum design.

The recommendations for the Gearing Up for Kindergarten program preschool and parent curriculum includes expansion for specific lessons integrating basic academic skills in language and numeracy. The curriculum could add a letter/sound component focusing on recognizing the letters of the alphabet and their basic sounds and a mathematical component for number identification, one-to-one correspondence, and oral counting. Future student and parent instruction should incorporate incoming kindergarten standards as stated by the North Dakota Department of Public Instruction. (North Dakota Department of

Public Instruction 2005a; North Dakota Department of Public Instruction 2005b). The addition of these features would corroborate with the North Dakota Department of Public Instruction's language and math expectations for incoming kindergartners and give parents the knowledge and skill necessary for adequate instruction and preparation the year before kindergarten.

The researcher recommends a greater emphasis built into the Gearing Up for Kindergarten program to strengthen support for the transition process as researched by Brofenbrenner (1979). The research hypotheses recommended support in the transition process through information regarding expectations, logistical information and practices, and that the transition not be made alone (Bronfenbrenner, 1979, p. 217). This information should be integrated into the parent curriculum lessons and shared with the Department of Public Instruction as the Gearing Up for Kindergarten program is promoted in the state.

Gearing up expansion. Future Gearing Up for Kindergarten programming should also take into consideration expansion into daycares and preschools. Preschool programming may not include a direct parent education and involvement component. The parent participation emphasis reflected in this program could greatly enhance parent understanding of child development and school readiness, and in turn increase parent involvement in children's development and learning. This expansion would also educate daycare providers in the importance of parent involvement in child development and school readiness.

**Closing thoughts.** The field of early childhood education is one of national importance and currently has the attention of educators everywhere. This program has a significant opportunity to create impact in children's readiness for school, transition into

school, and success in life. Educational reform will not go away and continued scrutiny will be necessary as standards for education increase along with the diversity of the learning population. This program brings a unique combination of parent/child programming that fosters children's growth and development along with strengthening the family relationship. The motto at an elementary school in the researcher's district is to produce lifelong learners and responsible citizens. Parents afforded the opportunity to participate in this program have the privilege of showing responsibility to their families by attending a lengthy series of classes and implementing the learned skills in their parent practices. The children have the opportunity to experience positive parent models and opportunities to begin school with a rich learning history, and best of all, ready to learn at expected levels of kindergarten readiness.

### REFERENCES

- Ackerman, D. J. & Barnett, S. W. (2005). Prepared for kindergarten: What does "readiness" mean? New Brunswick: Rutgers, The State of New Jersey, National Institute for Early Education Research
- AIMS Web. (2011). Assessment and data management for RTI. Retrieved from http://aimsweb.com
- Appleseed. (2006). It takes a parent: Transforming education in the wake of the No Child Left Behind Act. Retrieved from www.appleseednetwork.org
- Barnett, W. S. (2007). *Early education: A sound public investment*. Paper presented at the 2007 Tennessee Early Childhood Summit. Retrieved from http://www.state.tn.us/education/prek/documents/Early\_ED\_Invest.pdf
- Barnett, W. S., Epstein, D. J., Friedman, A. H., Boyd, J., & Hustedt, J. T. (2008). The state of preschool: 2008 state preschool yearbook. New Brunswick: Rutgers, The State University of New Jersey, National Institute for Early Education Research.
- Barnett, W. S., & Hustedt, J. T. (2005). Head Start's lasting benefits. *Infants & Young Children, 18*(1), 16-24. New Brunswick: Rutgers, The State University of New Jersey, National Institute for Early Education Research.
- Brofenbrenner, U. (1979). *The ecology of human development: Experiments by nature and design*. Cambridge, MA: Harvard University Press.
- Brotherson, S. E. (2007). *Gearing Up for Kindergarten: A school readiness and parent education program* [Impact Report]. Retrieved from North Dakota State University Extension web site: http://www.ag.ndsu.edu/gearupkindergarten

- Brotherson, S. (2011). Parental knowledge and skills grow through Gearing Up for Kindergarten. Retrieved from http://www.ag.ndsu.edu/impactreports/ 2011impactreports/11state-s.brotherson2-hdfs.pdf/view
- Brotherson, S. E., Query, S., & Saxena, D. (2007). Countdown to Kindergarten Ready to Learn: Project overview & year-end report for 2006-2007. Retrieved from http://www.ag.ndsu/gearupkindergarten
- Brotherson, S. E., Query, S., & Saxena, D. (2008). Countdown to Kindergarten Ready to Learn: Project overview & year-end report for 2007-2008. Retrieved from http://www.ag.ndsu/gearupkindergarten
- Brotherson, S., Query, S., & Saxena, D. (2009). Gearing Up for Kindergarten: Project overview & year-end report for 2008-2009. Retrieved from http://www.ag.ndsu/gearupkindergarten
- Catsambis, S. (2002). Expanding knowledge of parental involvement in children's secondary education: Connections with high school seniors' academic success. *Social Psychology of Education*, *5*(2), 149-177.
- Christenson, S. L., Godber, Y., & Anderson, A. R. (2005). Critical issues facing families and educators. In E. N. Patrikakou, R. P. Weissberg. S. Redding, & H. J. Walberg (Eds.), *School-family partnerships for children's success* (pp. 21-39). New York, NY: Teacher's College Press.
- Christenson, S. L., & Sheridan, S. M. (2001). Schools and families: Creating essential connections for learning. New York, NY: Guilford Press.
- Creswell, J. W. (2008). *Educational research planning, conducting, and evaluating quantitative and qualitative research*. Upper Saddle River, NJ: Pearson Education.

- De Jong, P. F., & Leseman, P. P. M. (2001). Lasting effects of home literacy on reading achievement in school. *Journal of School Psychology*, *29*(5), 389-414.
- Department of Public Instruction. (2005a). North Dakota English language arts content and achievement standards: Kindergarten. Bismarck, ND: Author.
- Department of Public Instruction. (2005b). North Dakota mathematics content and achievement standards: Kindergarten. Bismarck, ND: Author.
- Dickinson, D. K., & McCabe, A. (2001). Bringing it all together, the multiple origin, skills, and environmental supports of early literacy: Emerging skills and environmental supports. *Learning Disabilities Research and Practice*, *16*(4), 186-202.
- Early school readiness. In *Child trends data bank*. Retrieved from http://www.childtrendsdatabank.org/?=node/291
- Epstein, J. L. (1986). Parents' reaction to teacher practices of parental involvement. *The Elementary School Journal*, *86*, 277-294.
- Epstein, J. L. (1987a). Parent involvement: What research says to administrators. *Education and Urban Society, 19,* 119-136.
- Epstein, J. L. (1987b). Toward a theory of family-school connections: Teacher and parent involvement. In K. Hurrelmann, F. X. Kaufmann, & F. Losel (Eds.), *Social intervention: Potential and constraints* (pp. 121-136). Berlin, Germany: deGruyter.
- Epstein, J. L. (1995) School/family/community partnerships—caring for the children we share. *Phi Delta Kappan*, *76*(9), 701-712.
- Epstein, J. L. (2001). School, family, and community partnerships: Preparing educators and improving schools. Boulder, CO: Westview Press.

Epstein, J. L. (2006). *Interview with Joyce L. Epstein*. Retrieved from http://www.preknow.org/documents/nationalcalls/200604\_JEpstein\_resource.pdf

- Epstein, J. L., Sanders, M. G., Simon, B. S., Clark Salinas, K., Rodriguez Jansorn, N., & Van Voorhis, F. L. (2002). School, family, and community partnerships: Your handbook for action. Thousand Oaks, CA: Corwin Press.
- Espinosa, L. M., Thornburg, K. R., & Mathews, M. C. (1997). Rural kindergarten teachers' perceptions of school readiness: A comparison with the Carnegie study. *Early Childhood Journal*, *25*(2), 119-125.
- Fabian, H., & Dunlop, A. (2007, May). Outcomes of good practice in transition processes for children entering primary school[Working paper 42]. The Hague, the Netherlands: Bernard van Leer Foundation.
- Fantuzzo, J., Tighe, E., & Childs, S. (2000). Family involvement questionnaire: A multivariate assessment of family participation in early childhood education. *Journal of Educational Psychology*, 92(2), 367-376.
- Fielding, L. (2006, April). Kindergarten learning gap. *American School Board Journal*, 193(4), 32-34.
- Fielding, L. (2009). Extraordinary parents. Kennewick, WA: New Foundation Press.
- Fielding, L., Kerr, N., & Rosier, P. (2004). Delivering on the promise. Kennewick, WA: New Foundation Press.
- Fuchs, L. S., (2004). The past, present, and future of curriculum-based measurement research. *School Psychology Review*, *33*(2), 188-192.
- Fuchs, L. S., & Fuchs, D. (2004). What is scientifically-based research on progress monitoring? Washington, DC: National Center on Student Progress Monitoring.

- Gay, L. R., Mills, G. E., & Airasian, P. (2006). Educational research: Competencies for analysis and applications (8th ed.). Upper Saddle River, NJ: Pearson Education.
- Gebeke, D. (2010) *School readiness and parental involvement: Do they matter?* (Research Brief June 2010). NDSU Extension, Fargo, ND.
- Gestwicki, C. (2004). *Home, school, and community relations*. Clifton Park, NY: Delmar Learning.
- Ginsburg, H. P., Lee, J. S., & Boyd, J. S. (2008) Mathematics education for young children: What it is and how to promote it. *Social Policy Report*, 22(1), 1, 3-11, 14-23.
- Gouley, K. K., Brotman, L. M., Huang, K-Y., & Shrout, P. E. (2008). Construct validation of the social competence scale in preschool-age children. *Social Development*, 17(2), 380-398.
- Green, S. D., Peterson, R., & Lewis, J. R. (2006). Language and literacy promotion in early childhood settings: A survey of center-based practices. *Early Childhood Research and Practice*, 8(1), para.8-9.
- Hanson, S. (2008). Public prekindergarten: An implementation model for public schools in states that do not fund prekindergarten. (Unpublished doctoral dissertation).
  University of North Dakota, Grand Forks.
- Harvard Family Research Project. (2006a, Spring). Family involvement in early childhood education [Research Brief]. Cambridge, MA: Harvard Graduate School of Education.
- Harvard Family Research Project. (2006b). *Family involvement makes a difference in school success* [Research Brief]. Cambridge, MA: Harvard Graduate School of Education.
- Henderson, A. T., (2007, March 28). Effective strategies for engaging parents and communities in schools. Providence, RI: Annenberg Institute for School Reform.
- Henderson, A. T., & Mapp, K. L. (2002). *A new wave of evidence*. Austin, TX: Southwest Educational Development Laboratory.
- High, P. C., the Committee on Early Childhood, Adoption, and Dependent Care; & the Council on School Health. (2008, April). School readiness. *Pediatrics: Official Journal of the American Academy of Pediatrics, 121*(4), 1007-1016.
- High/Scope Educational Research Foundation. (2006). From implementation to impact: An evaluation of the South Carolina First Steps to School Readiness Program: 2006.
  Ypsilanti, MI: Author.
- Hinkle, D. (2000). School involvement in early childhood. Jessup, MD: U.S. Department of Education Office of Educational Research and Improvement.
- Jordan, C., Orozco, E., & Averett, A. (2001). *Emerging issues in school, family, & community connections*. Retrieved from Southwest Educational Development Laboratory web site: www.sedl.org/connections
- Kagan, S. L., Moore, E., & Bredekamp, S. (Eds.). (1995). *Reconsidering children's early development and learning: Toward common views and vocabulary*. (National Education Goals Panel). Retrieved from http://govinfo.library.unt.edu/negp/reports/child-ea.htm

- Kagan, S. L., & Rigby, E. (2003). Policy matters: Setting and measuring benchmarks for state policies. Washington, DC: Center for the Study of Social Policy.
- Kamerman, S. B., & Gatenio-Gabel, S. (2007). Early childhood education and care in the United States: An overview of the current policy picture. *International Journal of Childcare and Education Policy*, 1(1), 23-24.
- Klein, L. G. (2002, Summer). The Kaufman early education exchange. Set for Success:
   Building a Strong Foundation for School Readiness Based on the Social-Emotional
   Development of Young Children, 1(1), 1.
- Marzano, R. J., Pickering, D. J., & Pollock, J. E. (2001). *Classroom instruction that works*. Alexandria, VA: Association for Supervision and Curriculum Development.
- Merriam, S. B., Caffarella, R. S., & Baumgartner, L. M. (2007). Learning in adulthood: A comprehensive guide. San Francisco, CA: Jossey-Bass.
- National Association for the Education of Young Children. (2009). *Developmentally appropriate practice in early childhood programs serving children from birth through age 8*. Retrieved from http://www.naeyc.org/files/naeyc/file/positions/ position%20statement%20Web.pdf
- National Education Goals Panel. (1991). *Goal 1 technical planning group report on school readiness*. Washington, DC: U.S. Department of Education.
- National Education Goals Panel. (1995). *The National Education Goals report: Building a nation of learners*. Washington, DC: U.S. Department of Education.
- National Governors' Association. (2005). Building the foundation for bright futures. Washington, DC: Author.

National Institute for Early Education Research. (2004). *Debunking the myths: Benefits of preschool*. Retrieved from http://nieer.org/resources/facts/sheets/13.pdf

National Scientific Council on the Developing Child, Center on the Developing Child. (2007). *The science of early development*. Cambridge, MA: Harvard University. Retrieved from http://www.developingchild.net

No Child Left Behind Act of 2001. Pub. L. No. 107-110, 115 Stat. 1425 (2002).

- North Dakota Department of Human Services. (2008a). North Dakota early learning guidelines: Ages 3 through 5. Bismarck, ND: Author.
- North Dakota Department of Human Services. (2008b). North Dakota early learning guidelines: Birth to 3 [Adopted and used with permission of Minnesota Department of Health and Minnesota Department of Human Services]. Bismarck, ND: Author.
- Patrikakou, E. N. (1997). A model of parental attitudes and the academic achievement of adolescents. *Journal of Research and Development in Education*, *31*(1), 7-26.
- Patrikakou, E. N. (2004). Adolescence: Are parents relevant? *The Harvard Family Research Project: Family involvement network of educators*. Retrieved January 2, 2010, from http://www.gse.harvard.edu.hfrp/projects/fine/resources/digets/ adolescence.html
- Patrikakou, E. N., Weissberg, R. P., Redding, S., & Walberg, H. J. (Eds.). (2005a). School-family partnerships for children's success. New York, NY: Teacher's College Press.

- Patrikakou, E. N., Weissberg, R. P., Redding, S., & Walberg, H. J. (2005b). School-family partnerships: Enhancing the academic, social, and emotional learning of children. In E. N. Patrikakou, R. P. Weissberg, S. Redding, & H. J. Walberg (Eds.), *School-family partnerships for children's success* (pp. 1-17). New York, NY: Teacher's College Press.
- Peisner-Feinberg, E. S., & Schaaf, J. M. (2007). Evaluation of the North Carolina more at four pre-kindergarten program: Children's outcomes and program quality in the fifth year (2005/2006). Retrieved from http://www.fpg.unc.edu/~mafeval/pdfs/ year\_5\_finalreport.pdf
- Piotrkowski, C. S., Botsko, M., & Matthews, E. (2001). Parents' and teachers' beliefs about children's school readiness in a high-need community. *Early Childhood Research Quarterly*, 15(4), 537-558
- PNC Financial Services Group, Inc. (2007). *PNC study of early childhood education*. Retrieved from https://www.pnc.com/webapp/unsec/Requester?resource=/wcm/ resources/file/eb2e1343e2cbe74/PNCStudy\_ParentTeacherFindings.pdf
- Poole, J. M. (2000). Fitness and the young child. *Childcare Information Exchange*, *131*, 41-56.
- Popkin, M. H. (1996). 1 2 3 4 Parents! Marietta, GA: Active Parenting Publishers.
- Raver, C. C., & Knitzer, J. (2002). Ready to enter: What research tells policy makers about strategies to promote social and emotional school readiness among three- and fouryear olds. New York, NY: National Center for Children in Poverty, Mailman School of Public Health, Columbia University.

- Rhode Island KIDS COUNT. (2005, February). *Getting ready: Findings from the National* School Readiness Indicators Initiative: A 17 State Partnership. Retrieved March 3, 2010, from www.gettingready.org
- Rockwell, R. E., Andre, L. C., & Hawley, M. K. (1996). *Parents and teachers as partners*. Fort Worth, TX: Harcourt Brace & Company.
- Schulting, A. (2008). Promoting parent-school relationships during the transition to kindergarten. *Harvard Family Research Project: The Evaluation Exchange*, 14(1 & 2), 8.
- Shinn, M. R. (2002). Best practices in using curriculum-based measurement in a problem-solving model. In A. Thomas & J. Grimes (Eds.), *Best practices in school psychology* (4th ed., pp. 671-697). Bethesda, MD: National Association of School Psychologists
- Shonkoff, J. (2007) Social, emotional, and cognitive development are inextricably intertwined. *Preschool Matters*, 5(5), 5.
- Sprenger, M. (2008). *The developing brain: Birth to age eight*. Thousand Oaks, CA, Corwin Press.
- Theories of Child Development and Learning. (n. d.). Washington, DC: American Institute for Research. Retrieved from http://www.ncrel.org/sdrs/areas/issues/students/ earlycld/ea7lk18.htm
- U. S. Department of Education. (1995). *Goals 2000: A progress report*. Washington, DC: Author.
- Vygotsky, L. S. (1978). *Mind in society: The development of higher psychological processes*. Cambridge, MA: Harvard University Press.

- Vygotsky, L. S. (1988). *Thought and language* [rev. and ed. by A. Kozulin]. Cambridge, MA: MIT Press.
- Weiss, H., Caspe, M., & Lopez, E. (2006). Family involvement makes a difference: Family involvement in early childhood education. Retrieved from Harvard Family Research Project web site: http://www.gse.harvard.edu/hfrp/projects/fine/resources/ research/early childhood.html
- Wesley, P. W., & Buysse, V. (2003). Making meaning of school readiness in schools and communities. *Early Childhood Research Quarterly*, 18(3), 351-375.
- Zigler, E., Gilliam, W. S., & Jones, S. M. (2006). A vision for universal preschool education. New York, NY: Cambridge University Press.

### **APPENDIX A**



September 7, 2010

Dear Parent of Upcoming Kindergartner – Fall 2011:

The Grand Forks Public Schools, in partnership with the NDSU Extension Service, is looking for families for the Gearing Up for Kindergarten readiness program. This program will help Grand Forks Public Schools and NDSU staff better assist parents in preparing their children for school.

- Did you know that many parents want help in preparing their child for kindergarten?
- Do you have a child that is preparing to enter kindergarten in Grand Forks Public Schools in the fall of 2011?

The Gearing Up for Kindergarten readiness program is designed to learn from parents who have a child getting ready for kindergarten in the fall of 2011. Also, it allows parents to understand their child's development and growth in preparing for school.

Parents who respond will have the opportunity to be involved with their child in kindergarten readiness activities and programming. Children who are prepared for kindergarten have a better kindergarten experience, enjoy school more, and are more likely to achieve school success.

Due to the large pre-kindergarten population and the limited class size, participants who enroll will be placed in one of two programming options. Enrollment in the program begins immediately and interested families should respond by September 15, 2010.

Participants in the program will receive information about school readiness and be involved in parent-child activities in a school setting. In addition, all participants will engage in assessments designed to further understand each child's school readiness. All participating families will receive three \$25 gift certificates to local retail stores.

Interested parents must have a child preparing to enter kindergarten in the fall of 2011 in the Grand Forks Public Schools. Please consider participation or sharing this information with possible participants.

To register, please complete the registration form attached to this letter and return by September 15<sup>th</sup> in the stamped and addressed envelope accompanying this letter. You will be notified of the details of the program with your acceptance letter into the program.

If you have questions, please call Judith Konerza at the Parent Information Center at (701) 787-4216.



# **REGISTRATION FORM**

Parents Name:
First and last name of child entering kindergarten in the fall of 2011:
Birth Date of child:
Phone number with area code:
Street address:
City, State, Zip:
Email address:
Birth Date of child: Phone number with area code: Street address: City, State, Zip: Email address:

Current elementary school for other children:

#### APPENDIX B



September 22, 2010

Dear Parents,

Thank you for your response to the Gearing Up for Kindergarten readiness program. You and your child have been selected to participate in this hands-on learning program that will take place on Monday or Tuesday evenings, beginning October 11th and 12th, 2010. Our Monday night class will be held at Phoenix Elementary School and the Tuesday night class will be held at the University Children's Center on the University of North Dakota (UND) Campus.

Gearing Up for Kindergarten begins at 6:00 p.m. and ends at 7:30 p.m. each night. Free onsite childcare is provided for brothers and sisters ages six months through 5<sup>th</sup> grade. The evening schedule is listed below:

6:00- 6:45 p.m. Children and parents together with teacher in the classroom

6:45-7:15 p.m. Children remain with teacher and parents with parent instructor

7:15-7:30 p.m. Parents rejoin children to play interactive learning game

During the class time parents and children will participate together in a variety of learning activities that include painting, math, science, reading, and music. Teachers will also conduct activities for the entire group.

The parent time will provide parents with information on parenting styles, normal child development, reading ideas, and other important topics relating to school readiness. During that time, parents will also make a related activity to take home and practice with their child that week. Children will also take home a book selection each week and bring it back to exchange for another book the next week.

The dates for Gearing Up for Kindergarten (16 week program):

Mondays: October 11, 18, 25, November 1, 8, 15, 22 and 29. Break. Classes begin again on January 24, 31, February 7, 14, 21, 28, March 7 & 14. Monday night classes are held at Phoenix Elementary School.

<u>Tuesdays</u>: October 12, 19, 26, November 2, 9, 16, 23 and 30. **Break.** Classes begin again on January 25, February 1, 8, 15, 22, March 1, 8 & 15. Tuesday night classes are held at University Children's Center on the UND campus.

You and your child will participate in assessments as mentioned in the recruitment letter. Parents and your upcoming kindergartner are asked to come to the Parent Information Center, located at Lewis and Clark Elementary School, Room 1, on October  $6^{th}$  or  $7^{th}$  (choose which ever night works better for you) between 5:00 and 7:00 p.m. During this time, you will complete school readiness assessments (your thoughts on your child's school readiness) and your upcoming kindergartner will complete an assessment of alphabet and number skills. You may come between 5:00 and 7:00 pm and can expect to be here for half an hour. You will receive your first \$25 gift card upon completion of the assessments, to be mailed to your home address.

Class for you and your child will begin the following week. On the return form, please indicate a preference for the Monday or Tuesday evening class. We will place participants in classes in the order we receive your response form.

The staff of Gearing Up for Kindergarten and the Parent Information Center look forward to partnering with you, as together, we prepare your child for kindergarten. If you have any questions, please call the Parent Information Center at 787-4216.

Your partner in Education,

Judith Konerza

#### APPENDIX C



September 22, 2010

Dear Parents,

Thank you for your response to the Gearing Up for Kindergarten readiness program. You and your child have been selected to participate in this study helping researchers and educators better assist parents in school readiness preparation.

You and your child will participate in assessments as mentioned in the recruitment letter. Parents and your upcoming kindergartner are asked to come to the Parent Information Center, located at Lewis and Clark Elementary School, Room 1, on October 6<sup>th</sup> or 7<sup>th</sup> (choose which ever night works better for you) between 5:00 and 7:00 p.m. During this time, you will complete school readiness assessments (your thoughts on your child's school readiness) and your upcoming kindergartner will complete an assessment of alphabet and number skills. You may come between 5:00 and 7:00 pm and can expect to be here for half an hour. You will receive your first \$25 gift card upon completion of the assessments, to be mailed to your home address.

The staff of Gearing Up for Kindergarten and the Parent Information Center look forward to partnering with you, as together, we prepare your child for kindergarten. If you have any questions, please call the Parent Information Center at 787-4216.

Your partner in Education,

Judith Konerza

Please complete this portion and return in the self addressed stamped envelope. Thank you!



Parents Name:	
Childs Name:	
Address:	
City, State, Zip:	
Phone number with area code:	
Yes, my child and I will participate	

## **APPENDIX D**

# Practical Parent Assessment for School Readiness

ID Code #
Name
Contact Information
Address
Phone ( )
E-mail

Note – Your personal information will be kept private and identified only to the project director and researcher. This is to provide you with feedback from your assessment if you are interested in that information. Please check whether you would like feedback.

🗆 Yes 🛛 No

# Practical Parent Assessment for School Readiness



Developed by Sean Brotherson, Ph.D., North Dakota State University

This questionnaire is designed to assess parents' perceptions of a preschool child's (ages 3 to 5) preparation in selected areas for entering school and feeling successful in that setting (i.e., "school readiness"). It is intended as both a learning and brief assessment tool, and focuses on your awareness of a child's practical and developmental skills and abilities. Also, it can help assess a parent's perception of a child's developmental progress through time.

#### Perceptions of Your Child

Please consider a particular child in your life and provide your best understanding of *how your child typically thinks and behaves in the following areas.* There are no wrong answers. Please be honest and accurate and complete the entire questionnaire. Circle a number for each answer. Children vary, so you should understand you are not "grading" your child, but rather sharing your awareness of his or her development in several areas.

In answering the questions, please use the following 1 to 5 scale:

1 = Hardly Ever 2 = Once in Awhile 3 = Sometimes 4 = Often 5 = Almost Always

#### I. Approaches to Learning

My child typically	Hardly Ever	Once in Awhile	Some- times	Often	Almost Always
1. Is curious and asks questions to learn more about a topic	1	2	3	4	5
2. Works with hands in putting together puzzles or building with blocks	1	, 2	3	4	.5
3. Explores the environment and participates in new opportunities	1	2	3	4	5
4. Learns and recites familiar songs, nursery rhymes, finger plays or stories	1 .	2	3	4	5
5. Enjoys participating in small groups to play games or do learning activities	1	2	3	4	5
6. Draws and uses art to express ideas or feelings	1	2	3	4	5
7. Is able to work independently on a task (painting, picking up clothes, etc.)	1	2	3	4	5
8 Likes to work at solving problems	1	2	3	4	5
9. Is persistent when solving problems	1	2	3	4	5

#### II. Well-Being and Motor Development

Му	child typically	Hardly Ever	Once in Awhile	Some- times	Often	Almost Always
10.	Holds a pencil or crayon with fingers (not fist)	1	2	3	4	5
11.	Can button, snap or zip pants or coat	. 1	2	3	4	. 5
12.	Moves easily from place to place when walking	1	2	3	4	5
13.	Creates things using hands and paper, crayons, scissors, etc.	1	2	3	4	5
14.	Is able to cut with scissors	1	2	3	4	5
15.	Can catch or pick up a rolled or thrown soft ball	1	2	3	4	5
16.	Shows large motor skills, such as hopping, skipping, running, jumping or throwing	1	2	3	4	5

#### **III.** Social Development

My child typically	Hardly Ever	Once in Awhile	Some- times	Often	Almost Always
17. Shares with other children (such as toys, crayons, etc.)	1	2	3	4	5
18. Works or plays cooperatively with friends	1	2	3	4	5
19. Listens in a group situation	1	2	3	4	5
20. Takes turns with others		2	3	4	5
21. Remembers and follows simple directions in playing games	1	2	3	4	5
22. Enjoys talking to or playing with other children	1	2	3	4	5
23. Gets along without quarreling or fighting	1	2	3	4	5
24. Understands and follows rules that are explained	1	2	3	4	5
25. Understands how to enter into talking or play with other kids	1	2	3	4	5
26. Tries to solve problems with friends	1	2	3	4	5

.

# IV. Cognition and General Knowledge and Language Development

My child typically	Hardly Ever	Once in Awhile	Some- times	Often	Almost Always
27. Is familiar with books and enjoys having others read to him or her	1	2	3	4	5
28. Tries to learn and use new words	1	2	3	4	5
29. Is able to speak and communicate thoughts, needs and feelings in a way others understand	1	2	3	4	5
30. Is familiar with (or able to recognize) letters of the alphabet	1	2	3	4	5
31. Is familiar with (or able to recognize) numbers between 1 and 10 (or highe	er) 1	2	3	4	5
32. Is able to do counting, understand basic numbers (adding items together,	etc.) 1	2	3	4	5
<ol> <li>Can sort things into basic groups (such as by color, shape, things that match, etc.)</li> </ol>	1	2	3	4	5
34. Is creative in asking questions or trying to solve problems	1	2	3	4	5

#### V. Emotional Development

My	child typically	Hardly Ever	Once in Awhile	Some- times	Often	Almost Always
35.	Feels loved and supported from family members at home	1	2	3	4	5
36.	Spends time away from me in a variety of settings (child care, friend's house) without too much anxiety	1	2	3	4	5
37.	Can identify his or her feelings (such as sad, angry, happy)	1	2	3	4	5
38.	Can talk about how others might feel	1	2	3	4	-5
39.	Manages feelings, such as anger or frustration, without hurting or being mean to others	1	2	3	4	5
40.	Can adapt to new situations within a reasonable period and feel comfortable	1	2	3	4	5
41.	Regularly shares how he or she is feeling with me or other caring adults	1	2	3	4	5
42.	Appreciates getting to know new people or friends through time	1	2	3	4	5

# Transition-to-School

Sector Construction

My child:	Not at all	To a slight extent	To a fair extent	To a great extent
6. Is excited about beginning kindergarten	1	2	3	4
<ol><li>Is familiar with the routines and expectations that exist in kindergarten (raising hands, etc.)</li></ol>	1	2	. 3	4
8. Is comfortable with the environment of the school setting	1	2	3	4
9. Talks positively about going to kindergarten	1	2	3	4
10. Is comfortable interacting with peers of a similar age who will be		•		:
	1 Belokter Britshie	Z	<b>3</b> 10010-00000-0000	<b>4</b> ************************************
11. Is comfortable separating from me as a parent	1	2	3	4
12. Is comfortable participating in a group with other children	1	2	3	4
13. Is curious about school and the experiences he/she will have there	. 1	2	3	4

### APPENDIX E

## Gearing Up for Kindergarten Study

# Parent Response Form – Time 1

This brief form is designed to gather feedback from parents and caregivers who participate in the Gearing Up for Kindergarten study. It should be completed at the beginning of participation in the study. Please take a moment to share your thoughts with us. Circle or check the answer that applies to you. When you have completed this form, please return it to the person who provided it to you, in the self-addressed stamped envelope.

#### Demographic Information

- 1. I am:
- □ Male
- □ Female
- 2. Age
- 3. Number of children \_ Ages of children
- 4. I live:
  - □ Farm/Ranch
  - 🛛 Rural nonfarm
  - □ Town less than 10,000
- □ Town/City from 10,000 to 50,000
- □ City more than 50,000
- 5. Family status □ Married
  - □ Single (never married)
  - □ Remarried
  - □ Separated/Divorced
  - □ Cohabiting
  - □ Widowed
- 6. Education
  - □ Some high school
  - □ High school/GED
  - □ Some college
  - □ 2-year college degree
  - □ 4-year college degree
  - Master's degree or higher



2011

serval orientation, status as a U.S. veteran, race ce President for Equity, Diversity and Global Ou 1-7708, County Commissions, NDSU and U.S. [

- 7. Employment status Employed full time
- (40+ hours a week)
- Employed 26 to 39 hours a week
- Employed less than 25 hours a week
- Seeking employment
- Not seeking outside employment (includes full-time homemaker, student, retired, etc.)
- Other
- 8. Racial or ethnic background Caucasian
  - Native American or Alaskan Native

  - Asian
  - Hispanic
  - □ Other
- 9. Are you eligible to participate in any of these programs? Please check the one(s) that apply.
  - U WIC
  - □ TANF
  - □ Food Stamps
  - Head Start
  - □ Medicaid
  - □ Free/reduced lunch
- 10. I have a child who will likely enter kindergarten next year □ Yes D No
- 11. Is this your first child to enter kindergarten? □ Yes D No
- 12. Has your child been enrolled as a participant in Early Head Start/Head Start? □ Yes D No

- 13. Has your child been in out-of-home care in any of the following settings? Please check all that apply. □ Preschool
  - Center-based child care
  - □ Family-based child care
  - □ Other child care
- 14. If your child has been involved in out-of-home care, please indicate the number of hours per week. (skip if does not apply) □ 6 hours or less
  - □ 6-12 hours a week
  - □ 12-20 hours a week
  - 20 hours or more
- 15. My relationship to this child is:
  - □ Biological mother (or adoptive)
  - Biological father (or adoptive)
  - □ Stepparent
  - Grandparent □ Relative (aunt, uncle)
  - □ Other caregiver
- 16. My child in this program is: □ Male
  - □ Female



□ African-American

# **APPENDIX F**



AIMSweb® Letter Naming Fluency - Benchmark Assessment #1 (Kindergarten - Fall) Copyright © 2003 NCS Pearson, Inc. All rights reserved. www.AIMSweb.com

# **APPENDIX G**



AIMSweb® Number Identification - Benchmark Assessment #1 (Kindergarten - Fall) Copyright © 2004 by NCS Pearson, Inc. All rights reserved. www.AIMSweb.com

L.

# **APPENDIX H**

#### AIMSweb® Oral Counting – Benchmark Assessment

Given To: _				Given B	y:		Date:			
1	2	3	4	5	6	7	8	9	10	
11	12	13	14	15	16	17	18	19	20	
21	22	23	24	25	26	27	28	29	30	
31	32	33	34	35	36	37	38	39	40	
41	42	43	44	45	46	47	48	49	50	
51	52	53	54	55	56	57	58	59	60	
61	62	63	64	65	66	67	68	69	70	
71	72	73	74	75	76	77	78	79	80	
81	82	83	84	85	86	87	88	89	90	
91	92	93	94	95	96	97	98	99	100	

Copyright © 2003 by NCS Pearson, Inc. All rights reserved. www.AIMSweb.com

### **APPENDIX I**

### NDSU North Dakota State University Dept. of Human Development and Family Science Dept. #7260, P.O. Box 6050 Fargo, ND 58108-6050 PH: (701) 231.8113

**Title of Research Study:** *Gearing Up for Kindergarten – Evaluation Project (2010-2011), Grand Forks site* 

**This study is being conducted by:** Sean Brotherson, Ph.D., Sharon Query, Ph.D., & Jim Deal, Ph.D., Human Development and Family Science, NDSU; Divya Saxena, M.S., NDSU Extension Service; Kathy Enger, Ph.D., School of Education, NDSU; Judith Konerza, Ph.D. candidate, School of Education, NDSU.

Why am I being asked to take part in this research study? What is the reason for doing the study? As a participant in Gearing Up for Kindergarten you will be asked your perceptions of your child's school readiness, aspects of your parent and child relationship, and to assess the value of the educational program of Gearing Up for Kindergarten.

# **What information will be collected about me?** *You will be asked to complete the following instruments during the parent education sessions:*

• Gearing Up for Kindergarten – Evaluation ID Sheet

- Parent Response Pre and Post Program Form Session A
- Practical Parent Assessment for School Readiness
- Family Information Form Times 1
- Teacher Assessment for School Readiness

Also, at your particular site, your pre-kindergarten child will participate in the following assessment:

• AIMSWeb diagnostic assessment for school readiness

Where is the study going to take place, and how long will it take? The study will take place at your Gearing Up for Kindergarten site in Grand Forks, ND, and completion of the study instruments will require 20 to 30 minutes. This will take place 3 times over a 12 month time frame.

What are the risks and discomforts? It is not possible to identify all potential risks in research procedures, but the researchers have taken reasonable safeguards to minimize any known risks to the participant. Risks might include loss of confidentiality and emotional or psychological distress.

What are the benefits to me? Through participation in the Gearing Up for Kindergarten program, participants are allowed to see how their children are doing in key areas related to school readiness. Parents get practical feedback specific to a child that could be used in discussing further needs regarding school readiness. The results from the study will be used to make program improvements to Gearing Up for Kindergarten, as well as to help

garner funding for program enhancement. However, you may not get any benefit from being in this research study.

**Do I have to take part in the study?** Your participation in this research is your choice. If you decide to participate in the study, you may change your mind and stop participating at any time without penalty or loss of benefits to which you are already entitled.

What are the alternatives to being in this research study? Instead of being in this research study, you can choose not to participate.

Who will see the information that I give? We will keep private all research records that identify you, to the extent allowed by law. Your information will be combined with information from other people taking part in the study. When we write about the study, we will write about the combined information that we have gathered. You will not be identified in these written materials. We may publish the results of the study; however, we will keep your name and other identifying information private.

We will make every effort to prevent anyone who is not on the research team from knowing that you gave us information, or what that information is. For example, your name will be kept separate from your research records and these two things will be stored in different places under lock and key. We will provide you with an identification code, such as #40001, that will be used in place of your identifying information. Links between this code and your identifying information will be disposed of upon completion of the project.

#### What if I have questions?

Before you decide whether to accept this invitation to take part in the research study, please ask any questions that might come to mind now. Later, if you have any questions about the study, you can contact the researcher, Sean Brotherson, at <u>sean.brotherson@ndsu.edu</u> or Judith Konerza, at <u>judith.konerza@gfschools.org</u>

#### What are my rights as a research participant?

You have rights as a participant in research. If you have questions about your rights, or complaints about this research, you may talk to the researcher or contact the NDSU Institutional Review Board (IRB) by:

- Telephone: 701.231.8908
- Email: <u>ndsu.irb@ndsu.edu</u>
- Mail: NDSU HRPP Office, NDSU Dept. #4000, P.O. Box 6050, Fargo, ND 58108-6050

The role of the IRB is to see that your rights are protected in this research; more information about your rights can be found at: <u>www.ndsu.edu/research/irb</u>.

### **Documentation of Informed Consent:**

You are freely making a decision whether to be in this research study. Signing this form means that

- 1. you have read and understood this consent form
- 2. you have had the consent form explained to you
- 3. you have had your questions answered, and
- 4. you have decided to be in the study.

You will be given a copy of this consent form to keep.

Your signature

Date

Your printed name

Signature of researcher explaining study

Judith Konerza Printed name of researcher explaining study Date

### APPENDIX J

NDSU

#### NORTH DAKOTA STATE UNIVERSITY

Institutional Review Board Office of the Vice President for Research, Creative Activities and Technology Transfer 1735 NDSU Research Park Drive P.O. Box 5756 Fargo, ND 58105-5756

December 5, 2008



701.231.8908 Fax 701.231.8098

Federalwide Assurance #FWA00002439 Expires April 24, 2011

Dr. Sean Brotherson Dept. of Child Development & Family Science 277 EML

Re: IRB Certification of Human Research Project: "Gearing Up for Kindergarten - Control Study" Protocol #: HE09112

Co-investigator(s) and research team: Sharon Query, Divya Saxena, Elise Pforr

Study site(s): NDSU

Funding: n/a

The IRB has determined that this human subjects research project qualifies for exempt status (category # 2) in accordance with federal regulations (Code of Federal Regulations, Title 45, Part 46, Protection of Human Subjects).

Please also note the following:

- This determination of exemption expires 3 years from this date. If you wish to continue the research after 12/4/2011, submit a new protocol several weeks prior to this date.
- The project must be conducted as described in the approved protocol. If you wish to make changes, pre-approval is to be obtained from the IRB, unless the changes are necessary to eliminate an apparent immediate hazard to subjects. A *Protocol Amendment Request Form* is available on the IRB website.
- Prompt, written notification must be made to the IRB of any adverse events, complaints, or unanticipated problems involving risks to subjects or others related to this project.
- Any significant new findings that may affect the risks and benefits to participation will be reported in writing to the participants and the IRB.
- Research records may be subject to a random or directed audit at any time to verify compliance with IRB policies.

Thank you for complying with NDSU IRB procedures; best wishes for success with your project.

Sincerely, Tervi Grosz, MS, CIP

**IRB** Director

NDSU is an equal opportunity institution.

.

#### **Institutional Review Board**

... for the protection of human participants in research

North Dakota State University Sponsored Programs Administration 1735 NDSU Research Park Drive NDSU Dept #4000 PO Box 6050 Fargo, ND 58108-6050 231-8995(ph) 231-8098(fax)

DADE CENVE Office of Sponsored Programs Administration

# Protocol Amendment Request Form

Changes to approved research may not be initiated without prior IRB review and approval, except where necessary to eliminate apparent immediate hazards to participants. Reference: <u>SOP 7.5 Protocol Amendments</u>.

Examples of changes requiring IRB review include, but are not limited to changes in: investigators or research team members, purpose/scope of research, recruitment procedures, compensation scheme, participant population, research setting, interventions involving participants, data collection procedures, or surveys, measures or other data forms.

Protocol Information	n:
Protocol #: HE09112 Title: Gearing Up for Kindergarten - Co	ontrol Study
Review category: 🛛 Exempt 🗌 Expedited	Full board
Principal investigator: Sean Brotherson Email address: sean. Dept: Human Development and Family Science	brotherson@ndsu.edu
Co-investigator: Sharon Query Email address: sharon.qu Dept: Center for 4-H Youth Development, Human Development	ery@ndsu.edu nt and Family Science
Principal investigator signature, Date:	n ; 1/20/2010
Description of proposed c	hanges:
<ol> <li>Date of proposed implementation of change(s)*: February 1, 2 * Cannot be implemented prior to IRB approval unless the IRB Chai to eliminate apparent immediate hazards to participants.</li> </ol>	2010 r has determined that the change is necessary K.Engur - 111712008
<ol> <li>Describe proposed change(s), including justification:         <ul> <li>a) Addition of two members to the research team. These School of Education, NDSU, and Judith Konerza, PhD car Judith Konerza would be using this data to pursue her dis</li> </ul> </li> </ol>	J. Konerza - 1/14/2069 individuals would be Kathy Enger, PhD, indidate, School of Education, NDSU. ssertation research project. The

b) Addition of a specific site to recruit participants for local comparison in Grand Forks, ND. This will allow a more intensive investigation at a specific site with study participants, as there will be one added dimension for this group (addition of a specific instrument).

addition will allow them to focus specifically on a specific site.

Protocol Amenancia Request Form NDSU Institutional Review Buard Form revised June 2009

1.364 bitmics 01 [10.2714; 1.50.0017]

c) Addition of the pre-kindergarten children of study participants at this specific new site only, to be assessed using a diagnostic educational test related to literacy and numeracy skills (the AIMSWeb survey tool). The measure would be completed on the same schedule as the other measures to be completed by participating adults (time 1, time 2, etc.).

d) Addition of two research instruments to be given to the study participants:

\* Family Information Form - this questionnaire asks participants about parental involvement, school readiness expectations, and child personality. This questionnaire would be included and given on the exact same schedule as other instruments (time 1, time 2, etc.)

\* AIMSWeb Early Numeracy & Literacy Kindergarten Benchmark Assessments - this questionnaire is done with a pre-kindergarten child and allows them to express literacy and numeracy skills. It takes 10 minutes. This instrument would only be used with the prekindergarten children of families recruited at the added study site in Grand Forks, ND.

(Copies of the research instruments are attached in Appendix A)

e) Expanded parental informed consent form. A slightly revised and expanded parental consent form for those individuals recruited to participate at the Grand Forks site, which includes information about the child's brief participation, is included.

☐ Yes: Include an Investigator's Assurance (last page of protocol form), signed by the new PI or coinvestigator.

Note: If the change is limited to addition/change in research team members, skip the rest of this form.

4. Will the change(s) increase any risks, or present new risks (*physical, economic, psychological, or* <u>soc</u>iological) to participants?

🛛 No

Section Yes: In the appropriate section of the protocol form, describe new or altered risks and how they will be minimized.

5. Does the proposed change involve the addition of a vulnerable group of participants? Children: ☐ no ☑ yes – include the *Children in Research* attachment form Prisoners: ☑ no ☐ yes – include the *Prisoners in Research* attachment form Cognitively impaired individuals: ☑ no ☐ yes\* Economically or educationally disadvantaged individuals: ☑ no ☐ yes\*

\*Provide additional information where applicable in the revised protocol form.

 Does the proposed change involve a request to waive some or all the elements of informed consent or documentation of consent?

 ∧ no

yes - include the Informed Consent Waiver or Alteration Request attachment form

Protocol Amendmish Request Form NDSU hrannional Review Board Porm revised June 2009

7. Does the proposed change involve a new research site? ☐ no ⊠ yes – include a letter of permission/cooperation, IRB approval, or grant application or contract

#### Attach a copy of the approved protocol, with highlighted change(s) incorporated within the relevant section(s).

	Impact for Participants (future, current, or prior):
1.	Will the change(s) alter information on previously approved versions of the recruitment materials, informed consent, or other documents, or require new documents? ☐ No ⊠ Yes - attach revised/new document(s)
2.	Could the change(s) affect the willingness of <i>currently</i> enrolled participants to continue in the research? ⊠ No ☐ Yes - describe procedures that will be used to inform current participants, and re-consent, if necessary:
3.	Will the change(s) have any impact to <i>previously</i> enrolled participants? ⊠ No □ Yes - describe impact, and any procedures that will be taken to protect the rights and welfare of participants:

-----FOR IRB OFFICE USE ONLY -----

Request is: ⊠ Approved ∐ Not Approved Review: ⊠Exempt, category#: (/ □Expedited method, category # □Convened meeting, date:		
IRB Signature: Krstyshinley	Date: 2/10/2010	

Protocols previously declared exempt: (Allow 5 working days) If the proposed change does not alter the exemption status, the change may be administratively reviewed by qualified IRB staff, chair, or designee. If the change(s) would alter this status, Expedited or Full Board review will be required.

Protocols previously reviewed by the expedited method: (Allow 10 working days) Most changes may also be reviewed by the expedited method, unless the change would increase risks to more than minimal, and/or alter the eligibility of the project for expedited review.

Protocol Amendment Request Form NDSU Institutional Review Board Form revised June 2009

Page 1 of 4 Last printed 01/20/2010 4:20:00 PM

### **APPENDIX K**



September 15, 2011

Hello Parents,

Your four year old is now five, and a kindergartner. We are eager to hear how he/she is doing. It is also time for the third round of assessments for the Gearing Up for Kindergarten Program you are participating in.

The Parent Information Center will be open on September  $28^{th}$  and  $29^{th}$  from 1:00 pm – 6:00 pm for you to stop by and fill out the final assessment. We look forward to learning how your child has grown over the summer and into the fall. We have the WalMart gift cards here, too, and will hand them out upon completion of the assessments.

Please park on the east side of the school in the Elks swimming pool parking lot and enter the school through Door #1 on the east side of the school. If you park anywhere else, you will most likely be ticketed.

If your child <u>is not</u> enrolled in the Grand Forks Public Schools, we invite you to bring him/her along so we can complete the AIMS Web assessment at that time.

Please call the Parent Information Center office at 701-787-4216 if you are unable to come on either of these days and we will make other arrangements for the assessments.

We look forward to seeing you again!

Judith Konerza, Parent Coordinator



# **APPENDIX L**



# Grand Forks Public Schools



Mission Statement: To Provide Opportunities for All Students to Develop Their Maximum Potential

Parent Information Center 3351 17<sup>th</sup> Avenue South Grand Forks, ND 58201 pic@gfschools.org www.gfschools.org Judith Konerza Coordinator Phone: 701.787.4216 Fax: 701.787.4079 judith.konerza@gfschools.org

January 21, 2010

To Whom It May Concern:

The Parent Information Center, Grand Forks Public Schools in partnership with NDSU Extension, is hosting the Gearing Up for Kindergarten program at the Lake Agassiz Head Start building and will be conducting related research by NDSU faculty, staff, and graduate student, Judith Konerza.

Grand Forks Public Schools will cooperate as needed to assist program research participants, to provide a location, and collect information to be used by NDSU for research purposes.

Thank you for your consideration.

Sincerely

Jody Thompson Assistant Superintendent Grand Forks Public Schools

\_Providing Equal Opportunities in Education and Employment\_

		Phone:
Date: Name:		Sal Della Hall
January 26, 2010 Judith	Konenzo	101-181-0414
Fax or Email: Koneya @ cf er	and a car	Research Advisors: Seen Emother
701-787-4079	00000	Plathy Engler
and Centum School		College or Dept.:
Address: 2361 17 the Aveso. Uncli Dept & Education Hum		NDSU pept & Education Human
Grond Forks, ND 5820		Developm
Research Title: Georing Up for Kendersonten - Evolution Project (2010), Sei		
Grand Jords sate		
Give a brief description of your resear	rch. Attach additional papers if necessa	ry. Please attach sample copies of
assessment instrument, tests, or comm	unications to be used:	A Alex Constanting Har
This research proj	eet will assess the effection	evens of the Gearing up
for Kindergonten program in preparing pre- & children for kindergerten.		
This unique approach works together with parets and children in		
a classrion setting. The assessment pieces will be used to refine		
the program in creating the most effective curriculum possible		
for Kinderconten readiness.		
see attached		
	No	Grade Level or Dent.:
Number of students needed for	Number of teachers needed for	Grade Develor Depart
research: wetween	researcu.	Pre-K
60-80	10	
What schools are you interested in conducting the research in? New Start @ Lake Agassiz site		
follow us assessment	will access Kinderearten	ADAS web conducted
at the child's fender 50	sten school location in Gr	as Jortes Public Schooles
Will confidential records be required	? (If yes, indicate type.) yes,	Length of time required to complete the research:
the print was appoint on file in the long and 2010 - long manifold		
Strangenten gleen at the churces strike a great the former former former former		
+ with the second to have a life		
To be completed by School District Official:		
Annound		
Not Approved:		
Assistant Superintendent Signature:	M~	Date: 1-27-10
Approved to conduct research in the following schools:		

# Request to Conduct Research in the Grand Forks Public Schools

Send completed form to: Grand Forks Public Schools, Box 6000, Grand Forks, ND 58206-6000 Attn: Assistant Superintendent's Office

Printed by: Judith Konerza Title: Re: research : gffirstclass		Monday, October 04, 2010 9:37:17 AM Page 1 of 1
From:	🗿 Jody Thompson	Monday, October 04, 2010 9:31:11 AM 📧 🎯
Subject:	Re: research	
To:	Judith Konerza	

Judith Konerza writes:

I have the Request to Conduct Research from last year and we gave it a time frame of Jan 2010 - January 2011. Should I fill out another one, or can we initial this one and extend it to September 2011? I will extend the current one to Fall 2011. Thanks. thanks much, Judith

Judith Konerza judith.konerza@gfschools.org Parent Resource Coordinator Grand Forks Public Schools 701-787-4216

We make a living by what we get. We make a LIFE by what we GIVE. Winston Churchill

Mr. Jody Thompson Assistant Superintendent of Teaching and Learning Grand Forks Public School District Grand Forks, North Dakota 58206 701-746-2200